Biogenic amines as mediators of queen mandibular pheromone's control of worker reproduction in the honey bee, *Apis mellifera*

Rosemary Anne Knapp

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

The University of Leeds School of Biology

May 2022

The candidate confirms that the work submitted is her own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

The right of Rosemary Anne Knapp to be identified as Author of this work has been asserted by Rosemary Anne Knapp in accordance with the Copyright, Designs and Patents Act 1988.

Acknowledgements

First and foremost, I would like to express my appreciation for my truly brilliant supervisor, Elizabeth Duncan. I'd like to thank Liz for her infectious enthusiasm, genuine kindness and unrelenting support from start to finish. From sparking my interest in social insects as an undergraduate to encouraging me to apply for this PhD, it is not an overstatement to say that I wouldn't be where I am today without her. I would genuinely struggle to envisage a better PhD supervisor, and I promise she hasn't bribed me to say that (though the many coffees, cookie cups, cheesy chips and gravy and celebratory pints she treated me to over the years have certainly acted in her favour).

Thank you also to my co-supervisor Bill Kunin, to the University of Leeds for providing my Doctoral Scholarship, and to the School of Chemistry for the use of their HPLC facilities. In particular, I am grateful to Jeanine Williams for her assistance with HPLC-FLD sample processing.

I am fortunate to have had the support of a fantastic lab group during my time at Leeds. Jens's vast statistical expertise has been hugely helpful, if at times slightly terrifying. I'd like to thank Kane and Ant for always making the lab a fun place to be – Kane for fostering an atmosphere of silliness and Ant for filling our heads with strange facts we didn't know we needed. My thanks also go out to Milo, for enduring hours upon hours of red-roomsims-music-accent-imitating delirium with me during behavioural assays and still agreeing to be my friend afterwards. Thank you also to the wonderful MSc students Emily and Jess who assisted with some of the data collection for this thesis.

Special thanks go out to two delightful post-docs, James Rouse and Victoria Norman, for all they have taught me. Although I can't say the Very Scientific Rock Pooling James taught me during the first year of my undergrad has been of much practical use here, his lessons in data analysis, RNA extractions and qPCR have been integral to this project and are truly appreciated. Thank you to Victoria for being aggressively supportive, making me laugh to the point of tears, and teaching me more than you will ever realise during the short time I was lucky enough to work with you. I would also like to mention Mackenzie, for being a fantastic scientist and for all the fun we had working together briefly. Here's to hoping our paths cross again in the future.

I am grateful for the friendships I have formed throughout my PhD which have made the past four years such a memorable time. To Jake, Harri, Chris, Louise and Steph, who's company during long lunch breaks and after work pints has been a solace from the inevitable stress that comes with postgraduate research. To my good friends Adele and Simon, for the wholesome weekends filled with adventures, board games and tea which were the perfect antidote to long weeks in the lab. To Britt, in whom I have found a lifelong friend and who means the absolute world to me.

Thank you also to Katie, who has been on this PhD rollercoaster with me from beginning to end. We have celebrated the highs and commiserated the lows, cohabited through a pandemic, and established healthy coping strategies centered around Chinese food, fizzy sweets and cats.

Finally, I would like to thank my family for their love and support, and for always doing their best to understand what exactly it is that I do. Lastly, thank you to my partner Ciaran for grounding me and never letting me give up. Nothing feels as daunting with you by my side.

Abstract

The occurrence of reproductive and non-reproductive castes is one of the major hallmarks of eusociality. Constraints on worker reproduction negate conflict over queenworker reproduction, playing an integral role in the maintenance of colony harmony. The primary aim of this thesis was to improve our understanding of one of the major mechanisms maintaining colony cohesion in honey bees: the inhibition of worker reproduction by QMP. A body of circumstantial evidence already implicated biogenic amines as mediators of QMP's command over the behaviour and ovarian physiology of workers. This thesis aimed to bring further clarity to the roles played by biogenic amines in regulating reproductive constraint, finding evidence that QMP's ability to modulate dopamine and octopamine is likely a pathway through which sterility in workers is induced.

Considering the accumulation of evidence that QMP 'chemically castrates' workers by manipulating fundamental reproductive pathways, the question as to how queens evade the repressive effects of their own pheromone was also addressed. The hypothesis that discrepancies in QMP exposure levels between queens and workers maintains castespecific differences in QMP's inhibition of reproduction was empirically tested, and recommendations for further study are suggested.

Finally, the importance of QMP to colony function was considered within the context of an applied issue: pesticide-driven declines of honey bee populations. Despite QMP being central to honey bee biology, its interactive effects with commonly used pesticides has been under-researched. I identify a number of synergistic effects between QMP and the neonicotinoid pesticide imidacloprid on workers which may have negative effects for colony health and productivity. Given the demonstrated breadth of QMP's effects on the neuroendocrine system, reproductive physiology, and behaviour of workers, a case is made for the inclusion of QMP in future assessments of pesticide lethal and sub-lethal effects on honey bees.

Acknowledgements	ii
Abstract	iv
Table of Contents	V
List of Figures	ix
List of Tables	xi

Table of Contents

List of Tables	xi
List of Abbreviations	xii
Charter 1. Concreduction	1
1 1 Europiality	1
1.1 Eusociality	ـــــــــــــــــــــــــــــــــــــ
	⊥
1.1.2 Inclusive fitness theory	2
1.1.3 Conflict over reproduction	2
1.2 Reproductive Constraints	3
1.2.1 Irreversible constraints	
1.2.2 Reversible constraints	5
1.3 Queen Pheromones	b
1.3.1 Ancestral Hymenopteran queen pheromones	6
1.3.2 Queen mandibular pheromone	7
1.4 The Honest signal vs Queen control hypotheses	9
1.4.1 Honest signal hypothesis	9
1.4.2 Queen control hypothesis – Is QMP a suppressive agent?	
1.5 How does QMP inhibit reproduction?	12
1.5.1 Ovarian response to QMP – Notch signalling	12
1.5.2 Neuroendocrine control of reproduction	13
1.5.3 Biogenic amines	15
1.6 Research Aims	
Chapter 2: General Methods	20
2.1 Honey bee maintenance	20
2.1.1 Laboratory microcolonies	20
2.1.2 QMP administration	21
2.2 Ovary dissection and scoring	22
2.3 Measurement of biogenic amines using HPLC-FLD	23
2.3.1 Sample collection	23
2.3.2 Preparation of samples and standards	23
2.3.3 HPLC-FLD system	24
2.4 Data analysis	27

Chapter 3: Transmission of QMP's signal to the ovary	28
3.1 Introduction	28
3.2 Methods	31
3.2.1 Generation and maintenance of QR and QL hives	31
3.2.2 Mark-release-recapture experiment	31
3.2.3 Establishing laboratory microcolonies	33
3.2.4 Data analysis	33
3.3 Results	35
3.3.1 Recapture success rate	35
3.3.2 Ovary activity of QR and QL workers	35
3.3.3 Serotonin is unaffected by QMP	
3.3.4 Brain dopamine correlates with ovary activity	
3.3.5 Octopamine is modulated by QMP	41
3.4 Discussion	42
Chapter 4. Testing the squark relationship between depending levels and every	
chapter 4: resume the causal relationship between dopamine levels and ovary	40

activit	ty	48
4.	1 Introduction	48
4.	2 Methods	51
	4.2.1 Laboratory microcolony set-up	51
	4.2.2 Dopamine supplementation treatments	51
	4.2.3 L-Dopa supplementation treatments	52
	4.2.4 Dopamine inhibition treatments	53
	4.2.5 Data analysis	54
4.	3 Results	56
	4.3.1 Dopamine has no effect on ovary activity in the presence or ab	sence of
	QMP	56
	4.3.2 Dopamine and L-Dopa do not enhance ovary activity in QL worl	kers at any
	dose tested	58
	4.3.3 Iodotyrosine represses reproduction to a degree comparable w	vith
	QMP	61
4.	4 Discussion	64
Chapt	ter 5: Dose-dependent effects of QMP on reproduction	70
5.	1 Introduction	70
5.	2 Methods	74
	5.2.1 Microcolony set-up and experiment overview	74
	5.2.2 QMP treatments	74
	5.2.3 HPLC-FLD	74
	5.2.4 Quantitative RT-PCR	75
	5.2.5 Data analysis	76

processes	131
7.4 Worker reproduction is likely constrained by multiple redundant	
7.3.3 How is the ovarian response coordinated?	130
to the ovary?	127
7.3.2 How are dopamine and octopamine modulated by QMP and	transmitted
7.3.1 How is QMP detected?	125
7.3 Mechanism of QMP-biogenic amine signalling	124
7.2.2 Octopamine	123
7.2.1 Dopamine	120
worker reproduction?	120
7.2 What evidence is there that biogenic amines mediate QMP's con	trol of
7.1 Thesis overview	119
Chapter 7: General Discussion	119
6.4 Discussion	
6.3.6 OMP exposure increases the lethality of imidacloprid	109
6.3.5 Imidacloprid increases worker attraction to OMP	100 108
workers	106 net unit
6.3.4 OMP and imidacloprid did not affect the locomotor activity	of five-day old
6.3.3 Imidacloprid interacts with OMP to lower brain dopamine le	102 vels 104
treatment	102
6.3.2 Ovary activity is repressed by synthetic OMD regardless of in	nidacloprid
intako	100
6.3.1 Effects of imidacloarid and OMD treatments on food and we	100 tor
6 3 Recults	98 100
6.2.5 Data analysis	
6.2.4 Locomotor activity assay	94 סס
6.2.2 Innuaciophia and Qivir treatments	94 مە
6.2.2 Imidadoprid and OMP treatments	92
6.2.1 Microcolony set up and experiment eventions	92 מים
6.2 Methods	88 دە
Chapter 6: Investigating the interactivity between QMP and imidaclopric	88
5.4 Discussion	
dose	
5.3.4 Notch signalling underlies reproductive constraint irrespective	ve of QMP
5.3.3 QMP's inhibition of worker reproduction is lost at high doses	s
5.3.2 Brain dopamine and octopamine are not affected by OMP do	ose79
5.3.1 QMP treatments do not affect food intake or survival	
5.3 Results	78

7.5 Queen control vs queen signal hypotheses	
7.6 Applied research	
7.7 Concluding Remarks	
Appendices	
Appendix A	
Appendix B	
Appendix C	140
Appendix D	
Appendix E	
Appendix F	
Appendix G	
Appendix H	
Appendix I	
Appendix J	
Appendix K	
Appendix L	156
References	

List of Figures

Figure 1.1 - Differences in honey bee ovary morphology between castes	4
Figure 1.2 - Queen pheromone phylogeny	6
Figure 1.3 - Chemical structures of QMP components	8
Figure 1.4 - Schematic of Notch signalling	13
Figure 1.5 - Neuroendocrine signalling in Drosophila melanogaster	14
Figure 1.6 - Chemical structures of HVA and dopamine	17
Figure 1.7 – Causal relationships among QMP, DA and ovary activity diagram	18
Figure 2.1 – Honey bee cages	20
Figure 2.2 – Ovary scoring morphological features	22
Figure 2.3 – HPLC-FLD chromatogram	25
Figure 3.1 – Transmission of QMP's signal to the ovary	28
Figure 3.2 – Paint-marked bees	32
Figure 3.3 – Ovary activity of QR and QL hive and laboratory workers	36
Figure 3.4 – Biogenic amine levels in workers from QR and QL hives	37
Figure 3.5 – Biogenic amine levels in workers from QR and QL laboratory	
microcolonies	39
Figure 3.6 – Brain biogenic amine titres vary with ovary activity level of QL	
workers	40
Figure 4.1 – Schematic of the dopamine biosynthesis pathway	52
Figure 4.2 - Dopamine supplementation in the presence and absence of QMP	57
Figure 4.3 - Effects of dopamine supplementation at a range of doses	58
Figure 4.4 - Effects of L-Dopa supplementation at a range of doses	59
Figure 4.5 - Biogenic amine levels following L-Dopa supplementation	60
Figure 4.6 - Ovary activity levels following iodotyrosine treatment	62
Figure 4.7 - Survival and food and water intake rates of iodotyrosine treated	
bees	63
Figure 5.1 - Schematic of modelled QMP transmission throughout the hive	73
Figure 5.2 - Effects of QMP doses on food intake and survival	78
Figure 5.3 - Brain biogenic amine levels at different QMP doses	79
Figure 5.4 - Dose-specific effects of QMP on ovary activity	80
Figure 5.5 - Her and bHLH2 ovary expression by ovary score and QMP dose	82
Figure 5.6 - Hormesis dose-response curve	84
Figure 6.1 - Diagram of the retinue response assay arena	95
Figure 6.2 - Photo of the retinue response assay filming set-up	97
Figure 6.3 - Photo of the locomotor activity assay filming set-up	95

Figure 6.4 - Food and water intake of workers treated with QMP and	
imidacloprid	101
Figure 6.5 - Ovary activity is repressed by QMP regardless of imidacloprid	
dosage	
Figure 6.6 - Effects of QMP and imidacloprid treatments on biogenic amine	
levels	104
Figure 6.7 - Locomotor activity of QMP and imidacloprid treated workers	106
Figure 6.8 - Attraction of workers to QMP	107
Figure 6.9 - Effects of QMP and imidacloprid on survival	109
Figure 7.1 - Stages of oogenesis in a honey bee ovary	121
Figure 7.2 - Model of the transmission of QMP's signal to the ovary	125
Figure C.1 - Actual dopamine doses received per bee per day	140
Figure D.1- Actual L-Dopa doses received per bee per day	141
Figure E.1 – Actual iodotyrosine doses received per bee per day	142
Figure F.1 - Expression of AmDop1 in the honey bee ovary	144
Figure H.1 - Validation of single reference gene usage	149
Figure I.1 - Validation of QMP's repression of ovary activity in experiment	
two	151
Figure J.1 - Assessment of imidacloprid lethal dosage in laboratory	
microcolonies	144
Figure L.1 - Ovary expression of vitellogenin measured using RT-qPCR	156

List of Tables

Table 1.1 - Reproductive constraints	3
Table 1.2 – Principal QMP components and their functions	10
Table 2.1 – HPLC-FLD Injection program	26
Table 2.2 – HPLC-FLD mobile phase gradients	26
Table 3.1 – Mark-release-recapture success rate	35
Table 3.2 – Pairwise comparisons of brain dopamine levels by ovary score	38
Table 4.1 – Pairwise comparisons of ovary scores between dopamine and QMP	
treatments	56
Table 5.1 – Pairwise comparisons of survival between QMP doses	79
Table 5.2 - Pairwise comparisons for ovary scores between different QMP exposure	9
treatments	81
Table 5.3 - Pairwise comparisons for expression of <i>bHLH2</i> and <i>Her</i> between ovary	
scores	83
Table B.1 - Pairwise comparisons of biogenic amine levels in QR/QL workers on day	/S
10/21	139
Table C.1 – Pairwise comparisons of actual dopamine intake	.140
Table D.1 – Pairwise comparisons of actual L-Dopa intake	.141
Table E.1 – Pairwise comparisons of actual iodotyrosine intake	143
Table G.1 – Pairwise comparisons of ovary scores by QMP/iodotyrosine	
treatment	.145
Table G.2 – Pairwise comparisons of water intake by QMP/iodotyrosine	
treatment	.146
Table G.3 – Pairwise comparisons of survival by QMP/iodotyrosine treatment	147
Table H.1 - Oligonucleotide sequences used for RT-qPCR	148
Table H.2 - N values for each QMP dose/ovary score combination	148
Table K.1 - Pairwise comparisons of food and solution intake by QMP/imidacloprid	
treatments	152
Table K.2 - Pairwise comparisons of brain dopamine levels by QMP/imidacloprid	
treatments	154
Table K.3 – Pairwise comparisons of survival distributions between QMP/imidaclop	prid
treatments	155

List of Abbreviations

- 20E 20-Hydroxyecdysone
- 5-HT serotonin
- 9-HDA (E)-9-hydroxydec-2-onoic acid
- 9-ODA (E)-9-oxodec-2-enoic acid
- ACh Acetylcholine
- AChE Acetylcholine esterase
- CA Corporus allatum
- **CBF** complete bee food
- CI confidence interval
- CLMM cumulative link mixed effects model
- CNS central nervous system
- **CPH** Cox proportional hazards
- DA dopamine
- ETH ecdysis-triggering hormone
- GLMM generalized linear mixed effects model
- HOB methyl p-hydroxybenzoate
- HPLC-FLD high performance liquid chromatography with fluorescent detection
- HR hazard ratio
- HVA 4-hydroxy-3-methoxyphenylethanol
- IMI imidacloprid
- JH juvenile hormone
- LMM linear mixed effects model
- nAChR nicotinic acetylcholine receptor
- NE norepinephrine
- **OA** octopamine
- PBS phosphate buffered saline
- ppb parts per billion
- QE queen equivalent

QE/µL - queen equivalents per microliter

QL – queenless

QMP – Queen mandibular pheromone

QR – queenright

- **RH** relative humidity
- **RPM** revolutions per minute

TA – tyramine

- Vg vitellogenin
- **YP** Yolk protein

Chapter 1

General Introduction

1.1 Eusociality

1.1.1 The success of social living

One of the major transitions in evolution was the shift from solitary individuals to eusocial colonies (Szathmáry & Smith, 1995). Characterised by the occurrence of reproductive and non-reproductive castes within a society (Wilson 1971), eusociality is a phenomenon that has confounded biologists for centuries. Charles Darwin himself struggled to reconcile the evolution of an effectively sterile worker caste with his theory of evolution by natural selection, famously citing eusociality as his 'one special difficulty, which at first appeared to me insuperable, and actually fatal to my whole theory' (Darwin, 1859).

Although various definitions of eusociality exist, the most widely accepted involves three key features that define eusociality: co-operative brood care, reproductive division of labour, and the overlap of generations living together (Wilson, 1975). The success of this life history strategy is demonstrated in the extent to which eusocial insects dominate the insect fauna of the ecosystems in which they live (Wilson, 1990). Most strikingly, although less than 2% of insect species are eusocial, they constitute an estimated 50% of the biomass in some ecosystems (Wilson, 2008).

Eusociality has evolved numerous times in insects, in further testament to the success of this life history strategy (Wilson & Hölldobler, 2005). Eusociality is most common within the Hymenoptera, within which this high level of social organisation has seen up to nine independent origins (Hughes *et al.*, 2008). However, it has also arisen in several other invertebrate orders, including Isoptera (Wilson, 1975), Hemiptera (Stern, 1994), Coleoptera (Smith *et al.*, 2009), Thysanoptera (Gadagkar, 1993), and a sponge-dwelling shrimp (Duffy, 1973). Thanks to naked mole-rats and damaraland mole-rats, there are even examples of eusociality in mammals (Burland *et al.*, 2002; Jarvis, 1981). With this, it is unsurprising that eusociality has drawn substantial scientific interest, amassing decades of research attention.

1.1.2 Inclusive fitness theory

The facet of eusocial biology of particular interest to many researchers is its evolution. In fact, the question of how an effectively sterile caste could evolve under natural selection has been a source of much debate. Two opposing theories outline how eusociality may evolve: Inclusive Fitness (Hamilton, 1964b, 1964a) and Group Selection (Nowak *et al.*, 2010; Wilson & Wilson, 2007).

Group selection theory posits that eusociality evolved through positive selection on groups of individuals due to the advantages of living together co-operatively (Nowak *et al.*, 2010; Wilson & Wilson, 2007). However, the central idea that selection could act at the level of a group has been widely criticized, as such a system would be highly vulnerable to cheating by individuals. On the other hand, Hamilton's theory of inclusive fitness (Hamilton, 1964a, 1964b), also known as "kin selection" (Maynard Smith, 1964), was the first proposed theory and remains the most widely accepted today (West & Gardner, 2013). It is rooted in the idea that an individual can accrue indirect fitness benefits from helping to rear the offspring of a close relation, as this aids the transfer of their genes to the next generation (Hamilton, 1964a, 1964b). This provides a general explanation for the reproductive division of labour that defines eusocial insects, as the intermediate levels of relatedness typically found in insect societies increase the benefits of altruism. With this, workers theoretically benefit from investing in the reproduction of the queen through co-operative brood care, as opposed to investing in their own direct fitness by attempting to reproduce themselves (Hamilton, 1964a, 1964b).

1.1.3 Conflict over reproduction

The inclusive fitness theory provides us a useful framework for understanding the ultimate mechanisms behind the evolution of eusociality, specifically the reproductive division of labour. However, extant eusocial insect societies are still subject to internal conflicts over reproduction (Ratnieks & Reeve, 1992). For instance, workers in a range of

eusocial species may attempt to increase their direct fitness by laying eggs (Bourke, 1988; Hammond & Keller, 2004; Wenseleers *et al.*, 2004), attempting to develop into queens (Bourke & Ratnieks, 1999; Wenseleers *et al.*, 2003), or even overthrowing the existing queen in species without morphologically distinct queens and workers (i.e. ponerine ants; Monnin *et al.*, 2002; Monnin & Ratnieks, 2001). Such conflicts have driven the evolution of numerous mechanisms that constrain the reproductive capabilities of workers relative to queens, known as reproductive constraints (Khila & Abouheif, 2008, 2010). By negating conflict over queen-worker reproduction, reproductive constraints are integral to the maintenance of social harmony within the colony (Ratnieks *et al.*, 2006). An appreciation of the mechanisms of reproductive constraint in social insects is thus fundamental to our understanding of the evolution of eusociality.

1.2 Reproductive constraints

Reproductive constraints constitute any mechanism that reduces the ability of workers to reproduce relative to the queen (Khila & Abouheif, 2010). This encompasses a diverse range of processes, from behavioural – such as the policing of worker-laid eggs (e.g. Wenseleers & Ratnieks, 2006) – to developmental and physiological constraints. Khila & Abouheif (2010) further classified the developmental reproductive constraints of ant species into five sub-categories, termed RC1-RC5 (Table 1.1). Although these sub-categories describe the developmental constraints present in ants, they are widely applicable to a variety of eusocial species.

Reproductive constraint	Mechanism
RC1	Mis-localisation of mRNA
RC2	Quantitative constraint on ovary activity
RC3	Loss of spermatheca
RC4	Reduction of ovariole number
RC5	Complete loss of reproductive organs

Table 1.1 Reproductive constraints. The five classifications of developmental reproductive constraints in ants described by Khila and Abouheif (2010). Reproductive constraints may be reversible (RC1-RC2) or irreversible (RC3-5).



Figure 1.1 Differences in honey bee ovary morphology between castes Differences in honey bee *Apis mellifera* ovary physiology between **A.** Queens, **B.** Queen-right workers (i.e. workers from a hive headed by a queen), and **C.** Queen-less workers (i.e. workers from a hive lacking a queen). Workers develop considerably fewer ovarioles than queens (where ovarioles are individual strings of developing oocytes; **C.** pictures full ovaries and the dissected individual ovarioles), and also lack a spermatheca (not pictured). In the absence of a queen, workers may activate their ovaries and produce male eggs. **C.** shows the four distinct stages of ovary activity, ranging from inactive (0) to possessing mature oocytes (3). Mature oocytes are denoted by asterisks, while arrows mark signs of differentiation (a detailed explanation of each stage is found in Chapter 2). Scale bars indicate 1 mm. Figure provided by Dr Elizabeth J. Duncan and used with permission.

1.2.1 Irreversible constraints

Constraints on worker physiology reduce their ability to reproduce in ways which are both reversible (i.e. RC1-RC2) and irreversible (i.e. RC3-RC5). The most extreme constraint, the complete loss of reproductive organs (RC5), is the only constraint that completely eliminates worker reproduction and is comparatively rare; It is found in only nine out of 283 ant genera (Khila & Abouheif, 2010). Other irreversible developmental constraints merely quantitatively or qualitatively lower the reproductive output of workers. For example, while the loss of the spermatheca (sperm storage organ) prevents hymenopteran workers from laying fertilized female eggs (RC3), they are still able to produce unfertilized eggs destined to become males due to their haplodiploid sex determination system (e.g. honey bees: Snodgrass 1956; some ant species: Khila & Abouheif, 2010). Reductions in ovariole numbers (ovarioles are strings of developing oocytes; Figure 1.1; RC4) are another irreversible developmental constraint which generally accompanies the loss of the spermatheca, arising during embryonic development in ants (Khila & Abouheif, 2010) and larval development in honey bees (Reginato & Cruz-Landim, 2003; Schmidt, Capella & Hartfelder, 2002). Ovariole loss severely reduces, but does not completely eliminate, the reproductive potential of workers relative to queens. In honey bees for instance, the feeding of royal jelly during the larval stage triggers the development of considerably larger ovaries in queens relative to workers, with queen ovaries consisting of 120-200 ovarioles while workers develop as few as 2-12 (Figure 1.1; Snodgrass 1956; Linksvayer *et al.*, 2011).

1.2.2 Reversible constraints

In contrast to these irreversible reductions to worker fecundity, some reproductive constraints confer an element of plasticity in adulthood based on environmental signals. For instance, quantitative constraints on ovary activity (RC2) are not an ultimate constraint, and in certain social conditions, workers have the capacity to increase their ovary activity and begin laying their own eggs (Khila & Abouheif, 2010). This is likely an adaptive developmental mechanism, allowing workers to regulate their reproductive physiology such that they can increase their fitness either indirectly (i.e. through the indirect fitness benefits of engaging in co-operative brood care) or directly (i.e. by activating their ovaries and reproducing themselves) depending on social cues.

The most common situation in which workers may favour their own reproduction is in the absence of a queen. For example, queenless (QL) workers (i.e. workers from a colony in which the queen or dominant reproductive has been lost) have been seen to activate their ovaries in many eusocial species, including the ponerine ant *Harpegnathos saltator* (Peeters *et al.*, 2000; Peeters & Hölldobler, 1995), the bumble bee *Bombus terrestris* (i.e. van Oystaeyen *et al.*, 2014), and the honey bee (Hoover *et al.*, 2003) to name just a few. To this end, queen pheromones play a key role in the mediation of RC2 by signalling the queen's presence and/or suppressing worker reproduction (le Conte & Hefetz, 2008). As with other reproductive constraints, the inhibition of worker reproduction by queen pheromones is an integral mechanism for the maintenance of colony harmony. Understanding how these pheromones have arisen, and the mechanisms by which they inhibit worker reproduction, is therefore central to the evolution of eusociality.

1.3 Queen pheromones

1.3.1 Ancestral Hymenopteran queen pheromones

Across the eusocial Hymenoptera, queens produce pheromones which signal their presence to workers. These queen pheromones show a remarkable degree of conservation across at least three independent origins of eusociality (Figure 1.2). Queens of three species of ants (Holman *et al.*, 2010, 2013; van Oystaeyen *et al.*, 2014), the bumble bee *Bombus terrestris* (Holman, 2014; van Oystaeyen *et al.*, 2014) and the common wasp *Vespula vulgaris* (van Oystaeyen *et al.*, 2014) all produce sterility-inducing



Figure 1.2 Evolutionary history of queen pheromones in eusocial insects demonstrating the structural difference of honey bee queen pheromone (QMP) and termite queen pheromones with those used in other species. Alternately shaded areas represent independent origins of eusociality. Non-volatile cuticular hydrocarbons (peach) are used as conserved queen pheromones in ants, bumble bees and vespine wasps (Holman et al., 2010, 2013; van Oystaeyen et al., 2014). Honey bees (light green) and termites (light blue) use other volatile compounds including keto acids, esters, and alcohols (Hoover et al., 2003; Matsuura et al., 2010; K. N. Slessor et al., 2005). Figure from Oi et al., (2015), used with author's permission.

queen signals in their cuticle consisting of non-volatile long-chain linear and methylbranched saturated hydrocarbons (Figure 1.2). The commonality of these non-volatile hydrocarbon components across diverse taxa indicates that they were likely derived from a signalling system already present in their solitary ancestors. Interestingly, similar long-chain hydrocarbons serve as signals of fertility in solitary species, presumably for the purpose of mate attraction (Oi *et al.*, 2015). Evidence suggests that the production of long-chain hydrocarbons is a direct by-product of ovary development, implicating these signals as "honest" indicators of fertility (i.e. cannot be cheated). In the house fly, for instance, production of the cuticular hydrocarbon sex pheromone (Z)-9-tricosene is directly triggered by ovary activity (Blomquist & Vogt, 2003). Phylogenetic analyses show that queen signals have a high degree of conservation with these fertility signals (Holman *et al.*, 2013; van Oystaeyen *et al.*, 2014), indicating that the majority of eusocial queen pheromones likely have origins as honest signals of fecundity.

1.3.2 Queen mandibular pheromone

One of the earliest discovered and best studied queen pheromones is the honey bee's queen mandibular pheromone (QMP). As the name suggests, QMP is produced not in the queen's cuticle, but in her mandibular glands (Slessor *et al.*, 1988). QMP's principal components differ considerably from the assumed ancestral non-volatile hydrocarbons described above, instead containing a blend of volatile carboxylic acids: (E)-9-oxodec-2-enoic acid (9-ODA) and two enantiomers of (E)-9-hydroxydec-2-onoic acid (9-HDA); two aromatics: metyl p-hydroxybenzoate (HOB) and 4-hydroxy-3-methoxyphenylethanol (HVA); and other minor compounds (Figure 1.3; Keeling *et al.*, 2003; Slessor *et al.*, 1988). This blend of five semiochemicals inhibits the activity of worker ovaries (Hoover *et al.*, 2003). QMP serves further roles in addition to the induction of sterility in workers, including attracting drones during mating flights (Winston & Slessor, 1998), the inhibition of queen rearing (Winston *et al.*, 1990), and the attraction of workers to the queen, known as retinue behaviour (Slessor *et al.*, 1988). As such, QMP is considerably more complex than the cuticular-hydrocarbon based pheromones that predominate the



Figure 1.3 Chemical structures of the five major QMP components: 9-ODA (E-9-oxodec-2enoic acid), both enantiomers of 9-HDA (E-9-hydroxydec-2-onoic acid), HOB (metyl phydroxybenzoate) and HVA (4-hydroxy-3-methoxyphenylethanol)

rest of the eusocial Hymenoptera, in terms of both its chemical complexity and its wide range of functions.

However, QMP is unlikely the only signal used by honey bee queens that effects worker reproduction, as queens from which the mandibular gland has been removed are still able to inhibit ovary activity (Maisonnasse, Alaux, *et al.*, 2010). Recent findings suggest cuticular hydrocarbons may also serve as repressive queen signals in honey bees, as tergal gland esters present on the queen's dorsal cuticle have been shown to inhibit the reproduction of workers in addition to QMP (Princen *et al.*, 2019). The brood pheromones ethyl palminate, methyl linoleate and the brood volatile E- β -ocimene are also inhibitors of worker reproduction (Maisonnasse, Lenoir, *et al.*, 2010; Mohammedi *et al.*, 1998; Traynor *et al.*, 2014), and likely encourage workers to care for the brood. There is now understood to be multiple levels of redundancy in the signals that repress worker reproduction in honey bees (Princen *et al.*, 2019). How this species came to possess such a highly derived and complex pheromone system makes them an interesting model for understanding the evolution of queen pheromones.

1.4 The Honest signal vs Queen control hypotheses

1.4.1 Honest signal hypothesis

Regarding the evolution of queen pheromones, there is debate as to whether these pheromones function as honest signals for the presence of a fecund queen (honest signal hypothesis; Keller & Nonacs, 1993; Monnin, 2006; Seeley, 1985), or as manipulative compounds that chemically enforce sterilization upon workers against their reproductive interests (queen control hypothesis; Fletcher & Ross, 1985; Strauss et al., 2008; Hefetz 2004; Hölldobler & Wilson 1990). Under the honest signal hypothesis, it is thought that workers under the presence of a fecund queen benefit more from rearing the queen's brood (i.e. their siblings) than attempting to reproduce themselves (Keller & Nonacs, 1993). As such, queen pheromones should be selected for their ability to reliably or 'honestly' signal the queen's fecundity (Oi et al., 2015). The prevalence of a conserved class of fertility-linked cuticular hydrocarbons as gueen pheromones lends considerable support to the honest signal hypothesis (Holman et al., 2010, 2013; van Oystaeyen et al., 2014). In addition, even workers commonly begin producing queen-like pheromones as they activate their ovaries, despite this making them more vulnerable to attack by policing workers (i.e. in social wasps; Smith et al., 2009; van Zweden et al., 2014), implicating these signals as uncheatable outcomes of fertility (Holman, 2012; Peso et al., 2015).

Although QMP's chemical composition is considerably removed from the fertility-linked cuticular hydrocarbons used as queen pheromones in other species, evidence points to it also serving as an honest indicator of fecundity. The production of QMP is tightly linked with the queen's reproductive state (Kocher *et al.*, 2009; see Table 1.2). For instance, the quantity and chemical composition of QMP produced differs between virgin and mated queens (Plettner *et al.*, 1997), between laying vs non-laying mated queens (Kocher *et al.*, 2008), with the number of drones she has mated with (Richard *et al.*, 2007), and with semen insemination volume (Niño *et al.*, 2012). A function of QMP as an

QMP component	Functions	Abundance in virgin vs mated
(E)-9-oxodec-2- enoic acid (ODA)	Complete inhibition of worker reproduction (Princen <i>et al.,</i> 2019; Tan <i>et al.,</i> 2015; Butler & Fairey, 1963) Weak retinue response (Complete QMP blend required for full retinue response; Slessor <i>et al.,</i> 1988)	Quantity produced increases from young virgin to 1-year old mated queens (Plettner <i>et al.,</i> 1997) Ratio of ODA to 10-HDA increases with queen age and mating status (Plettner <i>et al.,</i> 1997)
	Does not inhibit queen rearing (Butler & Fairey, 1963)	
(E)-9-hydroxydec-2- onoic acid (9-HDA)	Complete inhibition of worker reproduction (Princen <i>et al.,</i> 2019; Tan <i>et al.,</i> 2015)	Quantity produced increases from young virgin to 1-year old mated queens (Plettner <i>et al.,</i> 1997)
	Weak retinue response (Complete QMP blend required for full retinue response; Slessor <i>et al.</i> , 1988)	Higher ratio of 9-HDA to 10-HDA indicative of "higher quality" queens (Rangel <i>et al.,</i> 2016)
Methyl p- hydroxybenzoate (HOB)	Partial inhibition of worker reproduction (Princen <i>et al.,</i> 2019)	Quantity produced increases from young virgin to 1-year old mated queens (Plettner <i>et al.,</i> 1997)
	Weak retinue response (Complete QMP blend required for full retinue response; Slessor <i>et al.,</i> 1988)	
4-hydroxy-3- methoxyphenyletha nol (HVA)	Partial inhibition of worker reproduction (Princen <i>et al.,</i> 2019)	Produced only by 1-year old mated but not young virgin queens (Plettner <i>et al.,</i> 1997)
	Weak retinue response (Complete QMP blend required for full retinue response; Slessor <i>et al.,</i> 1988)	

Table 1.2 Principal QMP components and their functions. A summary of what is currently known about the functions of each of the principal QMP components in isolation and how their abundance differs between virgin and mated queens

honest indicator of fecundity even in this highly derived pheromone provides additional support for the queen signal hypothesis.

1.4.2 Queen control hypothesis – is QMP a suppressive agent?

The queen control hypothesis, on the other hand, suggests that queen pheromones act as suppressive agents and chemically sterilize workers against their best interests (Fletcher & Ross, 1985; Strauss *et al.*, 2008; Hefetz 2004; Hölldobler & Wilson 1990). Under this model, it is predicted that queen pheromones should evolve rapidly owing to an evolutionary 'arms race' between queens a workers, whereby workers are continually selected to evade repression and queens are under pressure to evolve novel repressive agents (Heinze & D'Ettorre, 2009). The remarkably conserved class of cuticular hydrocarbon-based queen pheromones do not support this (van Oystaeyen *et al.*, 2014). However, the complexity of honey bee QMP (Slessor *et al.*, 1990) and the functional redundancy of its components (Princen *et al.*, 2019) is in line with the predicted 'arms race' of the queen control hypothesis. Could the highly derived QMP have taken on a novel function as a suppressive agent?

There is evidence pointing towards QMP chemically suppressing reproduction. Most convincing is the fact that QMP has been found to repress reproduction across a broad phylogeny of species. For instance, honey bee QMP represses oogenesis in virgin *Drosophila melanogaster* females (Camiletti *et al.*, 2013), despite these species being ~340 million years diverged (Misof *et al.*, 2014). Reproduction is also repressed by QMP in a species of ant (Carlisle & Butler, 1956), a house fly (Nayar, 1963), a termite (Hrdy *et al.*, 1960), and even a prawn (Carlisle & Butler, 1956). There is clearly no adaptive benefit in being responsive to honey bee QMP in these species, as is predicted for worker responses under the queen signal hypothesis. Instead, it seems likely that QMP may have evolved to chemically repress worker reproduction by targeting pathways involved in regulating reproduction already present in solitary ancestors. This modulation of reproductive pathways appears to be a feature unique to QMP, as ancestral

hymenopteran queen pheromones do not share these broad phylogenetic repressive effects (Lovegrove *et al.*, 2019). Despite QMP being one of the most highly studied queen pheromones, we still lack a complete understanding of the specific molecular pathways involved in its repression of ovary activity. Improving our understanding of the mechanisms underlying QMP's suppression of reproduction is therefore key to addressing whether this pheromone serves as an agent of sterilization, as predicted by the queen control hypothesis, in addition to being an honest signal of fecundity.

1.5 How does QMP inhibit reproduction?

1.5.1 Ovarian response to QMP – Notch signalling

Although the complete molecular signalling pathway of QMP's induction of sterility has not yet been fully elucidated, we have a good understanding of the ovarian response to QMP. In the presence of QMP, ovary development is understood to be inhibited by active Notch signalling, which acts in the germarium to block oogenesis (Figure 1.4; Duncan *et al.*, 2016). Oocytes that do develop are then possibly destroyed via programmed cell death (Ronai *et al.*, 2015). Active Notch signalling is triggered by the binding of the ligands Delta and Serrate to the Notch receptor, which is cleaved by the γ -secretase complex. This allows the translocation of the Notch intracellular domain (NICD) to the nucleus, leading to the recruitment of histone acetyltransferases (HATs) which enable the transcription of the enhancer of split complex E(spl)-C and other genes (Bray, 2006). This is an ancient and pleiotropic signalling pathway, with functions in the development of tissues and organs in a diverse range of species (Guruharsha *et al.*, 2012).

Importantly, Notch signalling has been demonstrated to be responsive to environmental cues (Hsu & Drummond-Barbosa, 2011). For instance, in *D. melanogaster*, it is involved in regulating reproduction in response to nutrient availability (Bonfini *et al.*, 2015). Duncan *et al.*, (2016)'s finding that QMP activates Notch signalling to inhibit oogenesis in workers indicates that this ancient, environmentally responsive signalling pathway has been co-opted into regulating reproductive constraint in honey bees. However, there



Figure 1.4 Schematic of Notch signalling's inhibition of oogenesis in response to QMP. In the presence of QMP, Notch signalling is active in the germaria (region of the ovaries where germ cells are specified) and egg production is blocked **(Left)**. In the absence of QMP, there is no active Notch signalling in the germaria, allowing presumptive oocytes to mature into fully developed eggs **(Right)**. Figure modified from Duncan et al., (2016), used with permission.

are still gaps in our knowledge as to how QMP triggers the activation of Notch signalling. In particular, we lack understanding of the mechanism linking the detection of QMP by workers with the transmission of this signal to the ovary. To understand how QMP may exert its effects on ovary physiology, we must look to the neuroendocrine signalling pathways governing reproduction in honey bees.

1.5.2 Neuroendocrine control of reproduction

In all animals, the neuroendocrine system links the regulation of reproduction with environmental cues. In insects, the core neuroendocrine system incorporates nutrient sensing (e.g. insulin signalling), hormones (e.g. juvenile hormone and 20hydroxyecdysone) and biogenic amines (e.g. dopamine and octopamine) (Figure 1.5; Reviewed by Knapp *et al.*, 2022). Much of our understanding of how these neuroendocrine signalling channels regulate reproduction and oogenesis comes from research carried out in D. melanogaster. In D. melanogaster, vitellogenesis (i.e. the



Figure 1.5 Major neuroendocrine signalling pathways involved in regulating reproduction in response to environmental cues in *Drosophila melanogaster*. Figure from Knapp, Norman *et al.*, (2022), In press.

accumulation of yolk into developing oocytes) is regulated by juvenile hormone (JH) and the insect steroid hormones ecdysteroids. These hormones act together to stimulate yolk protein (YP) synthesis in the fat body and uptake of YP by the ovary (Jowett & Postlethwait, 1980; Postlethwait & Handler, 1979). The synthesis of JH occurs in the corporus allatum (CA), and is triggered by stimulation of the endocrine Inka cells by 20hydroxyecysone (20E) to express and secrete ecdysis-triggering hormone (ETH). This regulatory pathway is responsive to environmental stimuli. For instance, vitellogenesis, ovulation and oviposition are triggered post-mating by the seminal substance sex peptide (SP), which acts via neuropeptide allatostatin-C (AstC)-producing neurons which gate the biosynthesis of JH (Zhang *et al.*, 2022).

While the regulation of oogenesis is well-understood in D. melanogaster, a shortage of empirical evidence from other species makes it difficult to understand how representative the reproductive neuroendocrine network of this model organism is to other insect species. For instance, an important distinction between neuroendocondrine regulation of oogenesis in honey bees vs D. melanogaster is the loss of JH's gonadotropic function. In honey bees, JH titres have been decoupled from the production of vitellogenin, and there is no evidence to suggest a role of JH in the regulation of reproduction (Rodrigues & Flatt, 2016). How the neuroendocrine network has been rewired to allow JH to lose this function in honey bees is not completely understood. In particular, it is not clear what may have taken on JH's role as the principal gonadotropin. This leads to interest in alternative signalling molecules involved in regulating reproduction in honey bees, as these would be likely targets for QMP to induce sterility in workers.

1.5.3 Biogenic amines

Among their diverse suite of functions, biogenic amines such as dopamine, serotonin, octopamine and tyramine are also involved in mediating reproduction in insects. These compounds coordinate a range of behavioural and physiological processes though their functions as neurotransmitters, neuromodulators and circulating neurohormones (reviewed by Sasaki & Harano, 2010). Many biogenic amines have overlapping properties in insects, functioning as neurotransmitters in localised interneuronal signalling and as neurohormones when released into the hemolymph and transported to target peripheral tissues via the circulatory system (Blenau & Baumann, 2001). The neural and endocrine functions of these compounds presumably also enable the mediation of insect behaviour and physiology over differing timescales. It is well known that neural connections produce rapid-fire electrochemical signals occurring on the millisecond scale, while the endocrine system functions over a longer time frame. For instance, in *Drosophila virilis*, the comparatively rapid neuroendocrine stress response produces an elevation in dopamine, tyramine and octopamine levels 15 minutes after exposure to a stressor (Hirashima *et al.* 2000).

The general functions of biogenic amines in insects have been well documented. Dopamine has roles both as a slow neurohormone and fast neurotransmitter in insects. In flies, for instance, dopamine neurons co-ordinate processes such as learning and memory, sexual drive, sleep, and hunger (Ichinose et al., 2017). Meanwhile, octopamine and its precursor tyramine are the invertebrate equivalents of vertebrate adrenergic transmitters, acting antagonistically to regulate the "fight or flight" response (Roeder, 2005). In invertebrate peripheral nervous systems, octopamine also regulates flight muscle activity and energy metabolism, and can modulate peripheral and sense organs such as the fat body, oviduct and hemocytes (Goosey & Candy, 1982; Orchard *et al.* 1982; Adamo *et al.* 1995). In the central nervous system, octopamine is also involved in regulating motivation and initiation, sensory inputs, and maintaining complex behaviours including learning and memory (Farooqui, 2007).

In addition to these general functions, biogenic amines are involved in regulating reproduction in many insect secies primarily via the neuroendocrine system. These biogenic amines indirectly affect reproduction by regulating JH titres in Drosophila species (e.g. Gruntenko et al., 2005, 2007), but can also directly regulate oogenesis and oviposition (Monastirioti, 2003; White et al., 2021). Correlative evidence suggests that biogenic amines may regulate reproduction in a range of social Hymenopterans. In Polistes paper wasps, brain levels of dopamine, serotonin and their metabolites are positively correlated with ovarian width in workers with developed ovaries (Sasaki et al., 2007). An increase in serotonin and octopamine in the brain is associated with the shift from pre-reproductive to reproductive stages in the facultatively social carpenter bee, Certina calcarata (Cook et al., 2019). Brain dopamine levels correlate positively with terminal oocyte length in bumble bees (Bloch et al., 2000; Sasaki et al., 2017) and with ovarian activity in the secondarily queenless (QL) ant Harpegnathos saltator (Penick et al., 2014) and QL honey bee workers (Sasaki & Nagao, 2001). Manipulative studies are less common, but those that have been carried out also implicate biogenic amines, in particular dopamine, in regulating reproduction. For example, inhibiting dopamine synthesis reduced rates of egg-laying and numbers of chorionated oocytes in the fire ant Solenopsis invicta (Boulay et al., 2001). Further, dopamine application has been found to enhance ovary activity in a derived QL Diacamma ant (Okada et al., 2015) and in QL honey bees (Dombroski et al., 2003). This implicates biogenic amines as being important for the regulation of reproduction across a broad range of eusocial Hymenopterans.



Figure 1.6 One of QMP's core components, HVA, is structurally very similar to dopamine. Figure modified from (Beggs et al., 2007)

Given this large body of evidence linking biogenic amines with ovary development, studies have begun to address whether this may be a target pathway for QMP in its regulation of reproductive constraint in honey bees (Figure 1.7). Circumstantial evidence implicates dopamine in particular as a mediator of QMP's repression of ovary activity. For instance, brain dopamine levels are lower in QR relative to QL workers (Harris & Woodring, 1995) and positively correlate with ovary activity in QL workers (Sasaki & Nagao, 2001). Importantly, there is also evidence that QMP can directly modulate dopamine levels; One of QMP's five principal components HVA has been shown to reduce dopamine titres in the brains of workers (Beggs *et al.*, 2007). The exact mechanism by which this occurs is not fully understood. However, HVA shares remarkable structural similarity with dopamine (Figure 1.6) and can selectively bind to the dopamine receptor AmDop3 (Beggs & Mercer, 2009), which may underly its capacity to directly modulate dopamine signalling. HVA's effects on the dopaminergic system have been directly linked with QMP's regulation of aversive learning and memory in young workers (Vergoz et al., 2007). However, it is unclear whether these modulations also affect worker ovarian development (Figure 1.7). Considering the suspected role of dopamine as a gonadotropin in a range of social Hymenopterans (i.e. as discussed above), HVA's suppression of dopamine could plausibly elicit sterilizing effects in honey bee workers. This raises the intriguing possibility that as well as being an honest indicator of fecundity (Kocher et al., 2009), QMP is capable of inhibiting the reproduction of workers by directly manipulating signalling molecules with likely pre-existing roles in regulating reproduction: biogenic amines. This would add to existing evidence for an



Figure 1.7 Diagram showing the causal relationships among QMP, DA/ other biogenic amines and ovary activation. Dotted lines indicate an uncertain causal relationship, while solid lines indicate a proven causal relationship.

additional function of QMP as an agent of reproductive control produced by honey bee queens.

1.6 Research Aims

By investigating the hypothesis that QMP exploits aminergic signalling pathways to cause sterility in workers, this research aims to improve our understanding of the mechanisms by which this highly derived queen pheromone induces reproductive constraint in honey bees. Empirical testing of if, and how, QMP chemically suppresses worker reproduction is central to our understanding of how the pheromones mediating reproductive division of labour in eusocial insect societies function: as agents of control, honest signals of fertility, or a combination of both.

In Chapter three, I attempt to elucidate how changes in biogenic amines in the brain are linked with ovary development by investigating a possible pathway by which they could

18

signal to the ovary. Specifically, I test the hypothesis that biogenic amines function as circulatory hormones in honey bees that are capable of interacting directly with ovarian tissue. In addition, the use of laboratory microcolonies of workers as a proxy for studying whole hives is established.

In Chapter four, I directly test the causal relationship between dopamine titres and ovary development through a series of manipulative experiments using dopamine supplementation and inhibition on laboratory microcolonies. This is to test the hypothesis that QMP's modulations to dopamine levels are a driver of worker reproductive constraint.

Given the idea that QMP acts as a suppressive agent to induce sterility in her workers, I address the inherent hypocrisy of how the queen's reproduction is not inhibited by her own QMP in Chapter five. I test one possible mechanism by which QMP could hypothetically serve as a repressive agent in workers but not queens, and speculate at other mechanisms that could be investigated further.

In Chapter six, I apply my understanding of QMP's effects on biogenic amine signalling pathways and investigate how resilient this delicately balanced system is to external environmental stressors. I test the effects of a commonly used pesticide, imidacloprid, on worker responses to QMP at molecular, physiological and behavioural levels, and hypothesize implications for whole-colony function.

Finally, in Chapter seven I bring together my findings with the wider literature and speculate as to the specific mechanisms by which QMP may regulate reproduction and identify key avenues for future research.

Chapter 2

General Methods

2.1 Honey bee maintenance

2.1.1 Laboratory microcolonies

Apis mellifera colonies were kept according to standard beekeeping practices in British National hives at the University of Leeds School of Biology Research Apiary. To set up laboratory microcolonies, frames of emerging brood were collected from three source hives and incubated overnight at 35°C. The following day, newly emerged workers were transferred into metal cages with a glass sliding door and holes for insertion of food caps and a water tube (10 x 10 x 5.5 cm, Small Life Supplies UK; Figure 2.1). 80-100 bees were randomly allocated to each cage, and cages were randomly assigned to treatments. Cages were maintained in the dark at 35°C and 20-40% relative humidity (RH). Caged experiments were limited to 10 days in duration. After this point there were large dropoffs in survival as bees reached maximum gut capacity due to their inability to exhibit natural defecation behaviours (personal observation).



Figure 2.1 Honey bee cages Metal cages used to house honey bees in laboratory experiments, showing **A.** falcon tube used to hold water or treatment solutions, **B.** food cap and aeration holes. Photograph taken by Jessica Bouwer (MSc student) and used with permission.

Cages were monitored daily for dead bees, which were recorded and discarded. All cages were fed complete bee food (CBF), a high protein diet formulated to permit ovary activity (Duncan *et al.*, 2016) consisting of: 20g pollen, 52g sucrose, 18.8g brewer's yeast and 9.2g lactalbumin. CBF was prepared in large batches and stored at -20°C. Cages received 2 g of CBF mixed with honey to form a paste each day. Food and water (or treatment solution) intake were recorded daily for each cage to determine whether experimental treatments were associated with changes in appetite or treatment solution drinking rates that could confound or exacerbate the results of any given experiment. Mean food and water consumed per bee per day was calculated by subtracting the weight of food/water out by that of the food/water given the previous day and dividing by the current number of bees in the cage. In the rare occurrence of obvious food and water tube leakages, these measurements were excluded from the dataset.

2.1.2 QMP administration

Synthetic QMP used in laboratory experiments (supplied by Intko Supply Ltd, Canada) contained the five major QMP components: (E)-9-oxodec-2-enoic acid (9-ODA), both enantiomers of (E)-9-hydroxydec-2-onoic acid (9-HDA), methyl p-hydroxybenzoate (HOB) and 4-hydroxy-3-methoxyphenylethanol (HVA). QMP is quantified in 'queen equivalents' (QE), where 1 QE is the mean quantity of QMP found in the mandibular glands of a laying queen. 1 QE consists of 150 μ g 9-ODA, 55 μ g 9-HDA, 13 μ g HOB, and 1.5 μ g HVA (Slessor *et al.*, 1988). QMP was dissolved in ethanol at a concentration of 0.01 QE/ μ L (queen equivalents per microliter) and stored at -20°C. queenright (QR) laboratory microcolonies were exposed to QMP by pipetting a 10 μ L droplet of 0.01 QE/ μ L onto a glass slide and placing this slide within the cage. This amounts to a QMP exposure of 0.1 QE per cage per day (0.01 QE/ μ L x 10 μ L). Laboratory microcolonies maintained in queenless (QL) condition received 10 μ L ethanol as a solvent control each day. Slides were replaced daily. QR and QL microcolonies were maintained in separate identical incubators to avoid cross-contamination of QMP components.

2.2 Ovary dissection and scoring

To assess ovary activity in caged experiments, bee ovaries were dissected and imaged on day 10. Cages of bees were incubated at 4°C prior to dissection to cold-anesthetise them. Images of dissected pairs of ovaries were captured at 2.5 x magnification using GXCaptureT software and a GXCAM-U3 camera mounted on a GXMXTL3101 stereomicroscope.

Images were later scored for activity level using a modified Hess scale as in previous work (Hess, 1942; Duncan *et al.*, 2016). Figure 2.2 depicts the key morphological features associated with each stage of ovary development. Ovaries that are thin and lacking signs of cell differentiation are scored as a zero, ovaries that are slightly thickened with signs of differentiated cells but no deposition of yolk are scored as a one, ovaries with clearly defined oocytes and yolk deposition are scored as a two, and ovaries with at least one fully mature oocyte are scored as a three (Figure 2.2). Where possible, scoring was carried out by three independent scorers who had been blinded to treatment. The final



Figure 2.2 Morphology of ovaries and single ovarioles in each ovary score category showing the four distinct stages of ovary activity. Mature oocytes are denoted by asterisks, while arrows mark signs of differentiation. Scale bars indicate 1 mm. Figure provided by Dr Elizabeth J. Duncan and used with permission.
score used was the mode of each three scores. In the rare occurrence that all scorers disagreed, the score of the most experienced scorer was used. If only one scorer was available, images were scored by one person (myself) using 'blinder' software to blind the scorer to treatment and randomise the order of images (Cothren *et al.*, 2018).

2.3 Measurement of biogenic amines using HPLC-FLD

2.3.1 Sample collection

Brain, ovary and hemolymph samples were often collected from workers following a range of experimental treatments for analysis of biogenic amine content. Workers were anesthetised by incubating whole microcolonies at 4°C prior to sample collection. 1 μ L hemolymph was extracted from the abdomen of each bee using a pulled glass capillary needle. Whole brains were dissected, with care taken to remove hypopharyngeal glands, optic and antennal lobes and any connective tissue to avoid contamination of brains with biogenic amines from other sources. Ovary pairs were dissected and scored immediately, as capturing images for later blind-scoring would have delayed tissue collection and resulted in degradation of biogenic amines. All samples were immediately snap frozen in liquid nitrogen then stored at -80°C until later analysis of biogenic amine content using HPLC-FLD.

2.3.2 Preparation of samples and standards

Samples were prepared for HPLC-FLD analysis as follows. Brain and ovary samples were homogenized on ice in 150 μ L ice-cold 0.1M perchloric acid (PCA) for one minute using a micropestle fixed to a pellet-pestle cordless motor (Kimble-Chase). 1 μ L Hemolymph samples were added to 149 μ L ice-cold 0.1M PCA (bringing the total sample volume up to 150 μ L) and vortexed for 30 seconds. On the rare occasion that under 1 μ L hemolymph had been collected, the volume of 0.1M PCA added was increased as required to bring the total volume up to 150 μ L, and adjustments to final biogenic amine content calculations were made to account for this. All samples were centrifuged at 13,000 RPM at 4°C for 30 minutes to remove cellular debris. The supernatant was purified of small

particulates prior to HPLC-FLD by centrifuging for a further five minutes through a filter column (0.45 μ m, Thermo Scientific), then transferred into a 250 μ L HPLC micro vial (Agilent Technologies) and refrigerated until HPLC-FLD analysis later that day. Samples were prepared simultaneously in batches of 12 or 24. The preparation batch of each sample was noted and included as a random effect during data analysis.

External standards of norepinephrine, dopamine, octopamine, serotonin and tyramine were prepared at 0.025, 1, 10 and 50 μ M concentrations. Standards were prepared in ice-cold 0.1M PCA from stocks stored at -80°C on each day of sample preparation and refrigerated until same-day HPLC-FLD analysis alongside samples.

2.3.3 HPLC-FLD system

The simultaneous determination of biogenic amines in biological samples was achieved using high performance liquid chromatography separation with fluorescent detection (HPLC-FLD) and pre-column derivatization (Figure 2.3). The HPLC-FLD instrument was based in the School of Chemistry HPLC Facility at the University of Leeds and operated by Jeanine Williams. The instrument was an Agilent 1290 Infinity II UHPLC system consisting of a multisampler (P.N. G7167B), 1290 high speed pump (P.N. G7120A), and 1290 FLD spectra (P.N. G1321B). The column used for separation of biogenic amines by their retention times was an Agilent InfinityLab Poroshell 120 EC-C18 (2.1 x 50 mm, 1.9 um, P.N. 699675-902) maintained at 40°C \pm 0.8°C in a 1290 MCT column compartment (P.N. G7116B).

Mobile phase A contained 10 mM Na₂HPO₄, 10 mM Na₂B₄O₇ and 5 mM NaN₃, adjusted to pH 8.2 using concentrated HCl and filtered and degassed under vacuum (Thermo Scientific Nalgene rapid-flow bottle filter, pore size 0.2 µm). Mobile phase B contained acetonitrile: methanol: water in a ratio of 45:45:10 (v:v:v). The injection diluent was prepared by adding 0.4 mL concentrated H₃PO₄ to 100 mL of mobile phase A. OPA and Borate buffer were used as derivatization reagents and were supplied as ready-made solutions by Agilent (Borate buffer: 0.4 M in water, pH 10.2, 100 mL, P.N. 5061-3339; OPA: 10mg/mL in 0.4M borate buffer and 3-mercaptoproprionic acid, supplied in 1mL



Figure 2.3 HPLC-FLD chromatograms showing the simultaneous determination of the

biogenic amines norepinephrine (NE), dopamine (DA), octopamine (OA), serotonin (5-HT), and tyramine (TA) in **A**) a 10 μ M mix of standards and **B**) a sample of honey bee hemolymph (prepared as described in Section 2.3.2).

ampoules, P.N. 5061-3335). The injection diluent and OPA derivatization reagent were stored at 4°C. Details of the injection program used can be found in Table 2.1. Mobile phase A and B were used in gradient as described in Table 2.2. The flow rate was kept constant at 0.5 mL/min.

Step	Action
1.	Draw 1.5 μL from Borate buffer vial
2.	Draw 2.5 µL from sample vial
3.	Mix $4\mu L$ in air default speed 5 times
4.	Wait 0.2-minutes
5.	Draw 1 µL from OPA vial
6.	Mix 5 μ L in air, 10 times default speed
7.	Draw 5 μL from injection diluent vial
8.	Mix 10 μL in air default speed 8 times
9.	Inject
10.	Wait 0.1-minute
11.	Valve bypass

Table 2.1 Details of HPLC-FLD injection program

Time (min)	% A	% B
0	98	2
0.5	57	33
3	5	95
3.1	0	100

Table 2.2 Details of the mobile phase gradient used for HPLC-FLD

Detection of separated biogenic amines was achieved using fluorescence detector settings of 345 nm (excitation wavelength) and 450 nm (emission wavelength). Chromatograms produced from detector signals over three minute sample run durations were recorded, integrated and analysed using Agilent OpenLab CDS software, ChemStation Edition (Rev C.01.10 [30]) with Agilent OpenLab Intelligent Reporting (5.0.0.352). Standard curves were generated for each biogenic amine by plotting their peak area at each concentration and fitting a linear regression line (see Appendix A, Figure A.1). Biogenic amine concentrations in samples were quantified from their peak areas using the equations of the lines fitted to the standard curves. Brain, ovary and

hemolymph biogenic amine levels levels are expressed as pmol/brain, pmol/ovary and pmol/µL respectively.

2.4 Data analysis

All data analysis was carried out in R version 4.1.2 (R Core Team, 2021). Paired T-tests were carried out using the *rstatix* package (Kassambara, 2021). Linear mixed effects models (LMMs) and generalised linear mixed effects models (GLMMs) were built using the package *lme4* (Bates *et al.*, 2015). The packages *survival* and *coxme* were used to construct Cox proportional hazards models (CPH) with mixed effects (Therneau, 2020; Therneau, 2021), and cumulative link mixed models (CLMMs) were carried out using the package *ordinal* (Christensen, 2019). *Post-hoc* testing was carried out on models using the package *emmeans* (Lenth, 2022). Graphs were prepared using *ggplot2* (Wickham, 2016) and *ggpubr* (Kassambra, 2020).

Chapter 3

Transmission of QMP's signal to the ovary

3.1 Introduction

In honey bee hives, queen mandibular pheromone (QMP) is one of the major factors inhibiting worker reproduction (Hoover *et al.*, 2003). In the presence of QMP, worker ovarian development is actively inhibited by Notch signalling (Duncan *et al.*, 2016), and possibly apoptosis of oocytes that do develop (Ronai *et al.*, 2015). In the absence of QMP worker ovaries are completely remodelled, oogenesis is initiated (Duncan *et al.*, 2016, 2020), and workers lay haploid eggs destined to become drones or male bees (Jay, 1968). Despite decades of research, the processes acting upstream of these mechanisms in the ovary and their integration with the signal from QMP are still uncertain (Section 1.5.2; Figure 3.1).



Figure 3.1 Transmission of QMP's signal to the ovary. Honey bee reproduction in workers is repressed, at least in part, by QMP. Workers are presumably exposed to QMP via the antennal lobes or gustatory receptors, and this signal is transmitted to the ovary to repress oogenesis. How QMP's signal is transmitted to the ovary represents a key gap in our knowledge.

Secreted from the queen's mandibular gland, QMP is picked up by young workers attending the queen and spread among colony members through worker-worker contact and trophallaxis (Naumann *et al.*, 1991; Watmough, 1997). QMP's signal is then transmitted to the ovary, where active Notch signalling inhibits ovary activity by blocking oogenesis (Duncan *et al.*, 2016). Although the processes linking the detection of QMP with this ovarian response have not been fully elucidated, biogenic amines have been widely implicated in playing a role (refer section 1.5.3, General Introduction).

Through their diverse functions as neurotransmitters, neuromodulators and neurohormones (Evans, 1980), biogenic amines such as dopamine, octopamine, tyramine and serotonin coordinate a range of behavioural and physiological processes in insects (reviewed by Sasaki & Harano, 2010). In eusocial insects, it has been proposed that biogenic amines have been co-opted into regulating elements of division of labour and control of worker reproduction (reviewed by Sasaki *et al.*, 2021).

In honey bees, octopamine is involved in coordinating task polyethism by mediating the behavioural maturation from nurse to forager (Schulz *et al.*, 2002; Schulz & Robinson, 2001; Wagener-Hulme *et al.*, 1999), while dopamine and tyramine are thought to be involved in regulating ovarian development (Linn *et al.*, 1994; Salomon *et al.*, 2012; Ken Sasaki & Harano, 2007). In queen-less (QL) workers, brain dopamine levels correlate positively with ovary developmental stage (Harris & Woodring, 1995; Sasaki & Nagao, 2001), and supplementary feeding of dopamine (Dombriski *et al.*, 2003) and tyramine (Salomon *et al.*, 2012) to QL worker honey bees further enhances ovary activity levels. Importantly, QMP also directly depresses brain dopamine through the component HVA (Beggs *et al.*, 2007; Figure 1.6). Exposure to HVA alone can reproduce some of QMP's inhibitory effects on worker reproduction (Princen *et al.*, 2019), implying there may be a link between dopamine and the regulation of ovarian development.

Little is known about the processes linking changes in brain biogenic amine titres with the mechanisms constraining oogenesis within the ovary (Notch signalling: Duncan *et al.* 2016; Apoptosis: Ronai *et al.* 2015). Crucially, multiple biogenic amine receptors are expressed in the ovarian tissue, including the dopamine receptors AmDop1 and AmDop3, the octopamine receptor AmOA1 and the serotonin receptor Am5-HT7 (Vergoz et al., 2012), implying that direct interaction of these biogenic amines with the ovary is possible. However, the mode of transportation of biogenic amines to the ovary has not been investigated. Given that insects have an open circulatory system, peripheral organs would be directly exposed to any compounds present in the hemolymph. Biogenic amines are present in the hemolymph of a number of insect species. For example, several biogenic amines, including dopamine and octopamine, have been detected in the hemolymph of male cabbage looper moths (Linn *et al.*, 1994), and hemolymph dopamine titres in the silkworm *Bombyx mori* mediate the onset of reproductive diapause in adulthood (Noguchi & Hayakawa, 2001). In honey bee males, hemolymph dopamine titres change in parallel with brain dopamine titres, and hemolymph dopamine is thought to interact directly with aminergic receptors on the testes to modulate mating flights and copulation (Akasaka et al., 2010; Sasaki & Watanabe, 2022). However, little is known concerning biogenic amines in honey bee worker hemolymph, and whether they are altered by QMP in parallel to the brain.

Here, it is hypothesised that biogenic amines act as QMP-responsive circulating neurohormones in the hemolymph and interact directly with biogenic amine receptors in the ovary to mediate reproductive constraint. To test this hypothesis, levels of dopamine, octopamine and serotonin in the hemolymph and ovaries of workers from experimental queen-less (QL) and queen-right (QR) hives were measured. To understand the interplay between hemolymph biogenic amine titres and previously reported effects of QMP on brain biogenic amines (Beggs *et al.*, 2007; Harris & Woodring, 1995), brain titres were also examined. This chapter aims to bridge the gap in our understanding of the link between QMP's effects on biogenic amines in the brain with its command of worker ovarian physiology.

3.2 Methods

3.2.1 Generation and maintenance of QR and QL hives

A. mellifera colonies were kept according to standard beekeeping practices in poly nucleus hives at the University of Leeds School of Biology Research Apiary. Genetically related queenless (QL) sister hives were generated from queenright (QR) hives in July-September 2020 by transferring frames of emerging brood into a single poly nucleus box. To prevent workers rearing a new queen from any queen-laid eggs from the brood frames, QL hives were monitored bi-weekly for queen cells, which were destroyed. Workers were routinely sampled from QR and QL hives to assess levels of ovary activity as further confirmation of queen-status of the colonies. It was expected that no workers in QR hives would have active ovaries (where ovaries are considered 'active' at scores of 2 or above; Figure 2.2), while in QL hives around 24-30% of workers will activate their ovaries (Jay, 1968; Miller & Ratnieks, 2001).

3.2.2 Mark-release-recapture experiment

Workers sampled for tissue collection from QL and QR hives were age-matched to control for age-related variations in biogenic amine titres (Harris & Woodring, 1992). Age-matching was achieved using mark-release-recapture. Frames of emerging brood were collected from three source hives and incubated at 35°C. All bees emerging within 24 hours were marked with a coloured dot on their thorax using a POSCA non-toxic paint marking pen, then returned to either the experimental QL or QR hive (Figure 3.2). With a view to resample equal numbers of workers with each ovary score from experimental hives, workers were returned to QR and QL hives in a ratio of approximately 1:3 respectively, as QL workers display substantial variation in ovary score as some bees readily activate their ovaries in response to queen absence (Miller & Ratnieks, 2001). Three marking trial replicates were carried out between August-September 2020, with a different paint colour used on each day. In trial one, 147 and 82 marked bees were released in QL and QR hives respectively. In trial two, respective numbers released were 267 and 123, and in trial three, 304 and 150.



Figure 3.2 Paint-marked bees Image of paint marked bees being released into an experimental hive

Marked bees were resampled from hives at two timepoints after release: 10 and 21 days. Day 10 was selected as a time-point as a direct age-based comparison with laboratory experiments, which are terminated on day 10 (Section 2.1.1, General Methods). Additional workers were sampled on day 21 as ovary activation was observed to be slower in the hive compared with the laboratory, possibly due to differences in nutrition (caged workers are fed a high protein diet specifically formulated to permit ovary activity; Refer section 2.1.1, General Methods). This second later timepoint allowed sufficient time for workers to reach later states of ovary maturity (i.e. scores of 2 and above; Koudjil & Doumandji, 2008) and enable physiologically similar comparisons with caged honey bee workers.

Brain, ovary and hemolymph samples were collected from workers captured from each time point for later measurement of biogenic amines as described in Section 2.3.1. Samples were prepared as described in Section 2.3.2 and HPLC-FLD was used to quantify

octopamine, dopamine and serotonin as described in Section 2.3.3. Numbers of workers sampled at each time point in each replicate is found in table 3.1.

3.2.3 Establishing laboratory microcolonies

QR and QL microcolonies were setup in cages in the laboratory to; 1. Isolate QMP's effects from other pheromones and behaviours in the hive and 2. Validate the laboratory system as a model for the hive. Microcolonies of 100 workers were set up and maintained in cages as described in section 2.1.1. Microcolonies were maintained in QR condition (i.e. with synthetic QMP) or QL condition (i.e. with solvent control) as described in section 2.1.2. Three replicates of each microcolony were set up in September 2021 within three days of each other. On day 10, brain and hemolymph samples were collected for later analysis of biogenic amine content from ten randomly selected bees per cage as described in Section 2.3.1. Samples were prepared as described in Section 2.3.2 and HPLC-FLD was used to quantify octopamine, dopamine and serotonin as described in Section 2.3.3.

To validate the efficacy of synthetic QMP's inhibition of reproduction in QR microcolonies and assess ovary activation rates in QL microcolonies, the ovaries of 20-30 randomly selected bees from each cage were also dissected on day 10. Dissected ovaries were imaged and scored blinded to treatment as described in Section 2.2.

3.2.4 Data analysis

Data analysis was carried out in R. Details of packages used for analysis are found in section 2.4.

Differences in ovary activity between QR and QL workers were analysed using Pearson's Chi Squared tests. In bees resampled from the hive for tissue collection, ovary activity was compared across four conditions: QR day 10, QL day 10, QR day 21 and QL day 21. In the laboratory experiment, ovary scores were compared between QR and QL microcolonies at day 10.

Levels of the biogenic amines dopamine, octopamine and serotonin in tissues of workers sampled from the hive were each analysed using linear mixed-effects models (LMMs) or generalized linear mixed-effects models (GLMMs). LMMs were fitted for each biogenic amine and each tissue, with time-point (ie. day 10 or 21), colony status (i.e. QR or QL) and their interaction as fixed effects and marking replicate and HPLC-FLD analysis batch as random effects. Normality and homoscedasticity of model residuals were tested by visual examination of qqplots, histograms and predicted vs fitted values. If these assumptions were not met, GLMMs with an inverse link function were fitted, the residuals of which met the assumptions of normality and homoscedasticity (i.e. GLMMs were used for analysis of brain dopamine, brain octopamine, hemolymph octopamine and ovary dopamine). P-values for fixed effects were obtained by comparing the likelihood ratio of the maximal model to that of the model without the fixed effect of interest (Bates *et al.*, 2015). If fixed effects were statistically significant, *post-hoc* pairwise testing was carried out on the maximal model using estimated marginal means. P-values were Tukey-adjusted to control for multiple testing.

In the laboratory experiment, differences in brain and hemolymph levels of dopamine, octopamine and serotonin between QR and QL microcolonies were each analysed using Mann Whitney-U tests. Prior to this, Shapiro-Wilk and Bartlett tests revealed deviations from normality and homogeneity of variance respectively for each biogenic amine in each tissue measured.

To determine whether biogenic amine titres in the brain, hemolymph and ovary correlate with ovary activity in QL workers, analysis of biogenic amine levels across each ovary score was carried out for workers from the QL laboratory microcolony using GLMMS. GLMMs with an inverse link function were fitted for each biogenic amine and each tissue, with ovary score as a fixed effect and replicate and HPLC-FLD analysis batch as random effects. As above, LMMs were initially fitted but failed to meet assumptions of normality and homoscedasticity. Residual plots of GLMMs were visually inspected and revealed no obvious deviations from these assumptions. The exception is hemolymph octopamine, for which a log link function was applied as this was deemed to be a better

fit for the data following visual assessment of model residual plots. The overall significance of ovary score was computed by comparing the likelihood ratio of the full model to that of a null model containing random effects only (Bates *et al.*, 2015). *Posthoc* testing was carried out using estimated marginal means with Tukey adjustment.

3.3 Results

3.3.1 Recapture success rate

Of the 1,093 bees marked and released in total across all three trials, just 86 were successfully recaptured, a recapture rate of 7.9%. Due to this low success rate, only 8 QR workers were recaptured in total for the 21-day timepoint, hence the lower n values in this group. A breakdown of the numbers of marked bees released and recaptured from experimental hives in each trial is displayed in Table 3.1.

	Released Rec (Da		Recap (Day 1	tured .0)	Recaptured (Day 21)		Total Recaptured		%Recapture Success		Overall % Recapture
Trial	QL	QR	QL	QR	QL	QR	QL	QR	QL	QR	Success
1	147	82	10	11	9	7	19	18	12.9%	22.0%	
2	267	123	10	6	17	1	27	7	10.1%	5.7%	
3	304	150	10	6	0	0	10	6	3.3%	4%	
Total	738	355	30	24	26	8	56	31	7.6%	8.7%	7.9%

Table 3.1 Numbers of marked bees released into and recaptured from experimental hives at each timepoint. Total bees recaptured at each time point represents n values for each group.

3.3.2 Ovary activity of QR and QL workers

As expected, workers responded to the absence of a queen (or synthetic QMP) by activating their ovaries (Figure 3.3). A higher degree of ovary activity was observed in QL relative to QR workers in both hive and laboratory microcolonies, though this was statistically significant only in the laboratory (lab: χ^2 = 35.92, df = 3, p < 0.001; hive: χ^2 = 9.20, df = 9, p = 0.4192). Repression of ovary activity was to a lesser degree in QR laboratory microcolonies maintained with synthetic QMP than by a real queen in the



Figure 3.3 Ovary activation rates in the hive and laboratory microcolonies. A. Ovary activity was measured in marked workers released and recaptured from queenless (QL – shades of green) or queenright (QR - shades of orange) experimental hives at 10 and 21 days old. **B.** Comparative ovary activation rates of workers from laboratory-maintained micro-colonies. 'QR' microcolonies were supplied with 0.1 QE of synthetic QMP to emulate queenright conditions. 'QL' microcolonies received just a solvent control to emulate queenless conditions. Ovary scores range from 0-3 (Figure 2.2), where a higher score (darker shade) relates to a higher degree of ovary development. Ovary activity is presented as the proportion of workers with each ovary score within each group. Numbers of workers in each group (n) is displayed at the base of each bar. Differences in ovary activation rates between groups were determined using Pearson's Chi Squared, statistical differences (p<0.001) are denoted by asterisks. In the hive **(A)**, ovary activation rates did not differ between QL and QR hives on the days sampled, whereas ovary activity levels were higher in QL than QR microcolonies by day 10 in the laboratory **(B)**.

hive, as a small number of workers with active ovaries (11.58%) were observed in the former but not the latter (where 'active ovaries' are scores of 2 or above; Figure 2.2).

In addition, ovary activation rates of QL workers were accelerated in laboratory microcolonies compared with that seen in the hive. In QL workers sampled from the hive, ovary activation (scores of 2 or above; Figure 2.2) was not observed until day 21, while 42% of their laboratory-maintained counterparts had activated their ovaries by day 10. Due to the lower than expected ovary activity rates in the hive along with low recapture success rates, only a limited number of QL workers representative of each

ovary score could be obtained. A meaningful analysis of the differences in biogenic amines between ovary scores of QL workers could therefore not be carried out for workers sampled from the hive, and was carried out only within laboratory microcolonies.

3.3.3 Serotonin is unaffected by QMP

Serotonin could not be consistently detected at measurable quantities in the hemolymph or ovary, hence only brain data is presented (Figure 3.4, Figure 3.5).



Figure 3.4 Biogenic amine levels in workers from QR and QL hives. Levels of biogenic amines in **1.** brain, **2.** hemolymph and **3.** ovary tissue of marked workers released and recaptured from queenless (**QL** - green) or queenright (**QR** - orange) hives at 10 and 21 days old. As in Table 3.1, N = 30 (QL day 10), 24 (QR day 10), 26 (QL day 21) and 8 (QR day 21) for each tissue. Biogenic amines **a.** dopamine (**DA**), **b.** octopamine (**OA**) and **c.** serotonin (**5-HT**) were quantified in parallel using HPLC-FLD. Serotonin was not detected in the hemolymph so is not shown. Dopamine was the only biogenic amine detected in the ovary. For data visualisation clarity, outliers lying beyond the range of both boxplots were not plotted, but were included in all analyses. Biogenic amine contents of tissues was analysed using LMMs and GLMMs (depending on normality) with *post-hoc* testing carried out using estimated marginal means. Significant pairwise comparisons to the level of p <0.05 and bellow are denoted by boxplots not sharing a letter. Non-significant pairwise comparisons are denoted by 'n.s'. For table of *post-hoc* comparisons see Appendix B, Table B.1

Although detected in the brain, serotonin was not affected by QMP exposure. In the hive, levels of serotonin in worker brains were unaltered by queen presence, age or the interaction of these factors (Figure 3.4; LMM; Queen status: $\chi^2 = 0.69$, df = 2,83, p = 0.71; Age: $\chi^2 = 1.39$, df = 2,83, p = 0.50; Interaction: $\chi^2 = 0.28$, df = 1,84, p = 0.60). Similarly, no differences in brain serotonin were observed between QL and QR workers in laboratory microcolonies (Figure 3.5; MWU; W = 395, df = 56, p = 0.703), suggesting no direct effects of QMP exposure on serotonin.

3.3.4 Brain dopamine correlates with ovary activity

An increase in brain dopamine was associated with increasing ovary activity in QL laboratory microcolonies (Figure 3.6; GLMM; $\chi^2 = 8.53$, df = 3,24, p <0.05). Workers with signs of yolk deposition (ovary score of 2; Figure 2.2) had significantly elevated brain dopamine levels compared with less active workers (ovary score of 1; Z = 3.19, p <0.01) Interestingly, in workers possessing ovaries with fully mature oocytes (ovary score of 3; Figure 2.2) brain dopamine levels returned to that seen prior to the initiation of oogenesis (scores of 0 and 1; Figure 2.2), consistent with a role for dopamine early in the process of ovary activation. For a complete list of all *post-hoc* pairwise comparisons between ovary scores, see Table 3.2 and Figure 3.6.

Brain dopamine levels were also depressed in the presence of QMP. In the hive, brain dopamine levels were significantly lower in QR workers (Figure 3.4; GLMM; χ^2 = 18.58, df = 2,80, p <0.001), while both age its interaction with colony queen status had no effect

Contrast (ovary scores)	Estimate	SE	df	Z ratio	P value	
0-1	-0.000157	0.000335	Inf	-0.457	0.9682	
0-2	0.000652	0.000324	Inf	1.973	0.1981	
0-3	0.000124	0.000362	Inf	0.335	0.9870	
1-2	0.000809	0.000255	Inf	3.186	0.0079**	
1-3	0.000280	0.000283	Inf	0.993	0.7533	
2-3	-0.000529	0.000306	Inf	-1.733	0.3065	

Brain dopamine post-hoc pairwise comparisons

Table 3.2 *Post-hoc* comparisons of brain dopamine levels in QL workers between all pairwise combinations ovary scores. *Post-hoc* testing was computed using estimated marginal means on the maximal GLMM. GLMMs were fitted with inverse link functions, hence contrast estimates are on the inverse scale. P-values are Tukey adjusted.



Figure 3.5 Biogenic amine levels in workers from QR and QL laboratory microcolonies. HPLC-FLD quantification of the biogenic amines a. dopamine (DA), b. octopamine (OA) and c. serotonin (5-HT) in 1. brain and 2. hemolymph tissue of 10-day old laboratory-maintained workers. 5-HT was not detected in the hemolymph so is not shown. Workers were maintained in laboratory microcolonies of 100 bees. QMP+ microcolonies (orange) were supplied with 0.1 QE of synthetic QMP to emulate queenright hive conditions. To emulate queenless hive conditions, QMP- microcolonies (green) received just a solvent control. There were three replicates of each microcolony and brain tissue was obtained from 10 randomly selected bees from each (total n = 30 for each treatment). For clearer data visualisation, outliers lying beyond the range of both boxplots are not plotted, but are included in all data analyses. Statistical significance (Mann-Whitney U) at the level of p<0.001 is denoted by asterisks, and "n.s" denotes statistical non-significance.

(GLMM; Age: $\chi^2 = 2.27$, df = 2,80, p = 0.32; Interaction: $\chi^2 = 1.88$, df = 1,81, p = 0.17). *Post-hoc* testing revealed the depression of brain dopamine in QR workers to be statistically significant at both sampling timepoints (Day 10: Z = -2.84, p < 0.05; Day 21: Z = -2.740, p <0.05; Complete list of pairwise comparisons displayed on Figure 3.4 and Appendix B, Table B.1). However, in QR laboratory microcolonies there were no significant reductions of brain dopamine levels compared with QL workers (Figure 3.5; MWU; W = 473.5, df = 56, p = 0.41), despite the observed inhibition of ovary activity by synthetic QMP (Figure 3.3).



Figure 3.6 Brain biogenic amine titres vary with ovary activity level of QL workers. HPLC-FLD quantification of the biogenic amines **a.** dopamine (DA), **b.** octopamine (OA) and **c.** serotonin (5-HT) in **1.** brain and **2.** hemolymph tissue of 10-day old laboratory-maintained QL workers with differing degrees of ovary activity. 5-HT was not detected in the hemolymph so is not shown. Ovary scores range from 0-3 (see Figure 2.2), where a higher score (darker shade) relates to a higher degree of ovary activity. N = 6, 10, 5 and 7 for ovary scores 0, 1, 2 and 3 respectively. Differences in biogenic amine levels between ovary development stages were analysed using GLMMs with *post-hoc* testing carried out using estimated marginal means. Significant pairwise comparisons to the level of p <0.05 and bellow are denoted by boxplots not sharing a letter. Non-significant pairwise comparisons are denoted by 'n.s'. For full table of pairwise comparisons for brain dopamine, see Table 3.2

I also hypothesised that dopamine mediates QMP's repression of ovary activity by acting as a circulatory hormone and interacting directly with ovarian tissue. Dopamine was detected in both hemolymph and ovary samples, indicating a possible role in signalling to peripheral tissues. However, dopamine levels in these tissues were not significantly altered by QMP presence (Figure 3.4). In workers from the hive, no effect of queen status, age, or their interaction on ovary dopamine levels was found (GLMM; Queen status: $\chi^2 = 3.52$, df = 2,69, p = 0.17; Age: $\chi^2 = 1.38$, df = 2,69, p = 0.50; Interaction: $\chi^2 =$ 0.01, df = 1,70, p = 0.93). In parallel with brain dopamine levels, Hemolymph dopamine levels were also reduced in QR workers (Figure 3.4), though these reductions in the hemolymph were not statistically significant (LMM; Queen status: $\chi^2 = 3.49$, df = 2,81, p = 0.17). Hemolymph dopamine levels did significantly decrease with age, though there were no significant interaction effects between age and colony queen status (Age: $\chi^2 = 9.20$, df = 2,81, p< 0.05; Interaction: $\chi^2 = 0.17$, df = 1,82, p = 0.68). Similarly, workers maintained in the laboratory showed no difference in hemolymph dopamine titres between QL and QR microcolonies (Figure 3.5; MWU; W = 363, df = 56, p = 0.3752), suggesting that QMP does not affect titres in the hemolymph as it does in the brain. Additionally, hemolymph dopamine levels did not differ significantly between QL workers with different degrees of ovary development (Figure 3.6; GLMM; $\chi^2 = 0.88$, df = 3,23, p = 0.83), providing no evidence that ovary activation is initiated by changes in hemolymph dopamine titres.

3.3.5 Octopamine is modulated by QMP

Evidence to support a function of octopamine as a circulatory hormone acting directly on ovarian tissue was mixed. Although not detected in the ovary itself, octopamine was present in the hemolymph, and titres were significantly reduced in QR relative to QL workers sampled from the hive (Figure 3.4; GLMM; $\chi^2 = 20.77$, df = 2,79, p <0.001). *Posthoc* testing showed differences in brain octopamine to be statistically significant only in 10-day old workers (Z = -2.36, p <0.01; For full table of all pairwise comparisons see Appendix B, Table B.1). No effect of QMP on hemolymph octopamine levels was observed in the laboratory (Figure 3.5; MWU; W = 430, df = 56, p-value = 0.89), and there was no correlation between hemolymph octopamine and ovary score in workers from QL microcolonies (Figure 3.6; GLMM; $\chi^2 = 1.63$, df = 3,23, p = 0.65). In the hive, hemolymph octopamine was further affected by age, with reduced levels at 21 relative to 10 days in both QL and QR hives (Figure 3.4; GLMM; $\chi^2 = 19.95$, df = 2,79, p <0.001). No interaction effect between age and colony queen status was observed (GLMM; $\chi^2 =$ 3.64, df = 1,80, p = 0.06). Multiple lines of evidence were found supporting a link between brain octopamine and QMP's repression of ovary activity. In the hive, brain octopamine levels were unaffected by age and its interaction with colony queen status (Figure 3.4; GLMM; Age: $\chi^2 = 5.62$, df = 2,82, p = 0.07; Interaction: $\chi^2 = 2.20$, df = 1,83, p = 0.14), but were significantly reduced in QR relative to QL workers (GLMM; $\chi^2 = 22.45$, df = 2,82, p < 0.001). Brain octopamine levels were also depressed in workers from QR laboratory microcolonies (Figure 3.5; MWU; W = 636.5, df = 56, p < 0.001), indicating that this is in direct response to QMP exposure. Similarly to dopamine, in QL microcolonies increasing brain octopamine levels were associated with increasing degrees of ovary activity (Figure 3.6), though this was not statistically significant (GLMM; $\chi^2 = 4.13$, df = 3, 24, p = 0.25). Nonetheless, this points to a tentative role of octopamine in the control of honey bee worker reproduction.

3.4 Discussion

Biogenic amines are a highly conserved class of molecules which function as neurotransmitters, neuromodulators and neurohormones (Evans, 1980). They have diverse roles in the co-ordination of division of labour in eusocial insect societies (Reviewed by Sasaki & Harano, 2010), with dopamine in particular implicated in the mediation of reproductive constraint in a range of eusocial species (Sasaki & Harano, 2010). In honey bees, QMP's modulation of dopamine in the brain is thought to underly the inhibition of worker reproduction (Beggs *et al.*, 2007; Harris & Woodring, 1995; Sasaki & Nagao, 2001), though there has been a distinct shortage of studies linking these changes in the brain with the modulation of peripheral ovarian tissue. In this chapter, the hypothesis that biogenic amines have additional roles as QMP-responsive circulatory hormones and interact directly with ovarian tissue was empirically tested. In parallel to this, the effects of QMP on these biogenic amines in the brain were characterised, highlighting for the first time a possible role of octopamine acting alongside dopamine in the regulation of worker reproduction.

Brain octopamine was consistently depressed in QR workers both in the hive and the laboratory, implicating this molecule as a key mediator of QMP's effects. Brain

octopamine titres have not previously been found to be altered by QMP (Harris & Woodring, 1995). However, the HPLC methodologies used for biogenic amine quantification differed considerably with those used in this study, where the derivatization of compounds in combination with fluorescent detection enabled highly sensitive determination of octopamine titres which may have revealed differences not previously seen.

QMP's depression of brain octopamine suggests an involvement of this molecule in the mediation of worker behavioural or physiological responses to the queen and QMP. The additional finding that brain octopamine levels positively correlated with ovary activity in QL workers (though not statistically significantly) points to a speculative role of octopamine as a positive regulator of ovary development. Octopamine is involved in the regulation of reproduction in a range of insects; For instance, egg laying is stimulated by the injection of octopamine in the rice leaf bug, *Trigonotylus caelestialium* (Yamane, 2013) and the diamondback moth, *Plutella xylostella* (Li *et al.*, 2020). In *D. melanogaster*, octopamine regulates ovulation and the muscle contractions involved in egg-laying (Meiselman *et al.*, 2018; White *et al.*, 2021). It is therefore not beyond reason to speculate that a conserved gonadotropic function of octopamine may also exist in honey bees.

The effects of octopamine in honey bees have previously been restricted to the mediation of behavioural maturation and division of labour. The elevation of brain octopamine levels with age is a driver of temporal polyethism by mediating the behavioural switch to foraging in older workers (Wagener-Hulme *et al.*, 1999). Treatment of young workers with octopamine has also been demonstrated to induce precocious foraging (Schulz *et al.*, 2002; Schulz & Robinson, 2001). The observed reduction in hemolymph octopamine between 10 to 21 days is contrary to this, and this discrepancy with previous studies is attributed to seasonal effects. Octopamine levels, along with foraging rates, decline into the Autumn (Harris & Woodring, 1992). As sampling from the hive occurred in September, overall seasonal reductions in foraging may explain the observed decrease in octopamine levels between timepoints. This may

limit the validity of between-timepoint comparisons, but is unlikely to confound comparisons of biogenic amine levels between QR and QL workers within each timepoint, as differences arising from exposure to queen pheromones should be unaffected.

In keeping with previous studies, brain dopamine levels correlated positively with ovary activity (Harris & Woodring, 1995; Sasaki & Nagao, 2001) and were reduced in QR relative to QL workers sampled from the hive (Beggs et al., 2007), adding to existing evidence that depression of brain dopamine by QMP may enforce worker reproductive constraint. However, brain dopamine levels were not reduced by QMP in workers from laboratory-maintained microcolonies despite its repression of ovary activity. This may indicate that dopamine, and other biogenic amines, are not the sole regulators of QMP's effects on reproduction. The reduced brain dopamine levels seen in QR workers from the hive may have arisen from an additional component not present in the laboratory, such as brood pheromone (Mohammedi et al., 1998; Pankiw & Garza, 2007). The presence of additional regulatory components in the hive would explain the tighter inhibition of reproductive constraint seen in the QR hive compared with QR laboratory microcolonies, where in the laboratory we see approximately 11% ovary activation even in the presence of synthetic QMP (Figure 3.3). However, while multiple mechanisms of reproductive constraint are probable, this is not a likely explanation for the lack of dopamine depression in QR laboratory microcolonies. Notably, the QMP component homovanillyl alcohol (HVA), which has been directly implicated in the depression of brain dopamine levels (Beggs et al., 2007), was present in the synthetic QMP blend used in laboratory trials.

The observed depression of brain dopamine levels in QR workers from the hive but not in the laboratory could be explained by the different rates of ovary development in each setting. Laboratory microcolonies received a high-protein diet specifically formulated to permit ovary activation (Duncan *et al.*, 2016), which may substantially accelerate ovary development compared with the hive. In the hive, workers sampled showed no signs of ovary activation until day 21, at which mature oocytes were present in only 12% of workers. Meanwhile, in laboratory microcolonies 42% of QL workers possessed fully mature oocytes within 10 days. Brain dopamine levels were found to positively correlate with ovary development during the initial stages of ovary activation (scores 0-2; Figure 2.2) followed by a drop in workers possessing fully mature oocytes, implying an involvement of dopamine in the initial stages of oogenesis. The lack of difference in dopamine levels in the laboratory may therefore be reflective of the higher proportion of workers at later stages of ovary maturity than was seen in the hive at either time point.

Multiple lines of evidence support a role for dopamine in regulating the early processes of oogenesis. Previous work has also found reduced brain dopamine levels in QR relative to QL workers in the hive which were not replicated in 6-12 day old workers in the laboratory (Harris & Woodring, 1995). Where depression of brain dopamine by QMP has been observed in the laboratory previously, workers were just two-days old (Beggs et al., 2007). At two-days old laboratory workers are at the beginning of the pre-vitellogenic stage, whereas by 10-days old they are reaching peak vitellogenic stage in which mature oocytes are formed (Koudjil & Doumandji, 2008). The mechanism that maintains worker sterility within the ovary tissue acts on the earliest stages of development; In the presence of QMP, Notch signalling acts in the germarium, the region of the ovary where oocytes are specified, to actively inhibit oogenesis (Figure 1.4; Duncan et al., 2016). Further, the expression of the dopamine receptor AmDop1 in worker ovaries is localized to the germarium (Duncan, unpublished data; see Appendix F, Figure F.1) and is upregulated in two-day old QL workers (Vergoz et al., 2012). This indicates a potential link between dopamine and the initiation of early stages of oogenesis via the inhibition of Notch signalling, though further study is required to establish whether dopamine receptor activity can modulate Notch signalling (refer section 7.3.3, General Discussion). An investigation into the temporal dynamics of brain dopamine levels following the loss of a queen and how these correspond to ovary developmental stage is also needed to establish the temporal dynamics of changes in dopamine levels preceding ovary activation in both the laboratory and in the hive.

Links between brain biogenic amine titres and the regulation of peripheral tissues are still somewhat uncertain. Evidence was found in support of the idea that dopamine and octopamine have additional roles as circulating neurohormones. As well as being present in the brain, both dopamine and octopamine were detected in the hemolymph, hence direct interaction of these biogenic amines with the AmDop1, AmDop3 and AmOA1 receptors present in the ovarian tissue is feasible (Vergoz et al., 2012). There was also evidence to suggest that queen pheromones are capable of modulating these biogenic amines the hemolymph. Hemolymph titres of both dopamine and octopamine were reduced in workers from QR relative to QL hives, though this was statistically significant only for octopamine (Figure 3.4). The reductions in hemolymph octopamine could directly regulate the repression of ovary activity through reduced activation of the ovarian AmOA1 receptor, and may be a key mechanism underlying reproductive constraint. However, the suppression of hemolymph octopamine seen in the QR hive was not observed in QR laboratory microcolonies receiving synthetic QMP (Figure 3.5), raising the same questions as discussed above for brain dopamine: it this due to the temporal dynamics of biogenic amine modulation, or the additional queen pheromones not present in the laboratory?

Regardless of the differences observed between workers from the hive and laboratory microcolonies, the suppression of hemolymph octopamine in QR hives could be an important regulator of reproductive constraint. Ovarian biogenic amine receptor expression is dynamic and is also influenced by QMP presence, which may serve as an additional axis for the mediation of ovary development. AmOA1 expression in the ovary is reduced at later developmental stages (Vergoz *et al.*, 2012), perhaps reflecting a loss of capacity to respond to changes in hemolymph octopamine that signal the queen's presence once ovaries are fully mature. Although hemolymph dopamine titres were not reduced by QMP to a statistically significant degree, ovarian responses to dopamine may also still be affected through the rapid alteration in expression levels of the antagonistic dopamine receptors *AmDop1* and *AmDop3* in the ovary in response to the absence of QMP (Vergoz *et al.*, 2012).

If dopamine and octopamine in the hemolymph function as circulating neurohormones mediating ovary development, the relationship between brain biogenic amines and reproduction becomes less clear. In particular, it is unclear whether QMP, and possibly other queen or brood pheromones, co-modulate biogenic amines in the brain and hemolymph through a shared mechanism, or whether these are distinct processes. The source of biogenic amines in the hemolymph, and hence how they may be regulated, is uncertain; For instance, they could be secreted from peripheral tissues, secreted from the brain itself, or synthesized in the hemocytes as in the moth, *Chilo suppressalis* (Wu *et al.*, 2015). In addition, dopamine and octopamine in the brain could regulate reproduction via additional mechanisms that have not been considered here, such as through regulation of the wider neuroendocrine signalling network (i.e. Insulin/20E pathways) or via neuronal signalling. These ideas are discussed in greater depth in section 7.3.2, General Discussion.

In this chapter, further supporting evidence for a regulatory role of dopamine in QMP's inhibition of worker reproduction is demonstrated, with a hypothesised role early in the initiation of oogenesis. Regulation of octopamine, whether by QMP or by other queen or brood pheromone components, is additionally proposed as a mediator of ovary development. The presence of these biogenic amines in the hemolymph, and the responsiveness of octopamine in particular to the presence of a queen, indicates that direct effects on ovary development via interaction with the ovary tissue is possible. Further study is required to determine the relationship between brain and hemolymph biogenic amine titres; In particular, whether these represent part of the same mechanism regulating ovary development, or distinct regulatory pathways.

Chapter 4

Testing the causal relationship between dopamine levels and ovary development

4.1 Introduction

Biogenic amines, in particular dopamine, have been proposed as mediators of QMP's induction of sterility in honey bee workers (Section 1.5.3; Also reviewed by Sasaki & Harano, 2010; Sasaki et al., 2021). In Chapter three, additional evidence was found to support a role for dopamine, and possibly octopamine, in regulating reproductive constraint. Most notably, brain dopamine levels were shown to be reduced in workers from queenright (QR) relative to queenless (QL) hives (consistent with previous studies; Beggs et al., 2007; Harris & Woodring, 1995; Sasaki & Nagao, 2001), suggesting that QMP, and possibly other queen pheromones, are capable of modulating dopamine levels in workers. In addition, it was demonstrated that brain dopamine levels positively correlate with the initial stages of ovary development in QL workers, consistent with the positive relationship between ovary activity and dopamine levels reported by Sasaki & Nagao (2001). While this adds to a large body of circumstantial support for a role of dopamine in mediating QMP's inhibition of reproduction, it could equally be concluded from the findings of Chapter three and of previous studies that engaging in ovarian development in the absence of the queen simply triggers a rise in dopamine titres in the brain. As such, it is integral to establish the causal relationship between dopamine levels and ovary activity to progress the idea that QMP's modulation of dopamine is a driver of sterility in workers.

In *Drosophila virillis*, dopamine's function as a gonadotropin has been well established. In this species, dopamine positively regulates oogenesis predominantly via regulation of the gonadotropin juvenile hormone (JH) (Gruntenko *et al.*, 2005, 2007). A shortage of empirical studies from other species means that the extent to which the neuroendocrine signalling network in *Drosophila* species is representative of all insects is unclear. For instance, the gonadotropic function of JH has been lost in some advanced eusocial species, including the honey bee (Rodrigues & Flatt, 2016), and it is not completely clear how the neuroendocrine network has been rewired to accommodate this loss of function. Such differences in neuroendocrine signalling between insect species emphasise the need for establishing the specific roles of hormones and signalling molecules on a species-by-species basis (Reviewed by Knapp, Norman *et al.*, 2022, in press).

In eusocial insects, evidence implicating dopamine as a regulator of ovary development is predominantly correlative. For instance, there is a positive association between brain dopamine levels and ovarian development in paper wasp workers (Sasaki *et al.*, 2007; Yoshimura *et al.*, 2021), bumble bees (Bloch *et al.*, 2000; Sasaki *et al.*, 2017), and reproductive females (gamergates) in derived queenless ant species (Okada *et al.*, 2015; Penick *et al.*, 2014). However, some manipulative studies have begun to address the causal relationship between dopamine levels and ovary activity. For example, supplementary dopamine feeding enhanced reproductive development in workers of the paper wasp *Polistes chinesis* (Sasaki *et al.*, 2009) and derived queenless *Diacamma* ants (Okada *et al.*, 2015). A gonadotropic function of dopamine has been most convincingly shown in the fire ant *Solenopsis invicta*, where the inhibition of dopamine biosynthesis through treatment with the dopamine synthesis inhibitor 3-iodo-L-tyrosine led to reduced oocyte numbers (Boulay *et al.*, 2001). Normal reproductive function was restored with the application of the dopamine precursor L-Dopa (Boulay *et al.*, 2001), suggesting that dopamine is required for ovarian development.

Circumstantial evidence for a gonadotropic function of dopamine in honey bees has been discussed in previous sections (i.e. Section 1.5.3, General Introduction) and also demonstrated in Chapter three, where brain dopamine levels positively correlated with ovary activity in QL workers (consistent with Sasaki & Nagao, 2001). It has also been well established from previous work that QMP depresses brain dopamine levels in workers (Beggs *et al.*, 2007; Harris & Woodring, 1995; Sasaki & Nagao, 2001; Chapter three), with the QMP component HVA shown to be at least partially responsible for this depression (Beggs & Mercer, 2009). From this, along with the gonadotropic function of dopamine in other related eusocial Hymenopterans (Boulay *et al.*, 2001; Okada *et al.*, 2015; Sasaki *et al.*, 2009), QMP's depression of brain dopamine levels has been assumed to be a cause of ovary inhibition in honey bee workers (i.e. Sasaki and Harano, 2010; Sasaki *et al.*, 2021). However, empirical support for a gonadotropic effect of dopamine in honey bees rests largely on just one highly cited study, in which dopamine supplementation was found to enhance ovary activation rates in queenless (QL) honey bee workers (Dombroski *et al.*, 2003). Despite a further two decades of research, the assumed causal effect of dopamine on ovarian development in honey bees still rests solely on this study, with no published replications of these findings to my knowledge.

In addition, Dombroski *et al.*, (2003)'s finding that supplementing QL workers with dopamine further promotes ovary activity suggests that dopamine has a positive effect on ovary development in workers already free from QMP's inhibitory effects. However, it does not test the hypothesis that QMP's depression of dopamine is responsible for the repression of ovary development in QR workers. For instance, if QMP's depression of dopamine is a mechanism by which reproductive constraint is enforced, I hypothesise that restoring the dopamine levels of workers to 'QL' levels through dietary dopamine supplementation should at least partially overcome QMP's inhibitory effects. Similarly, if dopamine is required for ovarian development, reducing the dopamine levels of workers through treatment with a dopamine synthesis inhibitor (i.e. as in Boulay *et al.*, 2001) should stunt the ovary development of QL workers.

This chapter aims to validate the assumed gonadotropic effect of dopamine in honey bee workers proposed in chapter three by seeking to replicate the results reported by Dombroski *et al.*, (2003). Initial dopamine supplementation regimes are carried out as by Dombroski *et al.*, (2003), with further experiments testing the effect of both dopamine and its precursor L-Dopa on ovarian development at a wider range of doses. In addition, the hypothesis that QMP's depression of brain dopamine inhibits ovary activation in QR workers is addressed by extending dopamine supplementation regimes to also include workers exposed to QMP. The hypothesis that dopamine is required for ovary activation was also tested by observing the effects of inhibiting dopamine biosynthesis on ovary development. This extensive series of manipulative experiments provides a broader framework of empirical evidence from which the causal relationship between dopamine levels and ovary activity in honey bee workers, and hence the mechanism behind QMP's induction of sterility, can be inferred.

4.2 Methods

4.2.1 Laboratory microcolony set-up

A. mellifera hives were maintained as described previously (Section 2.1.1). Laboratory microcolonies containing 80-100 workers were setup in cages as in Section 2.1.1 and underwent a variety of dopamine supplantation and inhibition treatments to elucidate the causal relationship between dopamine levels and ovary activity. The ovaries of all remaining workers from each microcolony were dissected, imaged and scored as described in Section 2.2 after 10 days.

4.2.2 Dopamine supplementation treatments

Initial dopamine supplementation experiments aimed to replicate the results of Dombroski *et al.*, (2003), in which dietary dopamine increased ovary activation in QL workers. QL laboratory microcolonies containing 100 workers were setup between June-July 2018 with three independent replicates of each treatment cage. Dietary dopamine was administered following the methodology of Dombroski *et al.*, (2003) by adding 0.01 mg dopamine/g food offered to DA+ cages, but not DA- (control) cages. To additionally assess whether dietary dopamine can overcome QMP's repression of ovary activity, dopamine treatments were tested in both the presence (QMP+) and absence (QMP-) of synthetic QMP. QMP treatments were administered following the methodology described in section 2.1.2.

In subsequent attempts to replicate the phenotype of Dombroski *et al.*, (2003), dietary dopamine was also tested at the increased dosages of 0.1, 0.5, 1 and 2 mg/g food in

cages of 100 workers maintained in the absence of QMP only. Cages were setup between August-October 2018 with three independent replicates of each treatment cage.

4.2.3 L-Dopa supplementation treatments

In further attempts to replicate the results of Dombroski *et al.*, (2003), treatment with the dopamine precursor L-dopa (Figure 4.1) was trialled as an alternative means of dopamine supplementation. L-Dopa is commonly used as a drug in humans to restore dopamine levels in the brain as a treatment for Parkinson's disease, and is favoured over treating with dopamine directly as it crosses the blood-brain-barrier more effectively (Haddad *et al.*, 2018). It was hence hypothesised that L-Dopa treatment may also lead to more effective elevation of dopamine levels in honey bee worker brains. Effects of the dopamine precursor L-dopa on ovary activation were tested in the absence of QMP only. Two experimental cage replicates were set up between July-August 2019, with a final third replicate carried out in August 2020. Microcolonies contained 80 workers due to constraints on ovary dissection capacity at the end of the experiment. As L-dopa has



Figure 4.1 Schematic of the dopamine biosynthesis pathway Tyrosine hydroxylase converts L-Tyrosine to the dopamine precursor L-Dopa. Inhibiting the action of tyrosine hydroxylase using the inhibitor 3-iodo-L-tyrosine therefore leads to reduced dopamine biosynthesis. Figure sourced from Haddad *et al.*, (2018) (open access)

lower water solubility than dopamine (0.99 mg/mL as opposed to 18.96 mg/mL), it was not possible to spike the food with concentrated solution as was carried out for dopamine treatments without the food mixture becoming too runny. L-dopa was instead administered by spiking water solutions offered to cages at concentrations of 0.01, 0.1 and 0.5mg/mL, while controls received just water. Every day L-dopa solutions were replaced with pre-made solutions stored at -20°C. To assess the impact of L-dopa supplementation on biogenic amine levels, brains were also collected from a subset of five randomly selected bees per cage on day 10 for later quantification of dopamine. Octopamine and serotonin levels were also quantified in case of possible effects of L-Dopa supplementation on non-target biogenic amines. Brain samples were collected, prepared and analysed using HPLC-FLD as described in section 2.3.

4.2.4 Dopamine inhibition treatments

To test whether a reduction in dopamine levels alone could mirror QMP's inhibitory effects on ovary activity, dopamine levels were experimentally reduced through administration of the dopamine synthesis inhibitor 3-iodo-L-tyrosine (iodotyrosine). Iodotyrosine inhibits tyrosine hydroxylase, the rate-limiting enzyme in the biosynthesis of the dopamine precursor L-Dopa (Figure 4.1), and has been shown to reduce dopamine levels in D. melanogaster (Neckameyer, 1996) and fire ants (Boulay et al., 2001). lodotyrosine was supplied to bees by spiking their water solutions at concentrations of 0.01, 0.1 and 0.5 mg/mL, while controls received just water (Similar to administration methodology used by Boulay et al., 2001). As above, iodotyrosine solutions were replaced with frozen pre-made solutions each day. To determine whether dopamine inhibition could exacerbate QMP's repression of ovary activity, two experimental cages were setup for each iodotyrosine treatment, one maintained in the presence (QMP+) and one in the absence of synthetic QMP (QMP-). QMP treatments were administered following the methodology described in Section 2.1.2. Three independent replicates for all eight treatment combinations were carried out between August-September 2019. Experimental cages again each contained 80 workers in line with ovary dissection capabilities.

4.2.5 Data analysis

Data analysis was carried out in R. Details of packages used for analysis are found in section 2.4. In all experiments in which a range of doses of each treatment solution were used (i.e. some dopamine experiments, L-Dopa experiments, iodotyrosine experiments), dose was considered an ordinal variable in all analyses, as the number of doses tested was insufficient to be considered a continual variable.

Food intake was compared across treatment groups for each experiment by fitting linear mixed effects models (LMMs) with mean food intake per bee per day as the response variable. Where treatments were administered in water solutions (L-dopa and iodotyrosine), additional analyses with water intake per bee per day as the response variable were also performed. In initial dopamine supplementation experiments in the presence and absence of QMP, dopamine, QMP presence and their interaction were included as fixed effects. In subsequent experiments testing dopamine and L-dopa at a range of doses in the absence of QMP only, dose was included as the sole fixed effect. For dopamine inhibition experiments fixed effects were iodotyrosine dose, QMP presence and their interaction. To account for repeated measures, day was included as a random effect and nested within replicate for all models. Visual examination of all model residual plots revealed no obvious deviations from normality or homoscedasticity. Statistical significance of fixed effects was determined by comparing the likelihood ratio of the maximal model to that of the null model, or model without the fixed effect of interest. If fixed effects were statistically significant, *post-hoc* testing was carried out using estimated marginal means to determine significance of all pairwise treatment comparisons. P-values were Tukey adjusted to control for multiple testing.

In each experiment survival distributions were compared between treatment groups using Cox proportional hazards (CPH) models with mixed effects. Analysis of initial dopamine supplementation experiments involving QMP included dopamine, QMP presence and their interaction as fixed effects. In subsequent dopamine and L-dopa dosage experiments dopamine or L-dopa dose was the sole fixed effect. Survival rates in dopamine inhibition experiments were analysed with iodotyrosine dose, QMP and their interaction as fixed effects. In all models replicate was included as a random effect and the assumption of proportional hazards was verified by visual examination of the correlation of scaled Schoenfield residuals with time to test for independence. In each analysis CPH models were used to predict Hazard Ratios (HR) and confidence intervals (CI) for fixed effects. HRs are presented as HR (±95% CI). Where fixed effects were statistically significant, *post-hoc* testing was carried out using estimated marginal means to determine the significance of pairwise treatment comparisons. P-values were Tukey adjusted to control for multiple testing.

To assess the effect of different dopamine and QMP treatments on ovary activity, cumulative link mixed effects models (CLMMs) were used with ovary score as an ordinal response variable. In initial dopamine supplementation experiments in the presence and absence of QMP, CLMMs were fitted with dopamine, QMP presence and their interaction as fixed effects. In subsequent experiments testing dopamine and L-dopa supplementation at a range of doses in the absence of QMP only, dose was considered an ordinal variable and modelled as the sole fixed effect. For dopamine inhibition experiments fixed effects were iodotyrosine dose, QMP and their interaction. Replicate was included as a random effect for all models. In each analyses statistical significance of fixed effects was determined by comparing the likelihood ratio of the maximal model to that of the null model, or model without the fixed effect of interest. Where fixed effects were statistically significant, *post-hoc* testing was carried out by computing least-squares means to determine the significance of pairwise comparisons. P-values were Tukey adjusted to control for multiple testing.

Brain levels of dopamine, octopamine and serotonin following different L-dopa treatment regimes were each assessed by fitting GLMMs with a gamma distribution and an inverse link function. L-dopa dose was included as the fixed effect and replicate and sample preparation block were included as random effects. GLMMs were deemed more appropriate for the data than LMMs, as LMMs which were initially fitted and were found not to meet the assumptions of normality or homoscedasticity of model residuals. Visual examination of GLMM model residual plots revealed no obvious deviations from normality or homoscedasticity. Statistical significance of L-dopa treatment on each biogenic amine was assessed by comparing the likelihood ratio of the full model to that of the null model containing no fixed effects.

4.3 Results

4.3.1 Dopamine has no effect on ovary activity in the presence or absence of QMP

Initial dopamine supplementation experiments saw dopamine administered to workers based on the methodology of Dombroski *et al.*, (2003) in an attempt to replicate these findings and additionally test dopamine's effects in the presence of QMP. Survival rates and food consumption levels were compared across treatments to examine whether dopamine supplementation regimes caused non-target effects on nutritional status or mortality (Figure 4.2). Dietary dopamine treatment did not significantly compromise survival rates (CPH: HR = 0.96 ± 0.17 SE, z = -0.28, p = 0.78), nor did QMP exposure (CPH: HR = 1.16 ± 0.16 , z = 0.91, p = 0.36) or the combination of the two (CPH: HR = 1.38 ± 0.22 , z = 1.44, p = 0.15). Food consumption rates were also consistent across treatments, with bees from dopamine-treated cages consuming an average of 0.13 µg dopamine per bee per day. Food intake did not differ significantly with dopamine, QMP exposure or their

Post-hoc pairwise comparisons of ovary scores between dopamine supplementation treatments (+/- QMP)						
Contrast	Estimate	SE	df	Z-ratio	P-value	
QMP- DA- vs QMP+ DA-	-1.268	0.213	Inf	-5.965	<0.0001***	
QMP- DA- vs QMP- DA+	0.185	0.192	Inf	0.965	0.7694	
QMP- DA- vs QMP+ DA+	-0.920	0.210	Inf	-4.379	0.0001***	
QMP+ DA- vs QMP- DA+	1.454	0.210	Inf	6.918	<0.0001***	
QMP+ DA- vs QMP+ DA+	0.348	0.221	Inf	1.580	0.3903	
QMP- DA+ vs QMP+ DA+	-1.105	0.208	Inf	-5.321	<0.0001***	

Table 4.1 Post-hoc pairwise comparisons of ovary scores between all QMP/DA treatment combinations. *Post-hoc* comparisons were computed using least squares means on the maximal CLMM.



Figure 4.2 Dopamine supplementation in the presence and absence of QMP A. Ovary activity is reduced in bees exposed to QMP (orange bars, QMP+) relative to unexposed bees (green bars, QMP-), while dopamine supplementation at 10 μ g/g food (DA+ vs DA-) has no effect. Ovary activity is shown as the proportion of bees within each treatment group with each ovary score (ranging from 0-3; Figure 2.2), where a higher score (darker shade) relates to a higher degree of development. N values for each treatment pooled across three replicates are displayed at the base of each bar. Statistical significance of *post-hoc* pairwise comparisons is denoted by bars not sharing a letter (least squares means, p<0.05; see Table 4.1) **B.** Survival probability distributions of cages of 100 bees over 10 days. Data shown consists of three replicates of each treatment cage pooled together. Y axis has been truncated to begin at a survival probability of 0.5. Statistical non-significance (CPH mixed effects models) is denoted by "n.s". **C.** Mean food intake per bee per day does not differ between treatment groups, as denoted by "n.s" (LMMs). Food intake was recorded daily for each cage (n = 10 days), each of which had three replicates (n = 30 for each treatment).

interaction (Figure 4.2; GLMMs: dopamine: $\chi^2 = 1.03$, d.f = 2, 117, p = 0.60; QMP: $\chi^2 = 1.82$, d.f = 2, 117, p = 0.40; Interaction: $\chi^2 = 0.37$, d.f = 1, 118, p = 0.54). There is therefore no evidence that dopamine supplementation treatments caused any confounding effects on mortality or nutritional status.

Assessment of ovary activity levels between QMP+ and QMP- treatments in dopamine controls (DA-) validated the efficacy of synthetic QMP as an inhibitor of ovary development (Figure 4.2). Ovary activity levels were significantly lower in QMP exposed bees (CLMM: χ^2 = 64.23, d.f = 2, 759, p < 0.001), with the proportion of workers with active ovaries (ovary score of 2 or above; Figure 2.2) reduced from 40% to 10% in its presence in DA- controls. However, in contrast to the findings of Dombroski *et al.*, (2003),

there was no evidence to support an effect of supplementary dopamine feeding on ovary activation (Figure 4.2; CLMM: $\chi^2 = 3.44$, d.f = 2, 759, p = 0.18). Dopamine supplementation also could not overcome QMP's inhibitory effects, as *post-hoc* testing revealed there to be no significant effect of dietary dopamine on ovary activation in the presence of QMP (All pairwise *post-hoc* comparisons displayed on Figure 4.2A and Table 4.1). Further, no interaction effects between QMP and dietary dopamine on ovary activity were observed (figure 1A; CLMM: $\chi^2 = 0.31$, d.f = 1, 760, p = 0.58).

4.3.2 Dopamine and L-Dopa do not enhance ovary activity in QL workers at any dose

Dopamine and its precursor L-dopa were administered to bees in the absence of QMP at a range of increased doses to assess whether effects on ovary activity are dose-dependent. Dopamine treatment at doses of 0.1-2mg/g food had no significant effect on



Figure 4.3 Effects of dopamine supplementation at a range of doses A. Dopamine supplementation does not enhance ovary activity of QL bees (housed without QMP) at any dose tested. Ovary activity is shown as the proportion of bees within each treatment group with each ovary score (ranging from 0-3; Figure 2.2), where a higher score (darker shade) related to higher degree of ovary development. N values for each treatment across three pooled replicates are displayed at the base of each bar. "n.s" denotes statistical non-significance (CLMMs). **B.** Survival probability distributions of cages of 100 bees over 10 days of dietary dopamine treatments. Data shown consists of three replicates of each treatment cage. Y axis has been truncated to begin at a survival probability of 0.5. Statistical non-significance (CPH mixed effects models) is denoted by "n.s". **C.** Mean food intake per bee per day does not differ with dietary dopamine dose, as denoted by "n.s" (LMMs). Food intake was recorded daily for each cage (n = 10 days), each of which had three replicates (n = 30 for each treatment).


Figure 4.4 Effects of L-Dopa supplementation at a range of doses A. L-dopa supplementation does not enhance ovary activity of QL bees (housed without QMP) at any dose tested. Ovary activity is shown as the proportion of bees within each L-dopa treatment dose with each ovary score (ranging from 0-3), where a higher score (darker shade) related to higher degree of ovary development. N values for each treatment across three pooled replicates are displayed at the base of each bar. "n.s" denotes statistical non-significance (CLMMs). **B.** Mean food intake and **D.** mean water intake is not affected by L-dopa treatment, as denoted by "n.s" (LMMs). Food and water intake was recorded daily for each cage (n = 10 days), each of which had three replicates (n = 30 for each treatment). **C.** Survival probability distributions of cages of 80 bees over 10 days of different L-dopa treatments. Data shown consists of three replicates of each treatment cage. Y axis has been truncated to begin at a survival probability of 0.5. Statistical non-significance (CPH mixed effects models) is denoted by "n.s".

survival rates at the doses tested (Figure 4.3; CPH: 0.1 mg/g HR = 0.94 \pm 0.21 SE, z = - 0.29, p = 0.78; 0.5 mg/g HR = 1.09 \pm 0.20 SE, z = 0.42, p = 0.68; 1 mg/g HR = 1.30 \pm 0.20 SE, z = 1.32, p = 0.19; 2 mg/g HR = 1.42 \pm 0.18 SE, z = 1.92, p = 0.05). Food consumption rates were also unaffected by dietary dopamine at these doses (Figure 4.3; LMM: χ 2 = 8.60, d.f = 4, 145, p = 0.07). L-dopa treatment similarly caused no negative effects on survival rates at the doses of 0.01-0.5 mg/mL water tested (Figure 4.4; CPH: 0.01 mg/mL



Figure 4.5 Biogenic amine levels following L-Dopa supplementation HPLC-FLD quantification of QL worker brain biogenic amines following 10-day L-dopa supplementation at a range of doses. Brains were collected from five randomly selected bees from each cage over three replicates (n=15 total for each L-dopa treatment). Levels of A. dopamine, B. octopamine and C. serotonin were not altered by L-dopa treatments, as denoted by "n.s" (GLMMs).

HR = 1.76 ± 0.40 SE, z = 1.41, p = 0.16; 0.1 mg/mL HR = 0.49 ± 0.55 SE, z = -1.29, p = 0.20; 0.5 mg/mL HR = 0.79 ± 0.47 SE, z = -0.49, p = 0.62). Additionally, food consumption rates did not differ with L-dopa dose (Figure 4.4; LMM: $\chi 2 = 5.08$, d.f = 3, 116, p = 0.17). Water or treatment solution intake was also monitored in microcolonies receiving L-dopa treatment so that dosages of L-dopa solution received could be assessed. Solution intake was consistent across L-dopa treatment regimes, indicating that received dosages were not confounded by differential rates of consumption (Figure 4.4; LMM: $\chi 2 = 2.08$, d.f = 3, 116, p = 0.56).

Despite the increased dosages of dietary dopamine tested compared with Dombroski *et al.*, (2003), neither dopamine nor L-dopa supplementation enhanced ovary activity rates in the absence of QMP, conflicting with previously published data. Within the doses tested dietary dopamine had no effect on ovary activity (Figure 4.3; CLMM: $\chi 2 = 1.68$, d.f = 4, 1009, p = 0.79), nor did the dopamine precursor L-dopa (Figure 4.4; CLMM: $\chi 2 = 7.71$, d.f = 3, 834, p = 0.05). However, L-dopa supplementation was not found to elevate levels of dopamine or other biogenic amines (Figure 4.5). At the doses tested, L-dopa treatment was not associated with differences in brain dopamine (GLMM: $\chi 2 = 3.09$, d.f

= 3, 55, p = 0.38), octopamine (GLMM: $\chi 2$ = 1.44, d.f = 3, 55, p = 0.70), or serotonin (GLMM: $\chi 2$ = 0.85, d.f = 3, 55, p = 0.84). The efficacy of this L-dopa supplementation method as an elevator of dopamine levels is therefore unclear.

4.3.3 lodotyrosine represses reproduction to a degree comparable with QMP

While ovary activity could not be further increased by dopamine supplementation, treatment with the dopamine synthesis inhibitor iodotyrosine at doses of 0.1 and 0.5 mg/mL reduced ovary activation rates (Figure 4.6), implicating depressed dopamine levels as a suppressor of reproduction. Ovary activity levels were significantly lower following treatment with both QMP (CLMM; $\chi 2 = 75.49$, d.f = 4, 1618, p<0.001) and iodotyrosine (CLMM; $\chi 2 = 139.65$, d.f = 6, 1616, p<0.001), and there was also a significant interaction effect between the two (CLMM; $\chi 2 = 28.59$, d.f = 3, 1619, p<0.001). *Post-hoc* pairwise comparisons revealed that supplementing QL workers with iodotyrosine at 0.1 mg/mL repressed ovary activity to the same extent as QMP exposure alone and significantly exceeded QMP's repression at 0.5 mg/mL (Figure 4.6). In bees already exposed to QMP, ovary activity was further repressed by iodotyrosine, though only at the highest dose of 0.5 mg/mL (Figure 4.6). A full table containing all *post-hoc* pairwise tests can be found in Appendix G (Table G.1).

Food consumption levels were consistent across all treatments, indicating that variation in ovary activity between treatment groups was not driven by differences in nutrition (Figure 4.7). Food intake was not affected by QMP (LMM; $\chi 2 = 3.06$, df = 4, 235, p = 0.55), iodotyrosine (LMM; $\chi 2 = 7.50$, df = 6, 233, p = 0.28), or the interaction between the two (LMM; $\chi 2 = 1.96$, df = 3, 236, p = 0.58). However, it should be noted that water or solution intake was affected by treatment group (Figure 4.7), being significantly higher in QMPrelative to QMP+ bees (LMM; $\chi 2 = 38.42$, df = 4, 235, p<0.001). Iodotyrosine dosage will therefore have been slightly elevated in the former, though this does not impact the interpretation of results as comparisons between iodotyrosine treatments can still be



Figure 4.6 Ovary activity levels following iodotyrosine treatment Ovary activity of bees following supplementation with the dopamine synthesis inhibitor iodotyrosine at different doses in **A**. the presence of QMP (QMP+) and **B**. the absence (QMP-). Ovary activity is repressed by both QMP and iodotyrosine. Ovary activity is shown as the proportion of bees within each treatment group with each ovary score (ranging from 0-3; Figure 2.2), where a higher score (darker shade) relates to a higher degree of development. N values for each treatment pooled across three replicates are displayed at the base of each bar. Statistical significance of pairwise comparisons is denoted by bars not sharing a letter (CLMM *post-hoc* least squares means, p<0.05). See Appendix G, Table G.1 for a full table of all *post-hoc* pairwise comparisons.

made within QMP+ and QMP- treatment groups and between iodotyrosine controls. While model likelihood comparisons also indicated an overall significant effect of iodotyrosine supplementation on solution intake (LMM; $\chi 2 = 18.05$, df = 6, 233, p<0.01), *post-hoc* tests revealed no significant differences between iodotyrosine doses within each QMP treatment group. These *post-hoc* pairwise comparisons are displayed in Appendix G (Table G.2).

Survival rates were also consistent across treatment groups, with the notable exception of 0.5 mg/ mL iodotyrosine for which there was a marked increase in mortality (Figure 4.7). CPH model likelihood comparisons showed survival to be significantly affected by QMP (CPH; $\chi 2 = 12.76$, df = 4, p<0.05) and more so by iodotyrosine (CPH; $\chi 2 = 359.17$, df = 6, p<0.0001), though their interaction was not significantly significant (CPH; $\chi 2 = 6.22$, df = 3, p = 0.10). However, in pairwise *post-hoc* comparisons statistical significance



Figure 4.7 Survival and food and water intake rates of iodotyrosine treated bees

A-B Survival probability distributions of cages of 80 bees over 10 days of iodotyrosine treatments **A.** in the presence of QMP (QMP+) and **B.** in the absence (QMP-). Survival was negatively affected by iodotyrosine treatment at doses of 0.5 mg/mL only. Data shown consists of three replicates of each treatment cage. Y axis has been truncated to begin at a survival probability of 0.5. A complete list of *post-hoc* pairwise comparisons found in Appendix G, Table G.3 **C.** Mean water intake and **D**. mean food intake at different iodotyrosine supplementation doses. Food and water intake was recorded daily for each cage (n = 10 days), each of which had three replicates (n = 30 for each treatment). Food intake was consistent across treatments as denoted by "n.s" (LMMs), while water intake was reduced in QMP+ bees. Statistical significance of *post-hoc* pairwise comparisons is found in Appendix G, Table G.2

lay only between comparisons with 0.5 mg/mL treatment groups (see Appendix G, Table G.3). The reduced ovary activity seen at 0.5 mg/mL in both the presence and absence of QMP therefore cannot be separated from general toxicity of iodotyrosine at this dose. However, survival rates were not compromised by iodotyrosine supplementation at any other dose tested (Appendix G, Table G.3), indicating that the repression of QL worker

reproduction by iodotyrosine at 0.1 mg/mL was due to the inhibition of dopamine synthesis rather than an artefact of toxicity.

4.4 Discussion

The inhibition of honey bee worker reproduction by QMP enables queens to retain reproductive dominance over her workers (Hoover *et al.*, 2003), a feature which is central to advanced eusociality (Wilson, 1971). In honey bees, dopamine is thought to be a key signalling molecule mediating QMP's constraint of reproduction in workers (Chapter three; Sasaki & Harano, 2010; Sasaki *et al.*, 2021). The inhibition of ovarian development in QR workers has been widely assumed to be downstream of depression of brain dopamine by QMP, though prior to this study just one investigation formed the basis of this assumed causal relationship in honey bees (Dombroski *et al.* 2003).

In contrast with Dombroski *et al.*, (2003), dopamine supplementation at 0.01 mg dopamine/ g food was not found to increase ovary activity levels in QL workers. The ineffectiveness of dopamine supplementation as a promotor of ovarian development in QL workers was consistent across the range of heightened dosage regimes. This discrepancy may be explained by the strains of bees used: the present study used European *A. mellifera*, while Dombroski *et al.*, (2003) used Africanized *A. mellifera*, a strain known to activate their ovaries more readily (Zillikens *et al.*, 1998). Alternatively, differences in the diet supplied to bees may explain this inconsistency. In this study, bees received a diet higher in protein (see Chapter two, section 2.1.1) than used by Dombroski *et al.* (2003), which may have facilitated ovary activation to such a degree that which dopamine supplementation was unable to elevate any further.

However, aspects of the experimental design may also contribute towards the results reported by Dombroski *et al.*, (2003). Notably, the scoring of ovary activation level was not carried out blinded to dopamine treatment, meaning unconscious bias was not eliminated from the process.

As well as dopamine, administration of its precursor L-dopa was similarly not found promote ovarian development in QL workers at any dose tested. However, L-dopa supplementation also did not increase brain levels of dopamine. Decarboxylation of Ldopa to dopamine is accomplished by the enzymatic action of dopa decarboxylase (Figure 4.1; Neckameyer, 1996). It is possible that this enzyme is the rate limiting step in the biosynthesis of dopamine from L-dopa, with expression levels in the brain being the factor limiting the production of dopamine as opposed to L-dopa availability. Alternatively, oral intake of L-dopa may not have been directed to the brain, instead elevating dopamine titres in other tissues. To my knowledge, there are no comparable studies assessing the trajectory of dopamine biosynthesis from orally administered Ldopa in invertebrates. However, dopamine supplementation is commonly achieved through treatment with L-dopa as a therapy for Parkinson's Disease in humans (Nagatsua & Sawadab, 2009). Whilst L-dopa is better able to cross the blood brain barrier than dopamine (Haddad et al., 2018), the inclusion of additional pharmacological components is still required for effective transportation into the brain (Nagatsua & Sawadab, 2009). Assessment of dopamine levels in whole bees following L-dopa supplementation is needed to confirm whether this method of administration increases dopamine across a range of tissues. Because of its susceptibility to oxidation, it may be that the effects of L-dopa are transient or simply non-existent (Pendleton *et al.*, 1996). If L-dopa treatment does increase dopamine levels in whole bees, the next steps would be to elucidate the tissue-specific effects. Likely destinations of orally administered Ldopa would be the gut, the hemolymph, or, as reported by Neckameyer (1996) for D. *melanogaster*, the central nervous system (CNS).

Assessment of internal dopamine levels of bees following treatment with dopamine itself would also be a useful tool to validate if, and where, elevations of dopamine levels occur. In attempting to replicate their findings, dopamine was administered to workers in their food using the methodology reported by Dombroski *et al.*, (2003). Effects on internal dopamine levels were not measured by Dombroski *et al.*, (2003), thus this would be key future work to validate the efficacy of this administration method for not only

this study but for previously published work. Similarly to L-dopa, dopamine is also highly susceptible to oxidation (National Center for Biotechnology Information, 2022), hence any effects on internal dopamine levels are likely to be transient. A possible alternative method which may lead to more stable elevations in dopamine levels would be to inhibit the degradation of dopamine by knocking down the expression of dopamine-N-acetyltransferase (the enzyme responsible for dopamine breakdown) using RNA interference. This being said, provision of the paper wasp *P. chinesis* with 1 mg/mL dopamine-spiked sucrose solution over a 10-day period reportedly increased brain dopamine levels (Sasaki *et al.*, 2009), therefore this method may be expected to have the same effect in honey bees. Nonetheless, an investigation into the efficacy of dietary dopamine supplementation as a way to increase dopamine levels in this context would validate the use of this methodology in future manipulative experiments.

Despite uncertainty over the efficacy of dopamine supplementation methods, support for an involvement of dopamine in mediating ovarian development in honey bee workers comes from the reductions in ovary activity seen following inhibition of dopamine biosynthesis with iodotyrosine. Dopamine synthesis from the amino acid tyrosine requires hydroxylation to L-Dopa, primarily via the enzyme tyrosine hydroxylase (Figure 4.1). As an inhibitor of tyrosine hydroxylase, oral iodotyrosine application has been demonstrated to depress dopamine levels in whole *D. melanogaster* (Neckameyer, 1996), though tissue-specific effects are unclear. At a dose of 0.1 mg/mL, iodotyrosine treatment caused a reduction in ovarian development in honey bees with no effect on mortality, thus it is assumed that the repression of ovary activity is in response to dopamine depression specifically as opposed to general toxicity.

Reduced reproductive capacity following experimental depression of dopamine has been reported in other species, suggesting a common necessity of dopamine for basic reproductive function. In *D. melanogaster,* inhibition of dopamine synthesis resulted in inhibited ovarian development (Neckameyer, 1996) and failure to produce viable progeny (Pendleton *et al.*, 1996), effects which were both reversed by co-treatment with L-dopa (Neckameyer, 1996; Pendleton *et al.*, 1996). Similarly, iodotyrosine treatment at 0.15 mg/mL reduced numbers of oocytes and deposited eggs produced by virgin fire ant females, and this was also rescued by co-treatment with 0.15 mg/mL L-dopa (Boulay *et al.*, 2001). The specific function of dopamine in the regulation of reproduction in honey bees is unclear. In some insects, dopamine is thought to stimulate the production and release of the gonadotropin JH by the corpora allata (CA), as is the case in the cockroach *Blatella germanica* (Cassier *et al.*, 1993), the larval tobacco hornworm *Manduca sexta* (Granger *et al.*, 1996), and *Drosophila* species (Gruntenko *et al.*, 2005, 2007). However, JH does not appear to function as a gonadotropin in honey bees, and it is also unclear whether it is under regulation by dopamine (Rodrigues & Flatt, 2016). Although dopamine appears to be required for reproduction in honey bees, its pathway of action is likely to differ to that seen in solitary insects, perhaps involving direct interaction with the ovary (Chapter three).

Regardless of the underlying mechanism, dopamine's apparent positive regulatory role in ovarian development in honey bees makes it a likely target for QMP in the regulation of worker reproduction. QMP is known to depress brain dopamine levels (Beggs *et al.*, 2007; Harris & Woodring, 1995; Sasaki & Nagao, 2001), with HVA in particular being the component at least partially responsible for this depression (Beggs & Mercer, 2009). Here, it is demonstrated that treatment of QL workers with a dopamine synthesis inhibitor led to reductions in ovary activity comparable with those induced by QMP, directly implicating QMP's lowering of dopamine as having a direct negative effect on reproduction in workers.

If dopamine levels are a major factor governing ovarian development, this may appear to be at odds with the finding that dopamine supplementation did not overcome QMP's inhibitory effects. Possible limitations of dopamine supplementation methods that may play into this have been discussed above. However, it is perhaps unsurprising that QMP's repression cannot be overcome simply via the restoration of dopamine given the multiple levels of functional redundancy within QMP's components (Princen *et al.*, 2019). QMP contains a blend of five major compounds: (2E)-9-oxo-dec-2-enoic acid (9-ODA), both enantiomers of (2E)-9-hydroxydec-2-enoic acid (9-HDA), methyl-4hydroxybenzoate (HOB) and the component responsible for lowering dopamine, homovanillyl alcohol (HVA) (Slessor *et al.*, 1988). Exposure of honey bee workers to HVA alone results in partial inhibition of reproduction (Princen *et al.*, 2019), while exposure to 9-ODA or 9-HDA causes equivalent levels of repression to that seen with the fivecomponent blend (Princen *et al.*, 2019). This indicates that the depression of brain dopamine by HVA is likely just one process among several redundant mechanisms acting to constrain worker reproduction in the presence of QMP. Two competing evolutionary theories may explain this existence of multiple redundant mechanisms – the queen signal hypothesis and the queen control hypothesis (refer Section 1.4, Introduction). Support for each theory in light of these findings is discussed further in Section 7.5, General Discussion.

In QL workers, the loss of QMP's depressive effects on dopamine is presumably a factor which enables ovarian development in some workers in the absence of a queen. Attempts to further elevate dopamine in QL workers in this study did not increase rates of ovary activation (in contrast with Dombroski et al. 2003), though this is perhaps unsurprising. While a minimum level of dopamine has been shown to be necessary for normal ovarian development (i.e. in Drosophila; Neckameyer, 1996; Pendleton et al., 1996), it is unlikely the limiting factor in the absence of QMP's depressive effects. For instance, in a QL hive, dopamine levels in workers are no longer being suppressed by QMP, yet still only 24-30% of workers respond to the queen's absence by activating their ovaries (Jay, 1968; Miller & Ratnieks, 2001). This is partly because the propensity of individual workers to activate their ovaries in a QL setting is under the influence of additional factors, including ovariole number (Ronai et al., 2017) and genetic relatedness to the queen (Rojek & Kuszewska, 2022). This is illustrative of the fact that multiple complex mechanisms interact to govern ovarian development in adult worker honey bees. While I have found evidence to support the idea that QMP's modulation of dopamine plays a part in maintaining the sterility of honey bee workers, it also highlights that this single signalling pathway is just one among several different reproductive constraints (refer Section 1.2, Introduction).

In summary, this chapter demonstrates that dopamine is likely required for ovarian development in adult worker honey bees, supporting the hypothesis that QMP's control over dopamine titres (demonstrated in Chapter three and in previously published studies) is a pathway involved in the queen's repression of worker reproduction. Due to the uncertainty over the final destination of dopamine supplementation treatments (i.e. whether dopamine was elevated in the brain, the hemolymph, or not at all following treatment), this chapter cannot shed light on whether dopamine's role in reproduction is driven by levels in the brain, in the hemolymph, or a combination of both (Chapter three). As such, the findings from this chapter could be interpreted with more nuance if the effects of dopamine supplementation and inhibition treatments on biogenic amine titres in specific tissues were known. Investigation into the tissue-specific effects of the dopamine supplementation and inhibition methods used in this study on biogenic amine titres is therefore also a key area for future research.

In Chapter three, a role of octopamine in mediating reproductive constraint was also speculated, and future work addressing the causal relationship between octopamine titres and ovary development, as has been carried out here for dopamine, would also be a key avenue for future research.

Chapter 5

Dose-dependent effects of QMP on reproduction

5.1 Introduction

The monopolization of reproduction by one or several reproductive females in a colony is a defining feature of eusociality (Wilson, 1975. The reproductive skew in honey bee societies is particularly extreme, with colonies of tens of thousands of facultatively sterile female workers headed by a single highly fecund queen responsible for laying up to 1,500 eggs per day (Snodgrass, 1956). Irreversible differences in queen-worker physiology are established during larval development (Cameron *et al.*, 2013; Kucharski *et al.*, 2008), which facilitate this high reproductive output from queens. In a process triggered by the feeding of royal jelly (Dixon & Shuel, 1963), queens develop considerably larger ovaries than workers, consisting of 120-200 ovarioles and possessing a fully developed spermatheca (Figure 1.1; Snodgrass, 1956; Linksvayer *et al.*, 2011). Worker ovaries, on the other hand, lack a functional spermatheca and consist of just 2-12 ovarioles (Figure 1.1; Snodgrass 1956), a result of programmed cell death in early development (Capella & Hartfelder, 1998; Hartfelder & Steinbrück, 1997). These developmental constraints severely reduce the reproductive capacity of workers relative to queens, though workers still retain some ability to reproduce.

In addition to developmental constraints on worker reproduction, queens impose further restrictions on worker reproduction during adulthood via QMP (Hoover *et al.*, 2003). There is considerable debate as to whether QMP, and queen pheromones more generally, chemically control worker reproduction or simply signal the presence of a fecund queen, two hypotheses known as the 'queen control' and 'queen signal' hypothesis (refer Section 1.4, General Introduction; Keller & Nonacs, 1993; Kocher & Grozinger, 2011; le Conte & Hefetz, 2008; Strauss *et al.*, 2008). Under the honest signal hypothesis, it is thought that workers under the presence of a fecund queen benefit more from rearing the queen's brood (i.e. their siblings) than attempting to reproduce

themselves (Keller & Nonacs, 1993). As such, queen pheromones should be selected for their ability to reliably or 'honestly' signal the queen's fecundity (Oi *et al.*, 2015). On the other hand, the queen control hypothesis predicts that QMP has evolved to target pathways mediating reproduction to chemically enforce sterility in workers (Reviewed by Kocher & Grozinger, 2011).

In line with the queen signal hypothesis, the production of QMP is tightly linked with the queen's reproductive state, depicting QMP as an honest signal of fecundity (Kocher *et al.*, 2009). For instance, the quantity and chemical composition of QMP produced differs between virgin and mated queens (Plettner *et al.*, 1997). QMP production also varies between laying vs non-laying mated queens (Kocher *et al.*, 2008), with the number of drones she has mated with (Richard *et al.*, 2007), and with semen insemination volume (Niño *et al.*, 2012).

An alternative though not necessarily mutually exclusive hypothesis, the queen control hypothesis, predicts that queen pheromones have evolved to target pathways mediating reproduction to exert direct control over workers (Reviewed by Kocher & Grozinger, 2011). QMP is highly derived from the ancestral class of cuticular hydrocarbons that function as queen pheromones in other hymenopterans (refer Section 1.3, General Introduction). Unlike these ancestral pheromones, QMP inhibits the reproduction not only of honey bees, but of a broad range of unrelated species (Lovegrove *et al.*, 2019). For instance, QMP has been found to reduce reproductive capacity in virgin D. melanogaster females (Camiletti et al., 2013), a species of ant (Carlisle & Butler, 1956), a house fly (Nayar, 1963), a termite (Hrdy et al., 1960), and a prawn (Carlisle & Butler, 1956). These broad phylogenetic effects suggest that QMP may induce sterility by targeting pre-existing ancestral pathways involved in regulating reproduction, depicting this pheromone as an agent of reproductive control. Though much is still unknown about the process by which QMP constrains worker reproduction, the pheromone's modulation of dopamine (and possibly octopamine; Chapter three) appears to be a factor inhibiting ovary development in workers (Chapter four). In fact, the direct modulation of brain dopamine by the QMP component HVA is sometimes cited in support of the reproductive control hypothesis (Beggs *et al.*, 2007; Beggs & Mercer, 2009; Kocher & Grozinger, 2011), as it points to QMP being able to manipulate an important signalling molecule to induce sterility. If QMP directly manipulates fundamental reproductive pathways, an unexplored question is how the queen is able to avoid self-repression with her own pheromone. Reconciling this apparent contradiction is thus a useful addition to the debate surrounding the queen control vs queen signal hypotheses.

The mean quantity of QMP found in the mandibular glands of a laying queen is 220 μ g, consisting of 150 μ g 9-ODA, 55 μ g 9-HDA, 13 μ g HOB, and 1.5 μ g HVA, or one queen equivalent (QE) (Slessor *et al.*, 1988). One QE is sufficient to induce the full retinue response of workers (Slessor *et al.*, 1988), inhibit the production of queen cells (Winston *et al.*, 1990), and repress the ovary activity of tens of thousands of workers in a hive (Hoover *et al.*, 2003). Although the majority of the QMP produced by the queen is disseminated across the colony by workers (Naumann *et al.*, 1991), 36% of these daily secretions are re-internalised by the queen, most likely via ingestion (Figure 1; Naumann *et al.*, 1991). Despite this, the queen still manages to retain extraordinarily high levels of fecundity, being seemingly immune to the effects of her own pheromone.

One possible explanation for QMP's lack of inhibitory effects in queens may lie in the differing degrees to which queens and workers are exposed. Once disseminated throughout the tens of thousands of workers in the hive, the dose of QMP received by individual workers is presumably very small compared with the approximately 0.36 QE internalised by queens (36% of 1 QE; Naumann *et al.*, 1991). Hormesis, a commonly observed dose-response phenomenon, is a term predominantly applied to the field of medicine describing biphasic dose-response relationships of U-shaped or inverted U-shaped nature (Calabrese & Baldwin, 2002). The repressive effects of QMP may adhere to this, whereby in low doses it inhibits ovary development, but is ineffective at high doses due to overstimulation or over compensatory mechanisms. Here, I test the hypothesis that QMP's inhibition of reproduction is biphasic with respect to dose by



Figure 5.1 Schematic of modelled QMP transmission throughout the hive. Adapted from Naumann *et al.*, (1991).

exposing workers to QMP at a range of concentrations and observing the impact on their ovary development.

To gain further insight into possible mechanisms involved in this U-shaped dose response, I also examine the effect of different QMP doses on some of our bestunderstood processes mediating reproductive constraint (refer Section 1.5, General Introduction). Based on current understanding, active Notch signalling in the ovary is the proximate inhibitory mechanism in response to low-dose QMP exposure (Section 1.5.1; Duncan *et al.*, 2016), and the depression of dopamine and octopamine is likely involved in the upstream sequence of events (Chapter three, Chapter four). Interestingly, despite continual QMP exposure, brain dopamine levels are elevated in queens compared with queenright (QR) workers (Sasaki *et al.*, 2012), and Notch signalling is inactive in the ovary (Duncan *et al.*, 2016). This indicates that at elevated doses, QMP's effects on biogenic amines and Notch activity may be lost, which may underpin the lack of reproductive inhibition. To test the hypothesis that QMP's effects on Notch activity in the ovaries of workers exposed to different doses of QMP, as well as brain levels of dopamine and octopamine and octopamine. To my knowledge, this study is the first of its kind to ask the question of how honey bee queens evade self-repression by QMP, and aims to empirically test one potential mechanism explaining the caste-specific responses of queens and workers to this pheromone.

5.2 Methods

5.2.1 Microcolony set-up and experiment overview

A. mellifera hives were maintained as stated (Section 2.1.1) and laboratory microcolonies were setup and maintained in cages as described previously (Section 2.1.1). Experimental cages each contained 80 workers and were exposed to QMP at varying doses. After 10 days the ovaries of all remaining workers in each cage were dissected, imaged and scored to assess ovary activity levels as described previously (Section 2.2). Ovaries and brains from a subset of five randomly selected bees per cage were collected, snap frozen in liquid nitrogen and stored at -80°C for later molecular analysis. Four experimental replicates were carried out for each cage between July-August 2020.

5.2.2 QMP treatments

Synthetic QMP was supplied as described previously (Section 2.1.2) but at a wider range of concentrations. As in other chapters (Chapter three, Chapter four, Chapter six) and previous work (Duncan *et al.*, 2016), QMP was supplied to some treatment cages at 0.01 QE/ μ L, as this concentration has been shown to inhibit the reproduction of 80-100 workers in a microcolony. Other cages received QMP at the elevated doses of 0.1 and 1 QE/ μ L, and QL microcolonies received just solvent control (0 QE/ μ L). Microcolonies received QMP treatments as 10 μ L droplets which were replaced daily (as described in Section 2.1.2). Therefore, the total QE received by microcolonies each day was 0, 0.1, 1 or 10.

5.2.3 HPLC-FLD

To test the hypothesis that QMP's modulation of biogenic amines is dose-dependent, dopamine and octopamine were quantified in brain samples using High Performance

Liquid Chromatography with Fluorescent Detection (HPLC-FLD) as described previously (Section 2.3).

5.2.4 Quantitative RT-PCR

To determine whether QMP's regulation of oogenesis via Notch signalling is dosespecific, expression of the genes *bHLH2* and *Her* were measured in the ovary as a proxy for Notch activity using RT-qPCR. These two E(spl)-C genes (enhancer of split complex; Duncan & Dearden, 2010; Jennings *et al.*, 1994) are targets of Notch signalling, and have previously been implicated in Notch-mediated ovarian development (Duncan *et al.*, 2016).

RNA was extracted from individual snap-frozen pairs of ovaries using RNAqueous micro kits (ThermoFischer) with removal of gDNA with DNAse treatment. 100 ng RNA or the maximum amount obtained was reverse transcribed and a 1 in 5 dilution of this cDNA was used as a template for RT-qPCR. Differences in cDNA quantities between samples are standardised by comparing the relative expression of genes of interest to that of the reference or 'housekeeper' gene (described in full below). Oligonucleotide primers used were also designed as described in (Duncan *et al.*, 2016). Oligonucleotide primer sequences are provided in Appendix H, Table H.1.

RT-qPCR was carried out using a BioRad CFX RT-PCR detection system. Each reaction contained 15 μ L SsoFast EvaGreen PCR mastermix, 5 ng of cDNA, and 300 nM of each primer, and each measurement was carried out in duplicate. Gene expression for every condition (every ovary score within each QMP dose treatment) was measured for three biological replicates. However, it was not possible to obtain three biological replicates of every single combination of ovary score and QMP dose due to the frequency distribution of ovary scores within each QMP treatment (i.e. Figure 5.4). For instance, no workers sampled for RT-qPCR receiving 0.01 QE/ μ L QMP had an ovary score of 3 due to the inhibition of ovary activity. A breakdown of the number of biological replicates for each condition is provided in Appendix H, Table H.2.

Expression of target genes *Her* and *bHLH2* was normalised to the relative quantities of the reference gene *Rpn2* using the Pfaffl method (Pfaffl, 2001). The sample with the highest Ct value was used as a calibrator sample to calculate Δ Ct. Initial reference gene selection based on their stability of expression among ovary samples was carried out as in (Duncan *et al.*, 2016). Insufficient RNA was obtained from individual ovaries to normalise expression of genes of interest against more than one housekeeping gene. The use of *Rpn2* only as a housekeeping gene was validated by an analysis of previous gene expression data normalised to either *Rpn2* alone or the geometric mean of *Rpn2* with a second housekeeper, *mRPL44* (Appendix H, Figure H.1). Gene expression data was log₁₀ transformed prior to analysis.

5.2.5 Data analysis

All statistical analyses were carried out using R. Details of packages used for analysis are found in section 2.4. For all analyses, QMP dose was considered an ordinal variable, as the number of doses tested was insufficient to be considered a continual variable.

Food intake and survival rates were compared across treatment conditions to establish whether there were any compounding effects of QMP doses. Mean food intake per bee per day was compared across treatment conditions by fitting LMMs with QMP dose as the fixed effect. To account for repeated measures, day was included as a random effect and nested within replicate. Statistical significance of QMP dose on food intake was determined by comparing the likelihood ratio of the full model to the null model containing no fixed effect. Survival distributions were compared between treatment conditions using CPH models with mixed effects, where QMP dose was included as a fixed effect and replicate as a random effect. The assumption of proportional hazards was verified through visual examination of correlated scaled Schoenfield residuals with time to test for independence. CPH models were used to predict Hazard Ratios (HR) and confidence intervals (CI) for fixed effects. HRs are presented as HR (±95% CI).

To compare ovary activation rates across different QMP doses, CLMMs were fitted with ovary score as the ordinal response variable, QMP dose as the fixed effect and replicate as the random effect. Statistical significance was obtained by comparing the likelihood ratio of the full model to the null model containing no fixed effect. *Post-hoc* testing was carried out by computing least-squares means to determine the significance of all pairwise comparisons, and P-values were Tukey-adjusted.

Differences in gene expression between ovaries of each score within each QMP dose treatment were analysed using LMMs with ovary score, QMP dose and their interaction as fixed effects and replicate as a random effect. Statistical significance of each fixed effect was computed by comparing likelihood ratios between the maximal model and the model without the fixed effect of interest. *Post-hoc* testing was carried out by computing estimated marginal means on the best-fitting model as indicated by AIC score.

Brain levels of dopamine and octopamine were compared between QMP dose treatments by fitting GLMMs with a gamma distribution and an inverse link function for each biogenic amine. QMP dose was included as the fixed effect and replicate and sample preparation block were included as random effects. The statistical significance of QMP dose on each biogenic amine was obtained by comparing the likelihood ratio of the full model to that of the null model containing no fixed effects.

5.3 Results

5.3.1 QMP treatments do not affect food intake or survival

Food intake was not affected by QMP dose (Figure 5.2; LMMs; $\chi 2 = 1.33$, df = 3,36, p = 0.72), confirming that there were no nutritional differences in workers between treatment conditions that could explain differences in ovary activity. Across all four replicates, survival rates of caged workers were very high, with cage survival ranging between 93-100% (Figure 5.2). Exposure to QMP at 0.01 QE/µL and 1 QE/µL was associated with respective increased mortality risks of 12% and 30% compared with QL workers (CPH; 0.01 QE/µL: HR = 1.12 ± 0.52, z = 2.17, p <0.05; 1 QE/µL: HR = 1.30 ± 0.51, z = 2.58, p <0.05), though due to the overall high survival in each treatment condition this is not deemed disruptive to experimental results. Full *post-hoc* pairwise comparisons computed using estimated marginal means can be found in Table 5.1.



Figure 5.2 Effects of QMP doses on food intake and survival. A. Mean food intake per bee per day at different QMP dose treatments. Food intake was recorded daily for each treatment cage (n = 10 days), for which there were four replicates (n = 40 for each treatment). Food intake was consistent across different levels of QMP exposure as denoted by 'n.s' (LMMs). **B.** Survival probability distributions of cages of 80 bees over 10 days of exposure to different concentrations of QMP. Data shown represents the pooled survival distributions over four experimental replicates. See table 5.1 for *post-hoc* comparisons of survival between QMP treatments

Contrast (QE/µL)	Hazard Ratio	SE	df	Z-ratio	P-value
0-0.01	-1.12	0.52	Inf	-2.17	0.13
0-0.1	-0.59	0.56	Inf	-1.06	0.71
0-1	-1.30	0.51	Inf	-2.58	0.05*
0.01-0.1	0.53	0.42	Inf	1.25	0.59
0.01-1	-0.19	0.35	Inf	-0.53	0.95
0.1-1	-0.71	0.41	Inf	-1.75	0.30

Table 5.1 list of *post-hoc* pairwise comparisons for survival rates between differentQMP exposure treatments.

5.3.2 Brain dopamine and octopamine are not affected by QMP dose

It was hypothesised that QMP's depression of biogenic amines may be lost at high doses, leading to misinterpretation of QMP's signal and impeding reproductive constraint. No evidence was found in support of this hypothesis, as brain levels of dopamine and octopamine in 10-day old workers were not affected by QMP at any dose (Figure 5.3; GLMMs; Dopamine: $\chi^2 = 2.79$, df = 3,74, p=0.42; Octopamine: $\chi^2 = 2.83$, df = 3,74, p=0.42). However, this includes a lack of expected depression by QMP at the low dose (0.01 QE/µL), as has been observed previously for octopamine in the laboratory (Chapter three) and for both dopamine and octopamine in QR hives (Chapter three).



Figure 5.3 Brain biogenic amine levels at different QMP doses. HPLC-FLD quantification of biogenic amines in the brains of 10-day old worker bees following exposure to QMP at a range of concentrations. Brains were sampled from five randomly selected bees per treatment cage over four replicates (n=20 total for each QMP dose). Pairwise GLMM model comparisons showed that levels of **A.** dopamine and **B.** octopamine were not significantly affected by QMP dose, as denoted by "n.s".

5.3.3 QMP's inhibition of worker reproduction is lost at high doses

Ovary development was affected by QMP in a dose-dependent manner (Figure 5.4). CLMM model comparisons demonstrated that QMP dose significantly affected ovary activity levels in workers (CLMM; χ^2 = 62.16, df = 3,1034, p <0.001). Consistent with other work (Chapter three, Chapter four, Chapter six, Duncan *et al.*, 2016), worker reproduction was repressed by synthetic QMP at a dose of 0.01 QE/ µL. At this lower dose, the percentage of workers possessing fully mature oocytes (ovary score of 3; Figure 2.2) was reduced from 30% in QL workers to 9%. *Post-hoc* testing using least marginal means on the full model showed this decrease in ovary activity levels between 0 QE/µL and 0.01 QE/µL was statistically significant (Z = -6.808, p <0.001). All *post-hoc* pairwise comparisons can be found in Table 5.2.



Figure 5.4 Dose-specific effects of QMP on ovary activity. Ovary activity levels of 10-day old workers maintained in the absence of QMP (0 QE/ μ L – QL) or in the presence of QMP at different concentrations. At 0.01 QE/ μ L, QMP represses ovary activity relative to QL workers, but this repression subsides with increasing QMP dose. Ovary activity is presented as the percentage of workers within each treatment condition with each ovary score (ranging from 0-3), where a higher score (darker shade) relates to a higher degree of development. N values for each treatment represent totals from four pooled replicates and are displayed at the base of each bar. Statistically significant pairwise comparisons (p <0.05) are denoted by bars not sharing a letter, where *post-hoc* testing was carried out on the full CLMM using least squares means. For full results from *post-hoc* testing see Table 5.2.

Contrast (QE/µL)	Estimate	SE	df	Z-ratio	P-value
0-0.01	-1.08	0.16	Inf	-6.81	<0.0001***
0-0.1	-0.55	0.16	Inf	-3.47	0.003**
0-1	-0.02	0.16	Inf	-0.14	0.999
0.01-0.1	0.53	0.16	Inf	3.37	0.004**
0.01-1	1.06	0.16	Inf	6.47	<0.0001***
0.1-1	0.53	0.16	Inf	3.22	0.007**

Table 5.2 list of *post-hoc* pairwise comparisons for ovary score distributions between different QMP exposure treatments.

Interestingly, as QMP dose was increased above 0.01 QE/µL, its repression of worker ovary activity subsided. At a dose of 1 QE/µL QMP's inhibitory effects were completely absent, and *post-hoc* testing showed that ovary activity levels did not significantly differ at this high dose from that seen at 0 QE/µL (Z = -0.14, p = 0.999). The intermediate QMP dose of 0.1 QE/µL represented an intermediate level of repression lying between full repression at 0.01 QE/µL and the absence of repression at 1 QE/µL (see Figure 5.4 and Table 5.2).

5.3.4 Notch signalling underlies reproductive constraint irrespective of QMP dose

In keeping with previous work (Duncan *et al.*, 2016), Notch signalling appears to be the process actively constraining ovarian development (Figure 5.5). While this has been demonstrated with pooled samples before, this is the first time this has been shown for individual bees. Expression of *Her* and *bHLH2* in the ovary, proxies for Notch activity, negatively correlated with the degree of ovarian development irrespective of QMP dose received (Figure 5.5). Model comparisons demonstrated QMP dose to have no significant effect on expression of either *Her* (LMMs; $\chi^2 = 8.10$, df = 9,28, p = 0.52) or *bHLH2* (LMMs; $\chi^2 = 7.40$, df = 9, 28, p = 0.60). Expression of both genes was instead more closely linked with ovary score, and this was statistically significant (LMMs; *Her:* $\chi^2 = 17.79$, df = 9, 28, p <0.05; *bHLH2*: $\chi^2 = 28.23$, df = 9,28, p <0.001). The best-fitting model for the data (as determined using AIC values) included just ovary score as the sole fixed effect, hence this is also presented graphically with pooled QMP doses (Figure 5.5).

reduced model and are presented in both Figure 5.5 and Table 5.3. There was no evidence for an interaction effect between QMP dose and ovary score on *Her* or *bHLH2* expression (LMMs; *Her*: χ^2 = 6.71, df = 6,31, p=0.35; *bHLH2*: χ^2 = 6.39, df = 6,31, p=0.38), indicating that Notch activity consistently regulates development in the ovarian tissue regardless of upstream changes to environmental cues or signalling pathways that may arise from different QMP doses.



Figure 5.5 *Her* and *bHLH2* ovary expression by ovary score and QMP dose. Expression of two genes from the E(spl)-C complex in worker ovaries measured using RT-qPCR as a proxy for Notch activity. Gene expression of **A.** *bHLH2* and **C.** *Her* was measured for each ovary score within each QMP dose treatment condition in triplicate where possible (see supplementary Table H.2, Appendix H for sample sizes). LMMs revealed ovary score but not QMP dose to significantly affect expression of both genes, hence expression of **B.** *bHLH2* and **D.** *Her* is also presented for each ovary with QMP dose treatment conditions pooled. Relative expression of both *bHLH2* and *Her* decrease with increasing ovary scores. *Post-hoc* tests were carried out using estimated marginal means on the best fitting model (lowest AIC) with ovary score as the sole fixed effect. Statistically significant pairwise comparisons (p<0.05) are denoted on plots **B.** and **D.** by boxplots not sharing a letter. *Post-hoc* pairwise comparisons are also reported in Table 5.3

bHLH2								
Contrast	Estimate	SE	df	T-ratio	P-value			
0-1	0.11	0.07	33.9	1.69	0.34			
0-2	0.27	0.07	33.8	3.93	0.002**			
0-3	0.31	0.08	33.7	4.14	0.001**			
1-2	0.16	0.05	34.0	2.90	0.03*			
1-3	0.20	0.06	32.9	3.29	0.01*			
2-3	0.04	0.06	32.7	0.73	0.89			
	Her							
Contrast	Estimate	SE	df	T-ratio	P-value			
0-1	0.03	0.16	32.3	0.21	0.99			
0-2	0.25	0.16	32.2	1.57	0.41			
0-3	0.45	0.18	32.6	2.52	0.08			
1-2	0.22	0.13	31.4	1.71	0.34			
1-3	0.42	0.15	30.7	2.89	0.03*			
2-3	0.20	0.15	31.0	1.40	0.51			

Table 5.3 List of *post-hoc* pairwise comparisons for the expression of *bHLH2* and *Her* between different ovary scores

5.4 Discussion

The absence of self-sterilizing effects of QMP in queens is a phenomenon which has received little research attention, yet is central to the validity of the queen control hypothesis; If QMP is a direct suppressor of reproduction, what mechanisms protect queen fecundity? Here, I provide empirical evidence that QMP's inhibition of reproduction is lost at high doses, consistent with the hypothesis that differential exposure levels between queens and workers is a mechanism underlying caste-specific repressive effects of QMP. This provides our first insight into one proposed mechanism by which queens evade self-sterilization, though alternative mechanisms will also be discussed which remain to be tested.

The inhibition of reproduction by QMP at low doses but loss of function at high doses resembles a hormetic response (Figure 5.6). Characterized by biphasic dose response relationships, hormesis is generally driven by overstimulation or over compensatory mechanisms (Calabrese & Baldwin, 2002). The processes underlying QMP's mediation of reproduction are complex and involve multiple stages (refer General Introduction, Section 1.5), and such a dose-response relationship could occur at any point along this





signalling pathway to drive the loss of reproductive inhibition at high doses. In the ovary itself, Notch signalling is the mechanism that actively blocks development in workers exposed to QMP at normal levels (Duncan *et al.*, 2016). In the absence of QMP, Notch activity is negatively correlated with worker ovary development, but is lowest in queens in spite of their re-internalisation of QMP (Duncan *et al.*, 2016). It is tempting to conclude that dysfunctional Notch activity at high QMP doses facilitates queen ovary activity. However, the finding that Notch signalling was the mechanism controlling ovary development even in workers exposed to high doses of QMP suggests that the hormetic processes driving the loss of inhibitory effects of QMP at high doses occur upstream.

As previously discussed, the depression of dopamine and possibly octopamine is involved in mediating QMP's repression of ovary development (Chapter three, Chapter four). Despite the re-internalisation of QMP, queens have elevated brain dopamine levels relative to workers (Sasaki *et al.*, 2012). It was thus hypothesised that QMP's depression of biogenic amines may be lost at high doses, facilitating ovary development despite the presence of QMP. The QMP component HVA is thought to modulate dopamine levels by directly interacting with AmDop3 receptors (Beggs & Mercer, 2009), hence overstimulation of these receptors by HVA at high QMP doses could lead to disinhibition of dopamine. Brain levels of both dopamine and octopamine were unaffected by QMP at any dose, providing no evidence for an involvement of these biogenic amines in QMP's loss of function at high doses. However, the lack of depressive effects of QMP observed at low doses despite its inhibition of reproduction reinforces previously discussed experimental limitations (Chapter three). In particular, sampling workers on day 10 is likely too late to observe differences in biogenic amines that precede changes in ovary development; Previous work reporting reductions in brain dopamine by QMP sampled laboratory-maintained workers at two-days old (Beggs *et al.*, 2007). A hormetic dose-response of key biogenic amines to QMP therefore cannot be ruled out as the mechanism underlying dose-specific differences in reproductive repression, and future studies sampling workers on day two of different QMP exposure regimes may be a more informative means of testing this.

Although the underlying mechanism is unclear, QMP's demonstrated dose-specific effects on reproduction supports the hypothesis that a hormetic dose-response relationship is a driver of caste-specific differences in QMP's repressive effects. However, this represents just one piece of evidence in favour of this hypothesis, and further study is required to test this more definitively. Firstly, the hypothesis rests heavily on the assumption that queens are exposed to QMP in greater quantities than workers, though empirical support for this is limited. One study in which 36% of QMP secretions were found to be re-internalized by the queen, likely via ingestion, forms the basis of this assumption (Naumann et al., 1991), with no other studies carried out on this subject to my knowledge. However, the authors point out that the estimated queen reinternalisation rate is the weakest element in their model of QMP transmission among the hive due to technical limitations of the model (Naumann et al., 1991). Obtaining an accurate measure of A. QMP re-internalisation rates of queens and B. QMP exposure levels of workers would provide a more robust foundation for the assumed heightened exposure of QMP in queens relative to workers. This would be difficult to study, but could theoretically be achieved using radio-labelled diet components fed to the queen which would be incorporated into QMP during synthesis by the queen.

Another key factor to consider is whether there are differences in the mode of QMP exposure between queens and workers that could explain QMP's caste-specific effects.

85

In workers, QMP is picked up by nurses attending the queen (Allen, 1955) and distributed through the colony via worker-worker interactions such as antennation, grooming and trophallaxis (Naumann *et al.*, 1991). Evidence points to gustatory exposure being a predominant route of QMP detection necessary for the inhibition of ovary activity in honey bees, as preventing direct physical contact with QMP (Lovegrove *et al.*, 2020) or restricting worker-worker trophallaxis in the hive (Katzav-Gozansky *et al.*, 2004) lifts the repression of reproduction. The detection of labelled 9-ODA in the hemolymph and gut of queens suggests that ingestion is the most probable route of QMP detection in queens. However, this has not been empirically tested, and if alternative exposure routes such as absorption through the cuticle are instead the main route of re-internalisation, these discrepancies in QMP uptake between queens and workers could in themselves represent a mechanism of self-sterilization evasion in queens.

The queen signal hypothesis posits that QMP acts as an honest indicator of queen quality (Keller & Nonacs, 1993; Kocher & Grozinger, 2011), and this is supported by empirical evidence demonstrating equivalence between queen QMP production and aspects of reproductive potential (Kocher *et al.*, 2008, 2009; Niño *et al.*, 2012; Plettner *et al.*, 1997; Richard *et al.*, 2007). That QMP is not repressive at high doses presents an interesting possibility on the causality of QMP production and fecundity; Perhaps increased QMP production actually precedes queen ovarian maturity post-mating, and the heightened QMP exposure of queens is a necessary facilitator of oocyte development. Put simply, could QMP production be a cause rather than a consequence of high fecundity in queens? Such a hypothesis is highly speculative, and rests on previously discussed assumptions that remain to be validated such as whether QMP exposure truly is heightened in queens and whether the mode of uptake is comparable to workers. Nonetheless, this is a novel angle from which to consider the evolution of QMP signalling which joins together the queen signal and queen control hypotheses in an intriguing way. Through its dose-specific effects, QMP could hypothetically serve as both a positive

and negative regulator of reproductive control, rendering it also an innately honest signal of fecundity.

A means of testing the causality of QMP exposure and ovary development in queens would be to look to queens themselves. An ideal experiment would use RNA interference (RNAi) to knock down QMP production in queens, allowing external QMP exposure to be manipulated at a range of doses and the effects on queen ovary development examined. However, as the genes involved in the biosynthesis of QMP have not been fully elucidated, the specific genes that would need to be targeted for knock down are unclear. Surgical removal of the mandibular glands, the site of QMP synthesis, would be an alternative approach to restrict queen QMP production. Mandibular gland removal has been carried out successfully in other studies with no detrimental effects to survival, and has been shown to prevent the production of QMP (Maisonnasse, Alaux, *et al.*, 2010).

To conclude, I present evidence that QMP's effects on ovary activity are lost at high doses, consistent with the hypothesis that caste-specific differences in QMP exposure rates underlie the evasion of self-sterilisation in queens. Further study is required at earlier timepoints to capture the hormetic mechanisms preceding this phenotype, with a possible contender being over-stimulation of dopamine receptors by HVA at high QMP doses. It is acknowledged that this proposed mechanism of self-sterilisation evasion rests heavily on the assumed higher queen exposure to QMP, and verification of this would be essential going forward. In addition, manipulative experiments in queens involving the inhibition of QMP production via mandibular gland removal and subsequent assessment of reproductive physiology would be a useful means of testing the relationship between QMP exposure and fecundity in honey bee queens. This is an emerging research area, and further investigations into alternative mechanisms of castespecific differences in QMP's repression would ultimately aid the reconciliation of the contradictions within the queen control hypothesis.

Chapter 6

Investigating the interactivity between QMP and imidacloprid 6.1 Introduction

Insects provide crop pollination services that are integral for food security, with almost half of the 115 leading global food commodities reliant on their pollination (Klein *et al.*, 2007). Honey bees in particular are among the most economically important pollinators in agricultural systems (Gallai *et al.*, 2009). Our reliance on the ecosystem services of honey bees has grown throughout the 21st century: Between 1961 and 2006, dependence on pollinators by the agriculture industry grew by an estimated 50% (Aizen & Harder, 2009). Despite the increased demand, honey bee populations are diminishing, with reports of unsustainably high colony losses from Canada, the US and Europe (Chauzat *et al.*, 2016; Lee *et al.*, 2015; van der Zee *et al.*, 2012; vanEngelsdorp & Meixner, 2010).

Declines of honey bee populations, and other pollinators alike, have been attributed to a combination of environmental stressors with synergistic negative effects on pollinator health (reviewed by Potts *et al.*, 2010). In honey bees, these include (but are not limited to): diseases and parasites (i.e. The parasitic mite Varroa destructor: Martin *et al.*, 1998 and Nosema: (Genersch, 2010), land use change (Woodcock *et al.*, 2017), weather and climate change (VanEngelsdorp *et al.*, 2008), and pesticide use (Tsvetkov *et al.*, 2017; Woodcock *et al.*, 2017). Within these stressors, the widescale use of agrochemicals has been identified as a particular concern to honey bee colonies given their role as agricultural pollinators.

The application of pesticides is a controversial but ultimately necessary agricultural practice to prevent otherwise devastating losses from pest species (Sharma *et al.*, 2019). Chemical substances developed to control herbivorous insect pest populations are frequently picked up by non-target pollinators such as honey bees, who come into contact with these insecticides through their foraging activity. Contaminated floral resources are frequently brought back to the hive, where they present a further

exposure risk to other colony members including young workers, the queen, and developing larvae (Rortais *et al.*, 2005).

Among the most widely used pesticides of recent years are neonicotinoids, a class of neurotoxic insecticides including thiamethoxam, imidacloprid, and clothianidin. These pose a prominent risk to honey bees and other pollinators. Their application has grown rapidly since the early 2000s, replacing organophosphates and pyrethroids to become the most widely used pesticides worldwide as of 2013 (Casida & Durkin, 2013). In response to a flood of high-impact studies reporting negative effects on non-target beneficial insects (e.g. Gill *et al.*, 2012; Henry *et al.*, 2012; Whitehorn *et al.*, 2012), neonicotinoids were banned by the EU in 2020, but may still be used in 'emergency' situations. In the UK, for instance, emergency approval has recently been granted for the use of neonicotinoids on sugar beet crops in 2022 (Barkham, 2022). Neonicotinoids have been shown to persist in the soil, thus even one-off use increases the exposure risk of non-target organisms for years to come (Reviewed by Pietrzak *et al.*, 2020).

Neonicotinoids are applied to crops as a seed dressing which is taken up upon germination and distributed systemically throughout the plant, making them direct targets of herbivorous pests that attack crop tissue. They act as neurotoxins, mimicking the action of the biogenic amine acetylcholine (ACh) by selectively binding to post-synaptic nicotinic aetylcholine receptors (nAChRs) in the insect brain. Usually, synaptic activation is terminated by the AChE enzyme, which rapidly breaks down ACh. However, neonicotinoids are insensitive to the action of AChE so persist in the synapse. The resulting persistent activation of nAChRs leads to convulsions, paralysis and eventual death in target pest species (Casida & Durkin, 2013; Casida & Quistad, 2004). Through exposure to the trace amounts of these pesticides present in pollen and nectar (Dively & Kamel, 2012; Mullin *et al.*, 2010), pollinators such as honey bees are at risk less from acute toxicity, but more from sub-lethal effects of chronic exposure.

The adverse effects of sub-lethal neonicotinoid exposure on honey bees are welldocumented. For example, neonicotinoids have been associated with reduced foraging activity (Schneider *et al.*, 2012), impaired navigation (Fischer *et al.*, 2014), homing-failure (Henry *et al.*, 2012), interrupted motor function (Williamson *et al.*, 2014), impaired learning and memory (Decourtye, Armengaud, *et al.*, 2004), and decreased colony performance and productivity (Sandrock *et al.*, 2014). Despite the research attention accrued by the behavioural effects of sub-lethal neonicotinoid exposure, very little is known about their molecular underpinnings (a pattern true for many pesticides – reviewed by Christen *et al.*, 2018). In particular, although the acute neurotoxic effects of neonicotinoids on target species are well understood, there is a shortage of research into the effects of chronic sub-lethal exposure on the neuroendocrine system in relation to behavioural modulation.

In honey bees, the neuroendocrine system of workers is under the significant additional influence of QMP (i.e. Chapter three). QMP is a multi-functional pheromone that acts as both a releaser of rapid behavioural responses (Wyatt, 2003) and a primer for long-term changes in behaviour and reproductive physiology. As a releaser, QMP stimulates retinue response behaviour, or the attraction of workers to the queen (Slessor *et al.*, 1988; Vergoz *et al.*, 2009). This attraction to the queen (or more specifically, to QMP) ensures workers feed and groom the queen and distribute her pheromones throughout the hive (Slessor *et al.*, 1988; Slessor *et al.*, 2005). Primer functions of QMP include the inhibition of ovary development (Hoover *et al.*, 2003) and the slowed behavioural transition from nurse to forager (Pankiw *et al.*, 1998). These functions of QMP are integral for colony productivity, ensuring queen-worker cooperation by suppressing individual worker reproduction (refer Section 1.2, General Introduction), as well as maintaining the division of labour and temporal polyethism in the hive (Pankiw *et al.*, 1998).

QMP exerts at least some of these varied effects on workers through modulation of neuroendocrine signalling pathways. In particular, biogenic amines have been implicated as key messengers of QMP's presence, inciting downstream effects on physiology and behaviour as discussed in Chapter three and Chapter four. Biogenic amines are a highly conserved class of molecules which function as neuromodulators, neurohormones and neurotransmitters (Evans, 1980), with additional evidence also implicating octopamine and dopamine as circulatory neurohormones (Chapter three). Although it has not been studied in honey bees, neonicotinoids disrupt biogenic amine signalling in other insect species via nAChR mediated dopamine release. The neonicotinoid imidacloprid induced dopamine release in larval D. melanogaster central nervous systems (Pyakurel et al., 2018), and long-term exposure caused a reduction in brain dopamine levels (hypothesised to be due to blockade of nAChR following prolonged exposure; Janner et al., 2021). Octopamine release is also mediated by nAChR stimulation in *D. melanogaster* (Fuenzalida-Uribe *et al.*, 2013), thus octopamine levels may also be altered by neonicotinoids. Disruptions to neural octopaminergic signalling have been proposed as an explanatory mechanism behind some of the behavioural effects of neonicotinoids in honey bees (Farooqui, 2013), though to my knowledge this has not been empirically tested. Considering the importance of QMP's modulations of dopamine and octopamine to worker activities and colony harmony, our lack of understanding into how neonicotinoids affect these biogenic amines in honey bees represents a crucial gap in our knowledge.

Here, I test the hypothesis that neonicotinoids disrupt worker responses to QMP by interfering with octopamine and dopamine signalling, both of which are A. implicated as messengers of QMP's effects on workers and B. likely to be affected by neonicotinoid exposure. This is split into two sub-hypotheses (H1 and H2):

H1) Neonicotinoids disrupt worker behavioural and physiological responses to QMP

H2) disruptions to dopamine and octopamine signalling are the mechanisms underlyingH1.

To test H1, worker responses to QMP were measured in bees undergoing a range of chronic imidacloprid exposure regimes. Three major endpoints were selected to assess worker responses to QMP, testing QMP's primer (long-term physiological and behavioural change) and releaser (short-term behavioural change) functions. The releaser function of QMP was tested by assessing the degree of attraction of workers to

the queen, or their 'retinue response' behaviour. Primer functions were tested by examining A. the locomotor activity of workers as a proxy for the behavioural transition to foraging, and B. ovary activity levels as a measure of QMP's induction of sterility.

To test H2, dopamine and octopamine levels were measured in workers following chronic imidacloprid regimes in combination with chronic QMP exposure. These biogenic amines are thought to act as both neurohormones (i.e. In the brain) and circulatory hormones (i.e. Interacting directly with peripheral tissues) to exert effects on workers (Chapter three). To understand their underlying role in coordinating worker responses to QMP and imidacloprid, octopamine and dopamine levels were measured in both the brain and hemolymph of workers following imidacloprid and QMP treatments.

To my knowledge, this is the first study investigating the effects of neonicotinoids on adult worker responses to QMP. In particular, the mutual manipulation of aminergic signalling pathways by both QMP and neonicotinoids have not previously been considered alongside one another. This is therefore a novel hypothesis, and this chapter sheds light on additional underlying mechanisms of the known negative effects of neonicotinoids on honey bee colony health and productivity.

6.2 Methods

6.2.1 Microcolony set-up and experiment overview

A. mellifera laboratory microcolonies were setup from source hives and maintained in cages as described in Section 2.1.1. Microcolonies each contained 100 workers and were exposed to imidacloprid at varying concentrations. To observe the interactive effects of imidacloprid with QMP, each imidacloprid treatment was tested both with and without continual QMP exposure. Two discrete experiments were conducted to address H1) that neonicotinoids disrupt worker behavioural and physiological responses to QMP and H2) that disruptions to dopamine and octopamine signalling are the mechanisms underlying H1. An overview of each experiment is detailed bellow.

Experiment one – behavioural and reproductive responses to QMP (H1)

The aim of experiment one was to assess the impacts of imidacloprid on QMP's modulation of worker behaviour and reproduction. Attraction of workers undergoing different imidacloprid and QMP treatment regimes to QMP was tested on day four in the retinue response assay, and locomotor activity levels were tested on day five as a proxy for foraging behaviour. In each assay, 10 randomly selected bees were tested from each cage. Bees used in behavioural assays were discarded following testing and not returned to their cage. On day 10, the ovaries of all remaining bees in each cage were dissected and imaged for later analysis of ovary activity levels as described previously (Section 2.2). Each experimental cage had three independent replicates (i.e. for behavioural assays n=30 bees in total). Within each replicate, all treatment cages were set up within two days of each other. All three replicates of experiment one were carried out between July – August 2021 under my supervision by two Biodiversity and Conservation MSc students, Jessica Bouwer and Emily Ross.

Experiment two – biochemical responses to QMP (H2)

Experiment two aimed to investigate the underlying mechanisms of experiment one by measuring honey bee internal dopamine and octopamine levels following each imidacloprid and QMP exposure regime. Cages were maintained as in experiment one. Whole brains and hemolymph samples were collected from 10 bees per cage on day 10 for future analysis of biogenic amine content using high-performance liquid chromatography with fluorescent detection (HPLC-FLD) as described (Section 2.3). Ovary activity levels and locomotor activity were measured in a subset of bees as a means of validating the consistency of QMP and neonicotinoid treatments between experiments one and two (see Appendix I, Figure I.1 for ovary activity in experiment two). Three replicates of each cage were carried out in September 2021 within three days of one another.

6.2.2 Imidacloprid and QMP treatments

Cages underwent chronic pesticide exposure regimes by spiking their water solutions with imidacloprid, while control cages received just water. The chronic lethal imidacloprid dose for honey bees (LC50), or the expected concentration associated with death of 50% of bees feeding on it, is estimated as 1,760 parts per billion (ppb) (Cresswell, 2011). A preliminary assessment of survival rates at a range of imidacloprid concentrations was carried out to verify pesticide efficacy and determine sublethal dosages within the laboratory setup. Cages of 100 workers were setup as described above and mortality rates recorded at 0, 100, 250, 500 and 1,000 ppb imidacloprid exposure in June-July 2019. In June-July 2021, survival was monitored at 0, 1 and 5 ppb imidacloprid. Determination of sub-lethal dosages was carried out in the absence of QMP, as is comparable with previous assessments of pesticide lethality. Imidacloprid was found to pose little risk to mortality under chronic exposure regimes of 100 ppb and bellow in our laboratory setup (see Appendix J, Figure J.1). Doses of 1 ppb and 5 ppb were selected for assessment, as these are within the realistic range in which honey bees are likely to come into contact with imidacloprid in the field (estimated to be between 0.7-10 ppb; Cresswell, 2011).

Cages therefore received one of six treatments: 0 ppb (control), 1 ppb or 5 ppb chronic imidacloprid exposure, each tested in both the presence (QMP+) or absence (QMP-) of continual QMP exposure. QMP treatments were administered as described previously (Section 2.1.2). Imidacloprid solutions were supplied ad libitum and replaced every three days. Mean solution intake per bee per day was recorded for each cage by weighing the solution tube daily.

6.2.3 Retinue response assay

Responses to QMP were tested on four day old workers, as this is the age at which workers are typically of nursing age and show the greatest attraction to QMP (Vergoz *et al.*, 2009). QMP attraction, or 'retinue response', was assessed using a Petri dish divided


Figure 6.1 Diagram of the retinue response assay arena. A petri dish divided into four concentric sectors was used as the behavioural arena for the retinue response assay following the methodology used by Vergoz *et al.*, (2009). Sectors were numbered to calculate area occupancy scores.

into four concentric sectors as a behavioural arena (Figure 6.1). Petri dish lids were modified to consist of a mesh layer to allow odours to dissipate and not fill the whole dish homogenously. The responses of bees to a 10 μ L droplet of 0.01 QE/ μ L synthetic QMP vs a 10 μ L ethanol droplet placed in the centre of the arena were compared. Bee responses were tested to ethanol droplets as a comparative control to account for individual differences in locomotor activity and inquisitiveness between bees. The order of droplet presentation was randomised for each bee and petri dishes were cleaned with 70% ethanol and aired between every trial. Trials were carried out under red light, as bees cannot see red light thus mimicking the darkness of the hive (Kimura *et al.*, 2014). Trials were recorded two at a time from a birds-eye view of the arena using a GoPro camera and tripod for later analysis of the attraction of each bee to the central droplet (Figure 6.2). Videos were three minutes in duration following an initial one minute acclimatization period which began immediately after the bee was placed in the arena.

Attraction to each test droplet was assessed from video footage using the methodology described by (Vergoz *et al.*, 2009). The sector in which the bee was located was recorded every 10 seconds. Sectors were numbered from one to four, with one being the central sector increasing to four in the outermost sector (Figure 6.1). A bee was classified as



Figure 6.2 Photo of the retinue response assay filming set-up. A GoPro was positioned in a birds eye view above two behavioural arenas to enable simultaneous recording of two trials at once. Though the setup is pictured in normal lighting, assays were performed under red light. Photo was taken by Emily Ross and used with permission.

being in a sector when over half of its body occupied it, or when the body was positioned equally between two sectors, the sector occupied by the head was recorded. The area occupancy score was calculated by multiplying each sector by the number of times the bee was recorded in that sector, then taking the sum for each sector. For example, if a bee occupied sector one once, sector two five times, sector three five times and sector four seven times, the area occupancy score would be:

 $(1 \times 1) + (2 \times 5) + (3 \times 5) + (4 \times 7) = 54$

A bee spending most of its time in the central sectors will have a lower area occupancy score than a bee spending more time in the outermost sectors. As sector one contains the central droplet, a lower area occupancy score corresponds to a greater attraction to the test droplet.



Figure 6.3 Photo of the locomotor activity assay filming set-up. A GoPro was positioned in a birds eye view above two behavioural arenas to enable simultaneous recording of two trials at once. Trials were recorded under red light, as pictured.

6.2.4 Locomotor activity assay

The locomotion assay was used to assess activity levels across treatment groups, as high activity levels are associated with foraging performance (Klein *et al.*, 2019). Locomotor activity may therefore be seen as a very rudimentary proxy for the behavioural transition from nurse to forager, similarly to task allocation and personality in ants (Dr Victoria Norman, personal communication). The locomotor assay used was modified from Williams *et al.*, (2020). The behavioural arena was a rectangular petri dish (12cm x 8 cm x 2 cm) divided into 6 equal square sections (Figure 6.3). The number of times a bee crossed a line from one section into another was recorded. A cross was only recorded if the bee's entire body crossed from one section into an adjacent section. Crosses made diagonally between sections were recorded as one line cross. As above, trials were carried out under red light and recorded for later analysis of video footage, and the trial duration was 3-minutes following an initial 1-minute acclimatization period. Bees were tested in blocks of ten at a time randomised across treatment groups. 'Assay block' was

recorded for each bee and included as a random effect in data analysis (described in section 6.2.5).

6.2.5 Data analysis

All statistical analyses were carried out using R. Details of packages used for analysis are found in section 2.4. For all analyses, imidacloprid dose was considered an ordinal variable, as the number of doses tested was insufficient to be considered a continual variable.

To assess whether imidacloprid and QMP treatments caused confounding effects on food and water consumption, analyses were performed on the effects of QMP and imidacloprid on food and water intake. Linear mixed effects models (LMMs) were fitted with mean food or water consumed per bee per day as the response variables and QMP, imidacloprid dose and their interaction were as fixed effects. To account for repeated measures, day was included as a random effect and nested within replicate. Visual inspection of residual plots showed no obvious deviations from normality or homoscedasticity. Statistical significance of fixed effects was obtained by comparing likelihood ratios of the maximal model to the model without the fixed effect of interest. *Post-hoc* testing was carried out on the maximal model by computing estimated marginal means. P-values were Tukey-adjusted to control for multiple testing.

The effects of QMP and imidacloprid dose on survival were tested using Cox proportional hazards models (CPH) with mixed effects. The fixed effects were QMP, imidacloprid and their interaction, with replicate included a random effect. The assumption of proportional hazards was verified for the model as a whole by correlating scaled Schoenfield residuals with time to test for independence. This model was used to predict Hazard Ratios (HR) and confidence intervals for different treatment combinations. HRs are presented as HR (± 95% confidence interval). P-values for all pairwise comparisons between treatments were obtained from the maximal model by computing estimated marginal means. Multiple testing was accounted for with Tukey-adjustment of P-values.

Area occupancy scores of each bee in the presence of the control droplet vs QMP droplet were compared using paired T-tests. Paired T-tests were carried out for every QMP/imidacloprid treatment combination to assess in which groups preferences for the QMP test droplet were seen. Effect sizes (d) were calculated by dividing the mean difference by the standard deviation of the difference of paired samples (Cohen's-d formula). Normality of differences between paired samples was confirmed through visual assessment of histogram plots.

Locomotor activity was assessed by LMMs on the number of line crosses with QMP presence, imidacloprid dose and their interaction as fixed effects. Replicate and assay block were included as random effects, with by-replicate and by-assay block random slopes for the effect of all fixed effects. Visual inspection of residual plots showed no obvious deviations from normality or homoscedasticity.

P-values for individual fixed effects and their interaction were obtained by comparing the likelihood ratio of the maximal model to that of the model without the fixed effect of interest (Bates *et al.* 2015). *Post-hoc* pairwise testing was carried out on the maximal model using estimated marginal means. P-values were Tukey-adjusted to control for multiple testing.

Ovary scores in each treatment group were analysed by fitting cumulative link mixed models (CLMMs) with Laplace approximation. QMP presence, imidacloprid dose and their interaction were included as fixed effects, and replicate was included as a random effect. Statistical significance of fixed effects was assessed by comparing the likelihood ratio of the maximal model to that of the model without the fixed effect of interest.

Analysis of brain and hemolymph octopamine and dopamine levels was carried out by fitting GLMMS. QMP, imidacloprid dose and their interaction were included as fixed effects and replicate and HPLC batch were included as random effects with random intercepts. LMMs were initially fitted and residual distributions were tested by visual examination of qqplots and histograms. In all cases these were not normally distributed so GLMMs were instead fitted. Histograms of the raw data were used to decide model

family and link function. For brain octopamine and dopamine, GLMMs with a gamma distribution and an inverse link function were fitted. For hemolymph octopamine and dopamine a log link function was used. Statistical significance of fixed effects and their interaction was tested by comparing the likelihood ratio of the maximal model to that of the model without the fixed effect of interest (Bates *et al.* 2015). If fixed effects were statistically significant, *post-hoc* pairwise testing was carried out on the maximal model using estimated marginal means. P-values were Tukey-adjusted to control for multiple testing.

6.3 Results

This study aimed to test the hypothesis that exposure to neonicotinoid pesticides disrupts honey bee worker responses to QMP (H1). Through the assessment of a series of endpoints testing established behavioural and physiological effects of QMP on workers, interactive effects between imidacloprid and QMP are studied here for the first time. The hypothesis that disruptions to aminergic signalling underly these effects was also tested by quantifying dopamine and octopamine levels in key worker tissues (H2).

6.3.1 Effects of imidacloprid and QMP treatments on food and water intake

Food intake was measured daily for each cage to assess whether there were any confounding effects of QMP and imidacloprid treatments on nutritional status that could impact experimental results. Water intake (or imidacloprid solution in pesticide treatments) was also recorded daily for each cage, to verify the administration of treatment solutions and estimate imidacloprid dosages received. As experiment one (assessment of ovary activity and behaviour – H1) and experiment two (assessment of internal biogenic amine levels – H2) were conducted at separate times, these are considered two discrete experiments and food and water consumption data from each are reported independently of one another.

in experiment one, food intake did not differ between treatment groups (Figure 6.4). Comparisons of LMMs with QMP, imidacloprid and their interaction as fixed effects



Figure 6.4 Food and water intake of workers treated with QMP and imidacloprid. Workers were housed with QMP (QMP+, orange) or solvent control (QMP-, green) and treated with 0, 1 or 5 ppb imidacloprid in **A.** Experiment 1 and **B.** Experiment 2. Food and water consumption was recorded daily for each treatment cage (n = 10 days), and each treatment had three independent replicates (n = 30 total). Pairwise *post-hoc* comparisons were carried out where LMM model comparisons revealed statistical significance of imidacloprid or QMP (denoted by "*"). Due to the large number of pairwise comparisons, these are not presented on the figure and can instead be found in Appendix K, Table K.1. "n.s" represents statistical non-significance.

showed that neither QMP, Imidacloprid, nor their interaction had statistically significant effects on food consumption (LMMs: QMP: $\chi 2 = 3.02$, df = 3, 161, p = 0.39; Imidacloprid: $\chi 2 = 0.88$, df = 4, 160, p = 0.93; Interaction: $\chi 2 = 0.17$, df = 2, 162, p = 0.92), indicating that nutritional intake does not contribute to any treatment-level differences in ovary activity, locomotor activity or retinue response results. In experiment two, in which biogenic amines were quantified, food intake was slightly lowered by imidacloprid treatment (Figure 6.4; LMMs: $\chi 2 = 10.26$, df = 4, 174, p < 0.05). Additionally, bees consumed more food when exposed to QMP (LMMs: $\chi 2 = 44.75$, df = 3, 175, p < 0.001), though there was no evidence for an interaction effect of imidacloprid and QMP (LMMs:

101

 χ 2 = 2.18, df = 2, 176, p = 0.36). A full table of all pairwise comparisons revealed by *posthoc* tests is displayed in Appendix K, Table K.1.

Across both experiment one and experiment two, bees maintained in the presence of QMP consumed significantly less water than those maintained without QMP (Figure 6.4; LMMs; Experiment one: $\chi 2 = 23.92$, df = 3, 158, p < 0.001; Experiment two: $\chi 2 = 50.18$, df = 3, 172, p < 0.001). It should therefore be noted that as imidacloprid was administered in water solutions, pesticide dosages received will be slightly elevated in QMP- treatment cages. See Appendix K, Table K.1 for *post-hoc* pairwise comparisons between treatments calculated using estimated marginal means. Imidacloprid treatment had no effect on water consumption in experiment one and two (LMMs; Experiment one: $\chi 2 = 1.86$, df = 4, 157, p = 0.76; Experiment two: $\chi 2 = 2.03$, df = 4, 171, p = 0.73). There was also no evidence to support an interaction effect between QMP and imidacloprid on water intake in either experiment (LMMs; Experiment one: $\chi 2 = 1.43$, df = 2, 159, p = 0.49; Experiment two: $\chi 2 = 0.24$, df = 2, 173, p = 0.89), indicating that there were neither repulsive nor addictive effects of pesticide solutions at the concentrations given.

6.3.2 Ovary activity is repressed by synthetic QMP regardless of imidacloprid treatment

To test whether imidacloprid treatment disrupts QMP's ability to inhibit worker reproduction, ovary activity levels were assessed in workers housed with or without QMP (QMP+/ QMP-) following different chronic imidacloprid treatment regimes. In this laboratory setting, the efficacy of synthetic QMP is demonstrated by its repression of worker ovary activity in imidacloprid control groups (Figure 6.5). When comparing 0 ppb imidacloprid controls, 24.22% of 10-day old workers housed without QMP had ovaries containing mature oocytes (ovary score of 3 – highest degree of ovary activity; Figure 2.2), while this number was reduced to 1.74% in the presence of QMP (Figure 6.5).

To test whether imidacloprid treatment affected QMP's regulation of worker reproduction (H1), CLMMs were fitted with the maximal model containing imidacloprid,



Figure 6.5 Ovary activity is repressed by QMP regardless of imidacloprid dosage. Ovary activity levels of 10-day old worker honey bees maintained in the presence of QMP (**QMP+**, shades of orange) or solvent control (**QMP-**, shades of green) following 0 ppb (control), 1 ppb and 5 ppb imidacloprid (IMI) treatment. Ovary activity is presented as the percentage of bees within each treatment group with each ovary score. Ovary scores range from 0-3 (see Figure 2.2), where a higher score (darker shade) relates to a higher degree of ovary development. All treatment cages began with 100 bees, and all remaining bees on day 10 (after deaths and removal for behavioural assays) were dissected. N values are displayed at the base of each bar. Data shown is from experiment 1 and is pooled over three replicates of each treatment. Exceptions are QMP+ 5 ppb, where all bees in replicate three died prior to ovary dissection, and QMP- 0 ppb, where a technological failure caused the loss of all ovaries imaged from replicate one.

QMP and their interaction as fixed effects, and reduced models containing all but the fixed effects of interest. As discussed above, model comparisons showed QMP's repression of ovary activity to be statistically significant (CLMM: $\chi 2 = 73.91$, d.f = 3, 885, p < 0.0001), which is consistent with previous studies using this laboratory system (Duncan *et al.*, 2016). Within this setting, QMP's effects on ovary development were unaltered by pesticide treatment; Imidacloprid had no effect on ovary activity levels at any dose either on its own (CLMM: $\chi 2 = 1.35$, d.f = 4, 884, p = 0.85) or through interactions with QMP (CLMM: $\chi 2 = 1.094913$, d.f = 2, 886, p = 0.58), indicating that the primer effects of QMP on worker ovary development are unaffected by neonicotinoid exposure.



Figure 6.6 Effects of QMP and imidacloprid treatments on biogenic amine levels. Levels of dopamine (DA) and octopamine (OA) in tissues of 10-day old laboratory-raised worker honeybees following different doses of IMI treatment, where workers were maintained in experimental cages of 100 bees and received 0.1 QE of QMP (+QMP) or solvent control (-QMP) every 24 hours. HPLC-FLD was used to quantify levels of DA and OA from samples extracted from 10 randomly selected bees per treatment for three experimental cage replicates (n=30 per treatment total). Statistical significance between treatments in *post-hoc* tests at the level of p < 0.01 is highlighted by (**). Full table of *post-hoc* pairwise tests is found in Appendix K, Table K.2. **(A)** Brain dopamine levels were lower in +QMP bees relative to – QMP. This was statistically significant only for bees receiving 5ppb IMI. **(B)** Brain OA levels were lower +QMP, but not significantly so, and are unchanged by IMI treatment. **(C, D)** DA and OA were detected in worker hemolymph but were unaffected by either IMI treatment or QMP exposure.

6.3.3 Imidacloprid interacts with QMP to lower brain dopamine levels

To test whether imidacloprid interferes with QMP's modulation of biogenic amines (H2), dopamine and octopamine were quantified in the brains and hemolymph of workers following different QMP and imidacloprid treatment regimes (Figure 6.6). For each biogenic amine and each tissue, GLMMs were fitted with QMP, imidacloprid and their interaction as fixed effects. As is consistent with in-hive studies, synthetic QMP was associated with a statistically significant reduction of both dopamine and octopamine levels in 10-day old worker brains (GLMMs: Brain dopamine: $\chi 2 = 15.78$, d.f = 3, 138, p <

104

0.01; Brain octopamine: $\chi 2 = 1.21$, d.f = 3, 137, p < 0.05). Pairwise *post-hoc* testing using estimated marginal means revealed QMP's reduction of brain dopamine levels had the largest effect size and was statistically significant only in combination with 5 ppb imidacloprid treatment (Z = -3.84, p <0.01; for full table of *post-hoc* comparisons see Appendix K, Table K.2). Additionally, GLMM model comparisons showed that although imidacloprid alone had no measurable effect on brain dopamine levels (GLMM: $\chi 2 = 8.26$, d.f = 4, 137, p = 0.83), there was a statistically significant interaction effect of imidacloprid on QMP (GLMM: $\chi 2 = 7.51$, d.f = 2, 139, p < 0.05). Put simply, 5 ppb imidacloprid treatment enhanced the depressive effects of QMP on brain dopamine levels in workers, a phenomenon which has not previously been demonstrated.

In contrast, there was no evidence for any effects of imidacloprid treatment on brain octopamine levels (GLMM: $\chi 2 = 2.94$, d.f = 4, 136, p = 0.57), nor for interaction effects with QMP (GLMM: $\chi 2 = 1.48$, d.f = 2, 138, p = 0.48). In the hemolymph, QMP treatment was associated with a slight reduction of octopamine levels (as has been shown in the hive – Chapter three), although this was not statistically significant (GLMM: $\chi 2 = 3.24$, d.f = 3, 175, p = 0.35). Imidacloprid also had no measurable effect on hemolymph octopamine levels, either alone (GLMM: $\chi 2 = 3.13$, d.f = 4, 174, p = 0.54) or through interactions with QMP (GLMM: $\chi 2 = 2.91$, d.f = 2, 176, p = 0.23). Hemolymph dopamine levels, which have not previously been found to be altered by QMP in 10-day old laboratory workers (Chapter three), showed a similar trend to brain dopamine levels whereby QMP's depression was exacerbated with increasing imidacloprid dose. However, no significant effects on hemolymph dopamine levels were caused by QMP, imidacloprid, or their interaction effect (QMP: $\chi 2 = 2.07$, d.f = 3, 175, p = 0.56; Imidacloprid: $\chi 2 = 1.68$, d.f = 4, 174, p = 0.79; Interaction: $\chi 2 = 1.24$, d.f = 2, 176, p = 0.54).



Figure 6.7 Locomotor activity of QMP and imidacloprid treated workers. Locomotor activity of five-day old bees undergoing different imidacloprid treatments in the presence (**QMP+** orange boxplots) and absence (**QMP-** green boxplots) of QMP in **A.** Experiment one and **B.** Experiment two. N = 30 for each imidacloprid treatment (three pooled replicates where 10 bees were assayed each replicate). Locomotor activity is measured by the number of line crosses made by a bee over the three-minute assay duration. Overall statistical non-significance from LMM model comparisons is denoted by "n.s."

6.3.4 QMP and imidacloprid did not affect the locomotor activity of five-day old workers

Locomotor activity was studied as a proxy for foraging behaviour. There was no evidence for any effects of QMP, imidacloprid, or their interaction on locomotor activity in fiveday old workers (Figure 6.7). Comparisons of LMMs showed that QMP and imidacloprid had no statistically significant impacts on locomotor activity in experiment one (LMMs; QMP: $\chi 2 = 4.02$, d.f = 3, 176, p = 0.26; IMI: $\chi 2 = 3.83$, d.f = 4, 175, p = 0.43; Interaction: $\chi 2 = 1.70$, d.f = 2, 177, p = 0.43). The locomotor activity assay was repeated in experiment two to verify consistent behavioural responses to imidacloprid and QMP treatments across experiments for validation of biogenic amine data. Consistently with experiment one, no effects on locomotor activity were found in response to QMP treatment, imidacloprid treatment or their interaction effects (LMMs; QMP: $\chi 2 = 0.56$, d.f = 3, 176, p = 0.91; IMI: $\chi 2 = 1.04$, d.f = 4, 175, p = 0.90; Interaction: $\chi 2 = 0.29$, d.f = 2, 177, p = 0.86; figure 7). However, there are likely limitations to this assay's efficacy as a proxy for forager behaviour, as discussed in section 6.4.



Figure 6.8 Attraction of workers to QMP Responses of imidacloprid (IMI)-treated four-day old worker honeybees to a test droplet of QMP (darker shaded boxes) vs a control droplet of ethanol (lighter shaded boxes) during a three-minute assay period. Workers were housed in cages with continual exposure to **A.** solvent control (**QMP-**, shades of green) or **B.** QMP (**QMP+**, shades of orange) prior to testing. Ten randomly selected bees were tested in each treatment group for three cage replicates (n=30 per treatment total). Relative attraction to test droplets is measured by area occupied score, where a lower score relates to a greater preference. Statistical significance between QMP vs control droplet areas at the levels of p<0.01** and 0.0001**** is highlighted by the corresponding numbers of asterisks. Statistical non-significance is denoted by "n.s". Bees that have had no prior exposure to QMP (**QMP-**) show no preference to QMP test droplets over control test droplets, and this is unaltered by IMI treatment. Bees that had seen continual exposure to QMP since adult emergence (**QMP+**) showed significantly increased attraction to QMP test droplets relative to control test droplets relative to control test droplets.

6.3.5 Imidacloprid increases worker attraction to QMP

The attraction of workers to a test droplet of QMP vs ethanol controls was compared in four-day old bees that had been housed with or without QMP following different imidacloprid exposure regimes (Figure 6.8). "Area occupancy score" is the metric used to measure attraction to the test droplet, with a lower score relating to a higher degree of attraction. For workers that had been exposed to QMP since adult emergence (QMP+), increasing imidacloprid dosages were associated with an increase in the strength of attraction to QMP. In QMP+ 0 ppb imidacloprid controls, the test droplet received had a moderate effect on area occupancy scores, with scores being lower for QMP droplets than ethanol control droplets (Paired T-test: T = 2.77, df = 29, p < 0.001, d = 0.51, effect size moderate). QMP+ bees treated with 1 ppb imidacloprid showed a stronger response to QMP, as demonstrated by the larger effect size of the test droplet on area occupancy scores (Paired T-test: T = 4.77, df = 29, p<0.0001, d=0.87, effect size large). Interestingly, the highest degree of QMP attraction was seen in QMP+ bees receiving 5 ppb imidacloprid, where the effect size of test droplet on area occupancy was greater than in any other treatment group (Paired T-test: T = 6.14, df = 29, p<0.0001, d=1.12, effect size large). In short, the attraction of QMP+ workers to QMP became stronger with increasing imidacloprid exposure.

Consistent with published work (Vergoz *et al.*, 2009), workers that had seen no prior exposure to QMP (QMP-) displayed no evidence of attraction to QMP. Paired T-tests showed there to be no statistically significant difference in area occupancy scores between QMP droplets and control droplets in –QMP 0 ppb imidacloprid controls (Paired T-test: T = 0.29, df = 29, p=0.778, d=-0.05, effect size negligible). Unlike those that had seen prior exposure to QMP, imidacloprid treatment did not alter the retinue responsiveness of QMP- bees, with 1 ppb and 5 ppb treatment groups also showing no preference for QMP droplets over control droplets (Paired T-test: 1 ppb and 5 ppb respectively: T = -0.60, 0.03, df = 29,29, p=0.55, 0.98, d=-0.11, 0.01, effect size negligible in both cases).



Figure 6.9 Effects of QMP and imidacloprid on survival Survival probabilities of cages of 100 workers maintained in the absence or presence of QMP receiving imidacloprid (IMI) solutions at concentrations of 0, 1 and 5 pbb. Increasing imidacloprid concentrations are denoted by darkening shades of green (QMP-) and orange (QMP+). Survival probabilities are shown for **A.** Experient 1 (July-August 2021) and **B.** Experiment 2 (September 2021). Data shown constitutes three independent replicates of each discrete experiment. One cage was removed from the dataset due to rapid unexpected death of all bees (Experiment 1, QMP+ 5PPB, replicate 3). Censoring events, for instance, the removal of bees from the cage for behavioural assays, are denoted by crosses. The Y axis has been truncated to begin at a survival probability of 0.5. For statistical significance of *post-hoc* pairwise comparisons of survival between different treatment groups, see Appendix K, Table K.3

6.3.6 QMP exposure may impact the lethality of imidacloprid

Analysis of survival rates between treatment groups was carried out using CPH models with mixed effects to assess whether the interactive effects of QMP and imidacloprid

affected mortality. As experiments one (assessment of ovary activity and behaviour – H1) and two (assessment of internal biogenic amine levels – H2) are considered two discrete experiments, survival data were analysed separately for each.

In cages maintained in the absence of QMP (as laboratory testing of imidacloprid lethality has previously been conducted), treatment with 1 and 5 ppb imidacloprid had no negative effects on survival (Figure 6.9) in both experiment one (1 ppb: HR = 0.47 ± 0.35 SE, z = -2.17, p = 0.06; 5 ppb: HR = 0.69 ± 0.31 SE, z = -1.17, p = 0.24) and experiment two (1 ppb: HR = 1.27 ± 0.24 SE, Z = 1.00, p = 0.32; 5 ppb: HR = 1.14 ± 0.24 SE; Z = 0.56; p = 0.58).

As a central facet of this study involved the determination of synergistic interactions between QMP and imidacloprid, the effect of QMP alone on survival in 0 ppb imidacloprid controls was analysed to assess whether QMP treatments caused inherent differences in survival. In experiment one, QMP treatment was not associated with any differences in survival in imidacloprid controls (HR = 1.25 ± 0.27 SE, z = 0.82, p = 0.41). In experiment two, exposure to QMP significantly increased the risk of mortality (HR = 3.34 ± 0.21 SE, z = 5.87, p < 0.001), although this trend was largely driven by one cage in which death rates were unusually high at 50%. It is suspected that this may have resulted from external factors, such as incidence of disease within the cage, as QMP treatment has generally not been associated with a reduction in survival throughout other chapters within this thesis.

Examining the combined effects of QMP and imidacloprid on survival rates revealed that in experiment one, exposure to QMP increased the lethality of imidacloprid. Pairwise *post-hoc* testing using estimated marginal means showed that bees receiving the same imidacloprid dose saw higher mortality rates when these were administered in combination with QMP. QMP exposure reduced the survival probabilities of bees at 1 ppb (z = -4.54, p <0.0001) and 5 ppb imidacloprid treatment (z = -3.60, p <0.01). The same trend was observed in experiment two, though only at the 5 ppb dose (z = -4.09, <0.001), and the high mortality rates of QMP+ imidacloprid control limit the extent to which synergistic effects of QMP and imidacloprid on mortality can be inferred For a complete table listing all *post-hoc* pairwise comparisons, see Appendix K, Table K.3.

6.4 Discussion

Produced by the queen, QMP is a major regulator of honey bee biology and in-hive dynamics (Hoover et al., 2003), promoting queen-worker cooperation and coordinating division of labour in the hive (Pankiw et al., 1998). Biogenic amines have been implicated as key messengers mediating the varied downstream behavioural and physiological effects of QMP (Beggs et al., 2007; Harris & Woodring, 1995; Sasaki & Nagao, 2001; Chapter three; Chapter four). There is also evidence, from other species, that neonicotinoids alter levels of the biogenic amines octopamine and dopamine (Fuenzalida-Uribe et al., 2013; Janner et al., 2021; Pyakurel et al., 2018), raising the possibility that these pesticides may interfere with worker interpretation of QMP's chemical communication. This chapter aimed to test two hypotheses: H1) that neonicotinoids interfere with established worker responses to QMP, and H2) that perturbations to aminergic signalling (namely octopamine and dopamine) play an underlying mechanistic role in H1. By investigating a series of well-documented worker responses to QMP, this chapter shows evidence for interactive effects between imidacloprid and QMP on worker behaviour, with speculative roles of biogenic amines underlying these behavioural alterations.

A notable finding was the increased attraction of workers to QMP following chronic imidacloprid treatment as measured by the retinue response assay. The attraction of young workers to QMP ensures workers feed and groom the queen and distribute her pheromones throughout the hive (Slessor *et al.*, 1988; Slessor *et al.*, 2005), and frequent contact with QMP slows the transition from nursing to foraging (Pankiw *et al.*, 1998). Attraction to QMP is usually highest in younger workers, and the decline in attraction with age is associated with the transition to foraging (Pankiw *et al.*, 1998; Vergoz *et al.*, 2009). There is also a high degree of individual diversity in QMP responsiveness (Kocher *et al.*, 2010; Walton & Toth, 2016), and these individual and temporal variations in

response thresholds facilitate effective division of labour in the hive (Walton *et al.*, 2018). Therefore, the correct balance of worker attraction to QMP is integral for ensuring labour needs of the colony are met, through maintenance of appropriate numbers of nurses and foragers.

To my knowledge, the effect of adult worker exposure to neonicotinoids on QMP attraction has not previously been investigated. Larval exposure to a different class of pesticides, the insect growth disruptors (IGDs) pyriproxyfen and methoxyfenozide, has been shown to negatively affect worker attraction to QMP in adulthood (Litsey *et al.*, 2021), in contrast with the increase in attraction seen with imidacloprid. These opposing phenotypes may be partially explained by the methodology used to measure QMP attraction. In both this study and Litsey *et al.*, (2021), attraction was assessed through observations of worker responses to a strip of synthetic QMP. However, Litsey *et al.* (2021) made no comparative observations of workers to a control strip, meaning their reported loss of attraction could have arisen from reduced locomotor capabilities, lower energy reserves or diminished inquisitiveness to novel stimuli in pesticide-treated bees. Further, concentrations of pesticides used in the study were particularly high; larvae were treated with methoxyfenozide at concentrations 10-fold higher than have been reported in pollen stores of hives from the field (Rennich *et al.*, 2014).

This being said, it may be expected that the effects on QMP attraction differ between classes of pesticides due to their range in target biochemical pathways. IGDs target pathways associated with insect growth and development (Pener & Dhadialla., 2012), with pyriproxyfen in particular being analogue of Juvenile Hormone (JH). JH functions as a gonadotropin in solitary and primitively eusocial insects, but a rewiring of this neuroendocrine network in worker honey bees has resulted in the loss of its reproductive function (Rodrigues & Flatt, 2016). JH instead plays a role in coordinating temporal polyethism (Robinson & Vargo, 1997), with a rise in JH associated with the transition to foraging (Schulz *et al.*, 2002) and reduced attraction to QMP (Mcquillan *et al.*, 2014). JH has been shown to reduce attraction of honey bee workers to QMP by lowering transcript levels of the octopamine receptor *AmOA1* in the antennae

(Mcquillan *et al.*, 2014). Antennal expression levels of *AmOA1*, as well as the D2-like dopamine receptor *AmDop3*, positively correlate with QMP attraction, acting antagonistically with the D1-like receptor AmDop1 (Vergoz *et al.*, 2009). Therefore, a reduction in worker QMP attraction caused by the JH analogue pyriproxyfen would be expected if this IGD has the same regulatory effects on antennal biogenic amine receptor expression as JH.

The potential mechanism behind the effects of imidacloprid on retinue response behaviour is more difficult to untangle. Here, it was hypothesised that neonicotinoids would disrupt biogenic amine levels in the brain, as has been demonstrated in D. melanogaster (Janner et al., 2021). Dopamine and octopamine both regulate JH degradation in Drosophila species (Gruntenko et al., 2005, 2007), though it is unclear whether this is true for honey bees also. Assuming biogenic amines also mediate JH levels in honey bees, neonicotinoids may reduce QMP attraction by decreasing JH titres indirectly through modulation of biogenic amine levels. However, further study is required on the relationship between dopamine and octopamine and JH titres in honey bees in order to test the occurrence and directionality of JH regulation. This could be achieved by manipulating internal biogenic amine levels in honey bees (i.e. using the methodologies discussed in Chapter four) followed by determination of JH titres using mass spectrometry (GC-MS) or HPLC-FLD with the addition of fluorescent JH labelling reagents (Rivera-Perez et al., 2012). Assessment of JH titres in honey bees following imidacloprid treatment would also clarify whether the increased attraction to QMP associated with imidacloprid exposure is underpinned by decreases in JH, with further measurements of antennal expression levels of the octopamine receptor AmOA1 (which has been implicated in JH's mediations of QMP attraction as mentioned previously; Mcquillan et al., 2014).

The effects of neonicotinoids on JH titres have not been investigated in adult honey bees. However, one study has been carried out on honey bees at the larval stage, finding that exposure to thiamethoxam (another neonicotinoid pesticide) was associated with an increase in larval JH titres (Li *et al.*, 2021). However, the effects of neonicotinoids on JH titres are likely to differ between developmental stages, as ontogenetic differences in the effects of dopamine on JH have been established in a range of insect species. For example, in *D. melanogaster* females, dopamine has opposing effects on JH levels depending on developmental stage, causing an increase in young females and a decrease in sexually mature females (Gruntenko *et al.*, 2005, 2012; Rauschenbach *et al.*, 2011). If neonicotinoids alter JH levels via disruptions to dopamine's neurohormonal signalling in honey bees, they could thus plausibly elicit opposing effects on honey bee JH titres at larval and adult stages.

Support for the hypothesis that imidacloprid alters brain biogenic amine levels in honey bees (H2) was mixed. QMP's depression of octopamine levels was unaffected by imidacloprid, though brain dopamine levels were altered by imidacloprid exposure through an intriguing interactivity with QMP. In *D. melanogaster*, long-term imidacloprid treatment reduces brain dopamine levels, possibly via a blockade of nAChR-mediated dopamine release following persistent activation (Janner et al., 2021). This was partially true for honey bees, but only via additive interactions with QMP, as chronic exposure to 5 ppb imidacloprid exacerbated QMP's depression of brain dopamine levels. The depressive effects of QMP on dopamine have been discussed in previous chapters (Chapter three, Chapter four). Given the increased attraction to QMP associated with imidacloprid treatment, it is tempting to conclude that heightened interaction of imidacloprid-treated workers with QMP exacerbated QMP's depression of dopamine. In contrast with the original predictions of H2 (i.e. that imidacloprid directly modulates biogenic amines), this would instead imply that imidacloprid alters dopamine indirectly via this increased attraction to QMP. Workers that have not been exposed to QMP from adult emergence rapidly lose their ability to detect it, an effect mediated by QMP's effects on the gene expression of antennal odorant receptors (Vergoz et al., 2009). It would therefore be expected under this proposed model that workers housed without QMP see neither increased attraction to QMP nor interactive depression of brain dopamine levels following imidacloprid exposure, as was observed in this study.

Another factor that underlies worker attraction to QMP is worker reproductive capacity, where workers with smaller ovaries (i.e. fewer ovarioles) show greater attraction to the queen and her pheromones (Kocher *et al.*, 2010). Ovariole numbers are determined during development (refer Section 1.2.1, General Introduction), and are under the control of genetic (Makert *et al.*, 2006; Robinson *et al.*, 1990) and environmental factors (Backx *et al.*, 2012). For instance, stressors during development such as nutrition limitation reduce ovariole numbers in adult honey bees (Hoover *et al.*, 2006). Though limited, there is also evidence of plasticity in ovariole numbers in adulthood via programmed cell death (Ronai *et al.*, 2017), raising the possibility that imidacloprid exposure during adulthood could act as a stressor and reduce ovariole numbers. However, empirical support for this hypothesis is limited due to uncertainty over the extent to which plasticity in ovariole number is environmentally responsive in adulthood, and whether such responses could occur within the ten day time-frame of this experiment.

Although the mechanisms underlying imidacloprid's effects on retinue response behaviour require further study, the consequences of this increased attraction to the queen for whole-colony function can still be considered. QMP maintains temporal polyethism among workers, and increased QMP exposure (as simulated by supplemental QMP treatment of colonies) has been shown to delay the onset of foraging (Pankiw *et al.*, 1998). Increased attraction, and hence exposure to, QMP could therefore decrease forager numbers in the hive by delaying the transition to foraging, leading to diminished food stores and poorer colony health. This idea is supported by a study of differentially expressed genes (DEGs) in honey bee workers following neonicotinoid exposure (Christen *et al.*, 2016); In their study, neonicotinoid treatment led to an up-regulation of *vitellogenin*, a gene associated with delayed foraging onset. Through its suppression of JH titres in workers, vitellogenin is thought to act as regulator of behavioural maturation, with reductions in vitellogenin associated with the transition to foraging (Amdam & Omholt, 2002; Hartfelder *et al.*, 2013). Indeed, down-regulation of *vitellogenin* using RNAi has been found to accelerate the behavioural shift to foraging and produce precocious foragers (Flatt *et al.*, 2013). As *vitellogenin* expression is regulated by QMP exposure (Nakaoka *et al.*, 2008; Duncan, unpublished data - see Appendix L, Figure L.1), the increased attraction to QMP demonstrated here could provide a missing explanatory link between neonicotinoid treatment and up-regulation of *vitellogenin*, which has been shown to restrict the transition of workers to foraging. Evidence from the field has previously shown that neonicotinoid exposure reduces foraging activity in the hive, though this has formerly been attributed to disturbed memory formation (Fischer *et al.*, 2014; Schneider *et al.*, 2012). Here, imidacloprid's increased attraction of workers to QMP is proposed as an alternative or additional explanation for these previously reported forager losses.

In the present study, locomotor activity was measured as a proxy for foraging activity to test QMP's effects on the transition to foraging. There was no evidence to support an effect of QMP on locomotor activity levels in five-day old workers, nor for any disruptive effects of imidacloprid. However, the extent to which activity levels can truly be viewed as a proxy for foraging activity is limited. Firstly, although high levels of locomotor activity are predictors of good foraging performance (Klein et al., 2019), phototaxis (movement towards light) is more closely linked with the transition from work inside the hive to outside, such as foraging (Tosi & Nieh, 2017). Circadian differences in locomotor activity are also tied to the onset of foraging, as nurses carry out their tasks within the hive and therefore have no need to be responsive to daylight cycles (Bloch & Meshi, 2007). As the laboratory environment may intrinsically affect natural circadian rhythms, examining the phototaxis of workers would be a better behavioural proxy for testing the onset of foraging. Secondly, workers were tested at five-days old, which is likely too young for the transition to foraging to be observed, as this usually occurs between four and eight weeks into a worker's life (Ruepell et al., 2007). However, due to limitations of the lifespan of caged laboratory workers (refer Section 2.1.1, General Methods), studying laboratory workers over ten days old would not have been feasible and would require an entirely different experimental design. One possibility would be to use markrelease recapture to age-match workers (as used in Chapter three), releasing marked

workers into hives undergoing different imidacloprid exposure regimes and recapturing at different time points for behavioural testing. This would further test the hypothesis that neonicotinoid exposure delays the onset of foraging in workers with behavioural evidence, and retinue response assays could additionally be carried out to link the transition to foraging with responsiveness to QMP.

An unexpected but finding from this chapter is the speculative combined toxicity effect of imidacloprid with QMP. To my knowledge, previous laboratory-based assessments into the lethality of chronic neonicotinoid exposure have not been carried out in the presence of QMP (Reviewed by Cresswell, 2011). Based on these studies, the LC50 of imidacloprid is estimated to be 1,760 ppb, but findings presented here suggest this number may be lower when combined with QMP. In this study, QMP exposure reduced the survival rates of bees treated with just 5 ppb imidacloprid over ten days from 87% to 73% in experiment one.. At the low doses used here, these differences in mortality are relatively subtle, and this acute toxicity risk may not constitute much of a threat to hives as a whole. However, imidacloprid has been detected at concentrations of 6-206 ppb in the pollen stores of hives (Mullin et al., 2010), and possible combined toxicity effects with QMP at these upper levels are unknown. Until now, the concentrations of neonicotinoids encountered by bees have not been thought a concern for acute toxicity, with studies instead focusing on sub-lethal effects of these pesticides (e.g. Decourtye, Devillers, et al., 2004; Henry et al., 2012; Sandrock et al., 2014; Schneider et al., 2012; Williamson et al., 2014). If QMP enhances the lethality of imidacloprid, this raises renewed concern into the acute toxicity of this pesticide at field-realistic concentrations. Further assessment into the lethality of neonicotinoids, as well as novel pesticides undergoing development, in combination with QMP at a wider range of chronic doses is needed to address this potential underestimation of toxicity in laboratory studies.

As an economically important species providing essential agricultural pollination services, declines of honey bee populations are a worldwide concern. Laboratory studies are a useful tool for investigating pesticide toxicity, sublethal effects on behaviour and their underlying molecular mechanisms, but they lack many elements present in the hive, such as social feedback loops and freedom of workers to exhibit 'natural' behaviours (Henry *et al.*, 2015). It is demonstrated here that QMP is another important element that differs between laboratory and field-based studies. QMP is central to honey bee biology, and failure to include this pheromone in laboratory studies may cause consistent underestimation of neonicotinoid toxicity.

To my knowledge, the interactive effects of imidacloprid on key worker responses to QMP have also not previously been considered. Most notably, the increased attraction of workers to QMP caused by imidacloprid is predicted to have knock-on effects for hive functionality, and may be a driver of previously reported reductions of foraging activity in neonicotinoid-treated hives. Key future areas of study have been identified from this research, including 1. Improved understanding of the mechanism underlying imidacloprid's increase in worker attraction to QMP, specifically elucidating imidacloprid's effects on JH titres and antennal octopamine receptor expression, and 2. Empirical testing into whether imidacloprid's increased QMP attraction translates to delayed-onset foraging and an overall decline in hive foraging activity. Given the interactive effects of pesticides with this pheromone is an area which merits further research attention.

Chapter 7

General Discussion

7.1 Thesis Overview

The primary aim of this research was to improve our understanding of one of the major mechanisms maintaining colony cohesion in honey bees: the inhibition of worker reproduction by QMP (Hoover *et al.*, 2003; Khila & Abouheif, 2008; Ratnieks *et al.*, 2006). A substantial body of evidence already existed implicating biogenic amines, particularly dopamine, as mediators of QMP's command over the behaviour and ovarian physiology of workers (i.e. Beggs *et al.*, 2007; Dombroski *et al.*, 2003; Harris & Woodring, 1995; Sasaki & Nagao, 2001; Vergoz *et al.*, 2012). This research aimed to bring further clarity to the roles played by biogenic amines by 1. focusing on the "missing link" between QMP's alterations to biogenic amine levels in the brain and how this comes to effect peripheral ovarian tissue (Chapter three), and 2. testing the causal relationship between biogenic amine titres (in particular dopamine) and ovary development through a series of manipulative feeding trials (Chapter four).

Given the accumulation of evidence that the depression of biogenic amines in the brain and hemolymph by QMP inhibits ovary development in workers, I then asked the question as to how queens manage to evade these effects of their own pheromone (Chapter five). I explored caste-specific differences in QMP exposure levels as a possible mechanism of self-sterilization evasion in queens, but emphasise that this is an emerging area of research which requires further study.

Finally, I aimed to apply my understanding of the molecular action of QMP to an applied issue: pesticide-driven declines of honey bee populations. In Chapter six I examined the interactive effects of QMP with a commonly used neonicotinoid pesticide, finding most notably a behavioural shift in imidacloprid-treated workers towards increased attraction to QMP. Given this pheromone's role in maintaining division of labour and colony harmony, this behavioural change may have implications for whole-colony function. These findings emphasize the significance of QMP to honey bee biology, and highlight a

need for the inclusion of this pheromone inclusion in future assessments of pesticide risks to honey bees.

7.2 What evidence is there that biogenic amines mediate QMP's control of worker reproduction?

7.2.1 Dopamine

Prior to this research, an accumulating number of studies proposed an involvement of dopamine in QMP's control of worker reproduction. Based on the finding that brain dopamine levels are reduced in QR workers (Harris & Woodring, 1995; Sasaki & Nagao, 2001), at least in part due to direct modulation by the QMP component HVA (Beggs *et al.*, 2007), dopamine was identified as a key second messenger to QMP. This, along with the positive correlation between ovary development in QL workers (Sasaki & Nagao, 2001) and the acceleration of this development by dopamine feeding (Dombroski *et al.*, 2003), led some to hypothesize that QMP inhibits worker reproduction by reducing dopamine titres in workers (i.e. Sasaki *et al.*, 2021). While QMP's modulation of dopamine perhaps represented our best understood mechanism underlying QMP's effects on peripheral tissues such as the ovary, the evidence thus far was still largely circumstantial.

In this thesis, additional evidence was found to support an involvement of dopamine in mediating QMP's inhibition of reproduction. In Chapter three, the reduced brain dopamine in workers from QR relative to QL hives was consistent with previous work (Beggs *et al.*, 2007; Harris & Woodring, 1995; Sasaki & Nagao, 2001). Further, the finding that brain dopamine levels correlated positively with ovary activity in workers from QL laboratory microcolonies (Chapter three) confirmed the results of past studies (Sasaki & Nagao, 2001). The depression of dopamine was causally linked with ovarian repression in Chapter four, in which treatment with the dopamine synthesis inhibitor 3-iodo-L-tyrosine (iodotyrosine) blocked ovarian development in workers to a degree comparable with QMP exposure. This finding would be strengthened by an assessment of internal dopamine levels in iodotyrosine-fed workers to confirm A. its efficacy as a depressor of



Figure 7.1 Stages of oogenesis in a honey bee ovary. Diagram the morphology of an active honey bee ovariole showing the distinct stages of oogenesis: i) previtellogenic, ii) vitellogenesis, iii) choriogenesis and iv) ovulation. The production of mature oocytes can be thought of as occurring on a 'conveyor belt' of development going from right to left. However, reaching complete oocyte maturity is not an inevitable process, and ovary development in workers frequently does not progress beyond certain stages of oogenesis. The corresponding ovary score associated with the completion of each stage is indicated by green arrowheads. For images of each ovary score, see also Figure 2.2.

dopamine levels and B. whether its effects were tissue specific, i.e. to the brain or the hemolymph. However, considered alongside evidence for a role of dopamine in regulating reproduction in other species (e.g. *Polistes,* Sasaki *et al.,* 2007; fire ants, Boulay *et al.,* 2001; bumble bees, Bloch *et al* 2000, Sasaki *et al.,* 2017; refer Section 1.5.3, General Introduction), the findings from Chapters three and four add to an accumulating body of evidence that QMP's depression of dopamine contributes to its sterilizing effects in workers.

Preliminary evidence from this thesis indicates that dopamine's role could be to mediate the earliest stages of ovarian development. In Chapter thee, brain dopamine levels were found to correlate with ovary activity during the initial stages of oogenesis (i.e. scores 0-2; Figure 7.1). However, the drop in brain dopamine levels between the ovary stages of two and three (refer Figure 3.6) indicates a possible stabilising of dopamine synthesis once full ovarian maturity is reached. The ovary development of QL workers was found to be accelerated in laboratory microcolonies relative to the hive (Chapter three), possibly due to their improved nutrition from their diet of CBF (refer section 2.1.1, General Methods). Consistently throughout this thesis (Chapters three, five and six), there has been no difference in brain dopamine levels between 10-day old workers from

QR and QL laboratory microcolonies, despite clear differences in ovary activity. One possible interpretation of this is that as a number of laboratory-maintained workers already possess mature oocytes by 10-days old, this timepoint is too late to observe differences in dopamine levels that precede this point of ovary maturity. In the hive, where ovary development is slower, brain dopamine levels are clearly elevated in QL relative to QR workers (Chapter three; Harris & Woodring, 1995). In laboratory studies, however, differences in brain dopamine levels between QL and QR workers have only ever been observed at 2-days old (Beggs et al., 2007). At 2-days old, laboratory workers are still at the beginning of the pre-vitellogenic stage (i.e. developing from a score of 0 into a 1; Figure 7.1; Koudjil & Doumandji, 2008), whereas by 10-days old many workers are reaching peak vitellogenic stage and mature oocytes are being formed (i.e. developing from a score of 2 into a 3; Figure 7.1; Koudjil & Doumandji, 2008). Therefore, the initial elevation of dopamine following the loss of the queen may be the trigger promoting the onset of early stages of ovary development such as previtellogenesis and vitellogenesis (Figure 7.1). Given that Notch signalling also regulates development in the ovary at the earliest stages of oogenesis (Duncan et al., 2016), this is consistent with the idea that dopamine levels effect ovary activity by modulating the activity of Notch signalling in the ovary. This idea is discussed in more depth in section 7.3.3.

Future studies should seek to empirically test the hypothesis that dopamine regulates the earliest stages of oogenesis. A key experiment to test this would be to examine the temporal dynamics of changes in dopamine levels in QL laboratory workers at a series of timepoints so that changes preceding the onset of full ovarian maturity by day 10 can be captured. As the processes by which dopamine levels are modulated by the presence of QMP are not fully understood (though the QMP component HVA is thought to be involved; Beggs *et al.*, 2007), it would also be of interest to measure the expression of genes involved in dopamine biosynthesis using RT-qPCR. The processes by which QMP may regulate dopamine levels in workers are further discussed in Section 7.3.2.

7.2.2 Octopamine

In addition to dopamine, I also found evidence for a role of octopamine in mediating QMP's effects. In Chapter three, I demonstrated that worker octopamine titres in both the brain and hemolymph are depressed by QMP. Octopamine is known to act alongside JH as a behavioural pacemaker promoting the onset of foraging (Wagener-Hulme *et al.*, 1999), but responsiveness of this amine to QMP has not previously been demonstrated.

I speculate that there may be a role of octopamine alongside dopamine as a mediator of QMP's inhibition of worker reproduction. In Chapter three, I observed a positive correlation between brain octopamine titres and ovary activity in QL workers (Figure 3.6), though this was not statistically significant so should be interpreted with caution. However, the fact that the octopamine receptors *AmOA1* and *OA62R* are expressed in the honey bee ovary alongside dopamine receptors (Vergoz *et al.*, 2012; Duncan, unpublished data) also supports a role of octopamine in mediating ovary development. Octopamine is involved in the regulation of reproduction in a range of insects; For instance, egg laying is stimulated by the injection of octopamine in the rice leaf bug (Yamane, 2013) and the diamondback moth (Li *et al.*, 2020). In *D. melanogaster*, octopamine regulates ovulation and the muscle contractions involved in egg-laying (Meiselman *et al.*, 2018; White *et al.*, 2021). It is therefore likely that a conserved gonadotropic function of octopamine might also exist in honey bees.

As was carried out in Chapter four for dopamine, experimental manipulations of octopamine levels are necessary to test whether octopamine is a positive regulator of reproduction in honey bees. One study has addressed this (Salomon *et al.*, 2012), finding that oral octopamine treatment did not increase the propensity of QL workers to activate their ovaries. However, octopamine's role in *D. melanogaster* is in the regulation of oviposition (Meiselman *et al.*, 2018; White *et al.*, 2021), and if this function is conserved in honey bees one may not expect to see an effect on ovary development. In addition, octopamine treatment did not cause an elevation of octopamine titres in the brain in Salomon *et al.*, (2012)'s study, leading to uncertainty as to the effectiveness of

dietary octopamine supplementation as a delivery method. As was discussed in Chapter four with respect to dopamine supplementation methods, more sophisticated methodology than dietary administration of biogenic amines is required to better understand their role in regulating reproduction. In *D. melanogaster*, the GAL4/UAS system has been used to study octopamine's role in reproduction by manipulating the expression of octopamine synthesis and receptor genes (Deady & Sun, 2015; Lee *et al.*, 2009; Li *et al.*, 2015; Lim *et al.*, 2014). While the genetic tools available to honey bees are less advanced, knockdown of octopamine synthesis and receptor genes using RNAi (i.e. Jarosch & Moritz, 2011; Maori *et al.*, 2019), or oral application of an octopamine synthesis inhibitor could instead be used to investigate the potential role of octopamine in honey bee reproduction.

7.3 Mechanism of QMP-biogenic amine signalling

As well as testing the hypothesis that biogenic amines mediate QMP's induction of sterility, this research aimed to shed light on the specific mechanisms involved in this signalling pathway. The action of QMP can be broken down into three stages as depicted in Figure 7.2. 1.) Detection - this is the process by which QMP is initially sensed, 2.) Transmission - the process by which QMP's signal is transmitted to the ovaries, and 3.) Ovarian response - the processes acting in the ovary itself controlling development. In the above section, I presented the evidence we have for roles of dopamine and octopamine as mediators in this process. In the following section, I discuss what is known about the possible mechanisms of biogenic amine-QMP signalling with consideration of each of the three stages identified above.



Figure 7.2 Model of the transmission of QMP's signal to the ovary consisting of three stages: **1. Detection:** the process by which QMP is initially sensed, **2.) Transmission:** the process by which QMP's signal is transmitted to the ovaries, and **3.) Ovarian response**; the processes acting in the ovary itself controlling development.

7.3.1 How is QMP detected?

Investigating the modes of QMP detection by workers (Figure 7.1; Stage 1) did not fall within the scope of this thesis. However, this research has emphasised the gaps in our knowledge regarding this stage of the QMP signalling process.

Current understanding of QMP transfer throughout the hive is that QMP is picked up by nurses attending the queen (Allen, 1955) and distributed through the colony via workerworker interactions such as antennation, grooming and trophallaxis (Naumann *et al.*, 1991). This indicates that there are two possible modes of QMP uptake by workers: antennal (i.e. through worker-worker antennation) and oral (i.e. through grooming and trophallaxis). Physical contact with QMP has been shown to be necessary to induce full inhibition of ovary activity (Katzav-Gozansky *et al.*, 2004; Lovegrove *et al.*, 2020), confirming that QMP components are non-volatile. However, the relative importance of antennal vs oral detection methods cannot be deduced from this. The mode of QMP uptake is relevant for our understanding of the downstream processes involved in regulating biogenic amines (Chapter three and Chapter four). Antennal detection implicates direct neuronal connections with the brain as being involved in the transmission of QMP's signal (Carcaud *et al.*, 2015). Previous work has shown that QMP induces neuronal responses in the antennal lobe of honey bees (Roussel *et al.*, 2014), though this was linked only to behavioural responses to QMP and effects on ovarian development were not measured. Whether neuronal pathways are also involved in mediating QMP's effects on reproduction therefore remains unclear. Oral uptake, on the other hand, would likely bypass neuronal signalling channels and instead exert effects via the circulatory system. It is unclear whether antennal or oral exposure to QMP act alone or in combination to induce the effects of QMP. Future studies should focus on developing an assay that enables workers to physically antennate, but not ingest, QMP to determine whether the full suite of behavioural and physiological responses to QMP can be induced through antennal exposure alone.

Understanding the relative importance of these QMP uptake methods would also be a valuable addition to the work addressed in Chapter five, which focused on understanding the mechanisms behind QMP's lack of repressive effects in queens. In addition to the dose-specific effects of QMP found in chapter five, differential QMP uptake between queens and workers could hypothetically also be a driver of self-sterility evasion. For instance, seeing as the most likely route of QMP re-internalisation by queens is through ingestion (Naumann *et al.*, 1991; although this remains to be definitively tested), the lack of antennal QMP exposure in queens could mean that possible neuronal effects of QMP on ovary activity are bypassed. Further study assessing the relative importance of antennal vs oral QMP uptake is thus essential not only for appreciating the downstream processes mediating QMP's transmission, but also for the question of queen self-sterility evasion.

7.3.2 How are dopamine and octopamine modulated by QMP and transmitted to the ovary?

Research into the molecular action of QMP thus far has been centered predominantly on QMP's effects on the brain (i.e. Beggs *et al.*, 2007; Beggs & Mercer, 2009; Carcaud *et al.*, 2015), with little work addressing how these changes are linked with the modulation of peripheral ovary tissue. A key finding from this research was the detection of dopamine and octopamine in the hemolymph (Chapter three). This provides evidence that the activity of these amines on ovary development may be mediated at least in part by their functions as circulating neurohormones. Hemolymph levels of dopamine and octopamine were also both reduced in workers from QR relative to QL hives (although reductions in hemolymph dopamine were not statistically significant; Chapter three), suggesting either direct or indirect modulation by QMP or other queen pheromones.

For dopamine, its presence in the hemolymph begins to bridge the gap in our understanding of the transmission of QMP's signal to the ovary (Stage 2; Figure 7.2). However, the link between brain and hemolymph dopamine levels is unclear. In male honey bees, dopamine increases with age in the brain and hemolymph in parallel to mediate the onset of mating behaviour (Sasaki & Watanabe, 2022), with hemolymph dopamine thought to interact with the dopamine receptors AmDop1, Amdop2 and AmDop3 in the testes (Matsushima *et al.*, 2018). However, it is not known whether the same processes regulate dopamine levels in the brain and hemolymph, or if these represent independent mechanisms. The source of biogenic amines in the hemolymph is unclear; For instance, they could be synthesized in the hemocytes (i.e. as in the moth *Chilo suppressalis;* Wu *et al.*, 2015), secreted from peripheral tissues such as the fat body, or even secreted from the brain itself.

QMP is known to regulate dopamine levels in the brain at least in part by HVA, a component structurally very similar to dopamine which has the potential to bind to the dopamine receptor AmDop3 as an agonist (Beggs & Mercer, 2009). However, the mechanism behind HVA's effects on brain dopamine has not been fully elucidated. Sasaki

& Watanabe (2022) proposed two ways in which HVA may control brain dopamine: 1. Through antennal detection of HVA and the transmission of neural signals to influence brain dopamine levels, and 2. Through oral HVA intake, which modulates brain dopamine via the hemolymph. While the neural responses to HVA in the antennal lobes support the former (Carcaud *et al.*, 2015), the discovery of potentially QMP-responsive dopamine in the hemolymph supports the latter (Chapter three). It is possible that antennal and oral QMP detection modulate dopamine via different pathways simultaneously to one another; For instance, while antennal detection modulates neural signals in the brain (Carcaud *et al.*, 2015), oral QMP uptake may bypass the brain to directly or indirectly effect hemolymph dopamine. As discussed above (Section 7.3.1), studying the relative importance of antennal vs oral mechanisms of QMP uptake would aid our understanding of QMP's modulation of dopamine in each uptake pathway.

Less is known about the process by which QMP may modulate octopamine levels. Beggs & Mercer (2009) show that HVA's activation of the dopamine receptor AmDop3 in the brain is highly selective, with neither agonist nor antagonist activity on the octopamine receptor AmOA1. However, the role of receptor activation in HVA's modulation of dopaminergic pathways is unclear, and the effect of HVA on octopamine levels has not been definitively tested. The involvement of other pheromone components in the modulation of octopamine is also uncertain, and a screening of the effects of each component on octopamine titres would be a useful way to identify components worthy of further investigation. Such a screen should include, but not be limited to, the five major QMP components: 9-ODA, both enantiomers of 9-HDA, HOB and HVA (Slessor et al., 1988). However, it is important to note that QMP alone does not inhibit worker reproduction as effectively as a real queen (Maisonnasse *et al.*, 2010). Given the finding that octopamine was lowered in the hemolymph only in workers exposed to a real queen (as opposed to synthetic QMP alone; Chapter three), this highlights the potential importance of additional pheromones to QMP in the modulation of octopamine. Other queen components to examine include tergal gland compounds, which have also been implicated in mediating ovarian development (Princen et al., 2019; Wossler & Crewe,

1999), and brood pheromone (Maisonnasse, Lenoir, *et al.*, 2010). In addition to octopamine, it would also be of value to assess the effects of these compounds on dopamine levels, as it has not yet been determined whether other compounds act in addition to HVA to modulate dopamine levels. Similarly to octopamine, dopamine levels were also reduced to a lesser extent by synthetic QMP than by a real in-hive queen (Chapter three), and a second possible interpretation of this (i.e. in addition to the temporal interpretation discussed in section 7.2.1) is the existence of additional modulatory compounds beyond the five major QMP components that are included in the synthetic blend.

When it comes to attempting to understand how octopamine may regulate reproduction in honey bees, D. Melanogaster is our best understood model. In D. melanogaster, Octopamine's effects on reproduction are mediated in part by the activity of octopaminergic neurons, which innervate peripheral tissues including the reproductive tract (Pauls et al., 2018) as well as being located throughout the brain (Busch et al., 2009; Sherer et al., 2020). There is also evidence that octopamine effects reproduction indirectly via regulation of the gonadotropins 20E and JH (Gruntenko et al., 2007; Rauschenbach et al., 2007). In honey bees, given the loss of gonadotropic function of JH (Rodrigues & Flatt, 2016), regulation of reproduction by octopamine would perhaps be more likely to function through direct interaction with ovarian tissue, either via neuronal connectivity, circulatory octopamine in the hemolymph or a combination of both. Support from the latter comes from the finding in Chapter three that octopamine was present in the hemolymph, and repressed by QMP in parallel with the brain. The relative importance of neuronal vs circulatory systems in the transduction of octopamine to the ovary may depend on the mode of detection used by workers to sense QMP, with antennal detection perhaps being more directly wired with neuronal connections than oral uptake.

In this research, I have demonstrated the potential for biogenic amines to act on the ovary via the circulatory system (Chapter three). However, the activity of octopaminergic and dopaminergic neurons was not assessed. It is possible that these two pathways act

together in the transduction of QMP's signal to the ovary, perhaps reflecting a degree of functional redundancy over antennal and oral QMP exposure. Further study using electrophysiology to examine the neuronal control of ovary development in honey bees would be a valuable addition to our knowledge, as have been carried out in *D. melanogaster* (Clark & Lange, 2003; Rodríguez-Valentín *et al.*, 2006) and the locust *Locusta migratoria* (Wong & Lange, 2014).

7.3.3 How is the ovarian response coordinated?

The final aspect of the mediation of QMP's effects on ovary development is the ovarian response (Figure 7.2; Stage three). Dopamine and octopamine could hypothetically mediate ovary physiology directly through activity with the ovarian receptors AmDop1, AmDop3 and AmOA1 (Vergoz *et al.*, 2012). QMP also effects the expression of these receptors (Vergoz *et al.*, 2012), and these changes in expression may act in parallel with modulation of dopamine and octopamine titres in the hemolymph to control ovary development, as is discussed in Chapter three.

There is preliminary evidence suggesting a link between dopamine receptor activation and Notch signalling in the ovary. Notch signalling acts in the germarium, the region of the ovary where germ cells are specified (see Figure 7.1), to actively inhibit oogenesis in response to the presence of QMP (Duncan *et al.*, 2016). Expression of the dopamine receptor *AmDop1* is also localised to the germarium (Duncan, unpublished data; see Appendix F, Figure F.1), and is up-regulated in two day old QL workers (Vergoz *et al.*, 2012). Assuming AmDop1 receptor activity is involved in regulating ovary development via the Notch signalling pathway, this is consistent with the idea that dopamine's role is to regulate the earliest checkpoints in oogenesis (as discussed in section 7.2.1). However, further study is required to ascertain whether there are links between ovarian biogenic amine receptor activity and the regulation of Notch signalling. This could theoretically be tested using RNAi to knockdown receptor expression in the ovaries and measuring the effects on ovary activation and the expression of Notch-responsive genes *Her* and *bHLH2* using RT-qPCR However, such an experiment is dependent on successful
tissue-specific gene knockdown, which we do not currently have the tools to achieve. In the absence of tissue-specific RNAi, the knockdown of biogenic amine receptors in the whole body would likely disrupt many fundamental processes and as such have lethal effects. An alternative approach would be to treat QR worker ovaries ex-vivo (i.e. in culture) with the levels of dopamine and octopamine found to be circulating in Chapter three. This could be carried out in the presence and absence of a transcriptional inhibitor, and RNA-seq could be used to assess the primary transcriptional response of the ovary to dopamine and octopamine. Such a study would confirm whether dopamine and octopamine's direct interactions with receptors in the ovary regulate Notchmediated reproductive constraint.

7.4 Worker reproduction is likely constrained by multiple redundant processes

While this thesis focused on understanding the role of just one candidate pathway that may be targeted by QMP, it also shines a light on the fact that biogenic amines are unlikely the sole mediators of QMP's effects. This is most apparent from findings in Chapter four, in which I observed that supplementary dopamine feeding did not restore ovary activity in workers exposed to QMP. In light of the hypothesised role of octopamine in regulating reproduction alongside dopamine (Chapter three), it is possible that co-administration of both dopamine and octopamine would be required to overcome QMP's inhibitory effects. An experiment to test this would be simple to conduct and would add to existing evidence for octopamine's involvement in mediating ovary repression.

This being said, there was a repeated occurrence throughout the thesis of inhibited ovary development in the absence of reductions to dopamine and octopamine in workers maintained in the laboratory (Chapter three, chapter five, chapter six). In experiments conducted using laboratory microcolonies, brain octopamine was inconsistently lowered by synthetic QMP (Chapter three, Chapter five, Chapter six), and brain dopamine was lowered by synthetic QMP only in conjunction with imidacloprid treatment (Chapter six). Neither dopamine nor octopamine were ever significantly lowered in the hemolymph by synthetic QMP in laboratory studies (Chapter three, Chapter six). Despite this, ovary activity was consistently inhibited by synthetic QMP across all chapters in the thesis. One interpretation of this, that it is the result of the temporal dynamics of ovary development in the lab vs the hive, was discussed above (Section 7.2.1), and requires further study over a range of timepoints to explicitly test. However, an alternative explanation is that synthetic QMP was inhibiting worker reproduction via a pathway that did not involve biogenic amines. This is at odds with the fact that HVA, a QMP component present in the synthetic blend used, has been directly implicated in the modulation of dopamine (Beggs *et al.*, 2007). However, Beggs *et al.*, (2007) exposed workers to a much greater amount of HVA than would typically be received within the QMP blend, in which 1 queen equivalent (QE) consists of just 2µg HVA (Slessor *et al.*, 1988) which is disseminated among tens of thousands of workers in the hive (Naumann *et al.*, 1991).

The idea that synthetic QMP may have inhibited reproduction through a process additional to biogenic amines is consistent with the finding that honey bee queens possess a functionally redundant set of pheromones (Princen *et al.*, 2019). Importantly, Princen *et al.*, (2019) demonstrate that worker ovary activity is repressed by the QMP components 9-ODA and 9-HDA, as well as tergal gland esters, as effectively as the full QMP blend. With the exception of HVA (Beggs *et al.*, 2007; Beggs & Mercer, 2009), little is known about the molecular action of other QMP components and queen pheromones, and this represents a crucial gap in our knowledge. While the modulation of biogenic amines represents our best-studied mechanism behind QMP's effects on worker ovaries, research into alternative pathways of mediation by the full range of the queen's pheromone components should not be neglected. In *D. melanogaster*, for instance, there is evidence that QMP inhibits oogenesis by modulating insulin signalling, causing an arrest in ovary development not dissimilar to a starvation response (Lovegrove *et al.*, 2021). A more holistic view of how QMP and other queen pheromones modulate the

broader neuroendocrine system is therefore essential to our understanding of how the queen maintains sterility in honey bee workers.

7.5 Queen control vs queen signal hypotheses

Whether queen pheromones more generally function as honest signals for the presence of a fecund queen (Keller & Nonacs., 1993; Monnin, 2006; Seeley, 1985) or as manipulative compounds that chemically enforce sterilization upon workers against their reproductive interests (Fletcher & Ross, 1985; Strauss et al., 2008; Hefetz, 2004; Hölldobler & Wilson, 1990) has been a source of much debate. These two contrasting hypotheses as to how queen pheromones may have evolved are known as the honest signal and queen control hypotheses (refer Section 1.4, General Introduction). Relative to the conserved class of cuticular hydrocarbon-based pheromones that constitute the queen pheromones of most eusocial hymenopterans (van Oystaeyen et al., 2014), QMP is highly derived and structurally complex (Hoover et al., 2003; Slessor et al., 1988; Refer Section 1.3.2, General Introduction). This contributes towards speculation that unlike cuticular hydrocarbons which likely arose from ancestral fertility signals (i.e. in line with the queen signal hypothesis; Chapuisat, 2014), QMP could have taken on a novel function as a chemical suppressor of reproduction (i.e. in line with the queen control hypothesis). Indeed, QMP's ability to repress reproduction across a broad phylogeny of species suggests that this highly derived queen pheromone may be targeting ancestral pathways involved in regulating reproduction (e.g. D. melanogaster: Camiletti et al., 2013; ant species: Carlisle & Butler., 1956; house fly: Nayar, 1963; termite: Hrdy et al 1960), suggesting a role as a chemical suppressor of reproduction. Dopamine and octopamine are thought to have roles in reproduction in a range of insect species (e.g. D. melanogaster: Lim et al., 2014, Rodríguez-Valentín et al., 2006; Polistes: Sasaki et al., 2007; fire ants: Boulay et al., 2001; bumble bees: Bloch et al 2000). Therefore, the modulation of these biogenic amines by QMP or other queen pheromones could be said to support the queen control hypothesis, particularly as it has been demonstrated that dopamine in particular is required for ovary development (Chapter four - inhibiting dopamine biosynthesis reduced ovary development in QL workers). Further, the fact that cuticular hydrocarbons do not share the same broad phylogenetic repressive effects as QMP (Lovegrove *et al.*, 2019) is consistent with the idea that QMP's possible role as a chemical suppressor of reproduction is a novel feature not seen in other more chemically and functionally simple queen pheromones.

This thesis also highlighted that QMP likely uses multiple processes to constrain reproduction, with biogenic amines representing just one aspect of a complex picture (discussed in Section 7.4). The finding that honey bee queens possess a functionally redundant set of pheromones has previously been argued in support of the queen signal hypothesis (Princen *et al.*, 2019). Princen *et al.*, (2019) argue that the complex combination of queen pheromone compounds could increase the information content of the queen's signal, making it a more reliable indicator of fecundity. However, the fact that QMP contains multiple bioactive compounds could equally be argued to support the queen control hypothesis. For instance, under the queen control model, it is predicted that queen pheromones should evolve rapidly owing to an evolutionary 'arms race' between queens a workers, whereby workers are continually selected to evade repression and queens are under pressure to evolve novel repressive agents (Heinze & D'Ettorre, 2009). The functional redundancy of QMP, along with additional honey bee queen pheromones, would be an expected outcome of such a queen-worker arms race.

A crucial contradiction of the queen control hypothesis is how the queen could evade any chemically repressive effects of her own pheromone. In Chapter five, this question was addressed, and the demonstrated dose-specific effects of QMP represent a possible mechanism underlying caste-specific differences in QMP's inhibition of reproduction. However, further study quantifying the specific dosages of QMP received by queens and workers is integral to the development of this theory. Further, as discussed previously (Section 7.3.1), the relative importance of antennal and oral QMP uptake deserves further study, as different routes of QMP exposure in queens and workers is also a possible mechanism by which caste-specific repressive effects of QMP could arise. While the findings in Chapter five do not completely negate the contradiction of queen fecundity inherent to the queen control hypothesis, they provide a useful framework from which future study can be directed.

7.6 Applied Research

Given the apparent complexity of QMP's effects on the neuroendocrine system in workers, in Chapter six I sought to determine whether this balance is disrupted by an environmental stressor: pesticide exposure. I tested the effects of a commonly used neonicotinoid pesticide, imidacloprid, on established worker responses to QMP, ranging from molecular, to physiological, to behavioural responses.

A key finding from this chapter was the increased attraction of imidacloprid-treated workers to QMP. Though the underlying mechanism behind this behavioural shift requires further study, its consequences have the potential to disrupt whole-colony function. Supplemental QMP exposure has been found to delay the behavioural shift of workers from nurses to foragers in the hive (Pankiw et al., 1998). Therefore, by increasing the attraction of workers to QMP, imidacloprid may lead to reduced forager recruitment in the hive, causing diminishing food stores and reduced colony productivity. This hypothesis is supported by empirical evidence from the field, in which neonicotinoid exposure has been found to reduce foraging activity in the hive (Fischer et al., 2014; Schneider et al., 2012). However, further study is required to robustly test this hypothesis. Firstly, an improved understanding of the mechanism underlying imidacloprid's increase in worker attraction to QMP is needed, with possible target pathways involving the modulation of JH titres and antennal expression of the octopamine receptor AmOA1 which is thought to correspond with worker attraction to QMP (Mcquillan et al., 2014; Vergoz et al., 2009). Secondly, empirical testing into whether an increase in attraction to QMP translates to delayed-onset foraging and an overall decline in foraging activity of the hive is required.

A second speculative finding from chapter six was the apparent increase in imidacloprid lethality when administered in combination with QMP. Though not consistent between experiments, QMP exposure was seen to reduce the survival rates of workers from laboratory microcolonies chronically exposed with 5 ppb imidacloprid from 87% in the absence of QMP to 73% in the presence of QMP. While these differences in mortality may be relatively subtle, it is worth noting that imidacloprid has been detected at concentrations of 6-206 ppb in the pollen stores of honey bee hives (Mullin *et al.*, 2010), and the combined toxicity effects with QMP at the upper end of this range is unknown. To my knowledge, previous laboratory-based assessments into the lethality of chronic neonicotinoid exposure have not been carried out in the presence of QMP (Reviewed by Cresswell, 2011), and interactivity with QMP is not a standard test that is considered during the screening process for the approval of the use of pesticides. This makes a case for the inclusion of QMP in future studies into the lethality of existing pesticides to honey bees, and equally of novel pesticides during their development.

Why such a combined toxicity effect of imidacloprid and QMP may exist is not fully understood. Based on other findings from this thesis, it is possible that the exacerbated depression of brain dopamine levels in the presence of both QMP and imidacloprid could contribute towards increased mortality. In chapter four, treating workers with the dopamine synthesis inhibitor iodotyrosine at a concentration of 0.5 mg/mL led to a marked increase in mortality, indicating that a certain amount of dopamine is required for basic survival functions. Long-term imidacloprid exposure causes a reduction in brain dopamine levels in *D. melanogaster* (Janner *et al.*, 2021), and the depression of dopamine by QMP and other queen pheromones has been demonstrated in this thesis (Chapter three). It is possible that this combination of processes, exacerbated by the increased attraction of workers to QMP (and hence increased exposure to its depressive effects), may reduce dopamine levels in workers to a point that survival is compromised.

7.7 Concluding remarks

A number of theories propose the ultimate reasons as to *why* eusociality may evolve, and these form an essential framework for reconciling the evolution of a sterile worker caste with the theory of evolution by natural selection (i.e. Hamilton, 1964b, 1964a; Nowak *et al.*, 2010; Wilson & Wilson, 2007). However, it is an appreciation of the proximate mechanisms by which worker sterility is maintained that informs us *how* the social structures as advanced as those seen in the eusocial insects came to be.

By negating conflict over queen-worker reproduction, QMP is central to the maintenance of colony harmony in honey bees (Hoover et al., 2003; Khila & Abouheif, 2008; Ratnieks et al., 2006). This thesis has provided an in-depth exploration into just one possible neuroendocrine signalling pathway by which QMP may exert its inhibition of ovary development in workers, finding that octopamine and dopamine are likely involved in the mediation of reproductive constraint. However, it also shines a light on the fact that through its multiple levels of redundancy (i.e. Princen et al., 2019), QMP may modulate the reproduction of workers via a plethora of different possible signalling pathways. Not only this, but it would appear that QMP is unlikely alone in its regulation of reproduction, with additional pheromones such as tergal esters and brood pheromone also implicated in the repression of ovary activity (Princen et al., 2019; Wossler & Crewe, 1999; Maisonnasse, Lenoir et al., 2010). The level of complexity within this system thus cannot be overstated. How such a functionally complex network of pheromones came to be, consisting of both suppressive agents alongside honest indicators of fertility, has the potential to teach us many lessons in how eusociality has arisen and is maintained. Further study into the specific mechanisms governing reproductive constraints is therefore essential to the progression of our understanding of the fascinating complexity of eusocial insect societies.

Appendix A

HPLC-FLD standard calibration curves



Figure A.1 HPLC-FLD standard calibration curves An example of a calibration curve used to quantify biogenic amines in biological samples from standards of known concentration.

Appendix B

Brain Dopamine						
Contrast	Estimate	SE	df	Z ratio	P value	
QL Day 10 – QL Day 21	0.000486	0.000546	Inf	0.890	0.8100	
QL Day 10 – QR Day 10	-0.001287	0.000454	Inf	-2.836	0.0236*	
QL Day 10 – QR Day 21	-0.002214	0.001080	Inf	-2.051	0.1694	
QL Day 21 – QR Day 10	-0.001773	0.000626	Inf	-2.833	0.0239*	
QL Day 21 – QR Day 21	-0.002700	0.000985	Inf	-2.740	0.0313*	
QR Day 10 – QR Day 21	-0.000927	0.001122	Inf	-0.826	0.8419	
Brain Octopamine		•				
Contrast	Estimate	SE	df	Z ratio	P value	
QL Day 10 – QL Day 21	-0.000477	0.000286	Inf	-1.652	0.3492	
QL Day 10 – QR Day 10	-0.001135	0.000347	Inf	-3.258	0.0062**	
QL Day 10 – QR Day 21	-0.003423	0.001342	Inf	-2.549	0.0528	
QL Day 21 – QR Day 10	-0.000658	0.000412	Inf	-1.599	0.3791	
QL Day 21 – QR Day 21	-0.002946	0.001342	Inf	-2.195	0.1246	
QR Day 10 – QR Day 21	-0.002288	0.001374	Inf	-1.665	0.3425	
Hemolymph Octopamine	I	1	1			
Contrast	Estimate	SE	df	Z ratio	P value	
QL Day 10 – QL Day 21	-4.10e-04	0.000120	Inf	-3.403	0.0037**	
QL Day 10 – QR Day 10	-4.24e-04	0.000133	Inf	-3.180	0.0080**	
QL Day 10 – QR Day 21	-1.63e-03	0.000426	Inf	-3.828	0.0007***	
QL Day 21 – QR Day 10	-1.43e-05	0.000163	Inf	-0.088	0.9998	
QL Day 21 – QR Day 21	-1.22e-03	0.000434	Inf	-2.810	0.0255*	
QR Day 10 – QR Day 21	-1.21e-03	0.000438	Inf	-2.750	0.0303*	

Table of *post-hoc* pairwise comparisons from Chapter 3

Table B.1 Pairwise comparisons of biogenic amine levels in QR/QL workers on days 10/21 *Post-hoc* comparisons of dopamine levels in the brain and octopamine levels in the brain and hemolymph between all pairwise combinations of QL and QR workers at days 10 and 21. *Post-hoc* testing was computed using estimated marginal means on the maximal GLMM. GLMMs were fitted with inverse link functions, hence contrast estimates are still on the inverse scale. P-values are Tukey adjusted.

Appendix C

Assessment of dopamine dosages received between treatments



Figure C.1 Actual dopamine doses received per bee per day. The actual dopamine doses received in each treatment group were calculated by multiplying the mean food intake (mg/bee/day) for each treatment by the concentration of dopamine supplied in the food. The actual quantity of dopamine received differed significantly between treatments (Kruskall-Wallis; $\chi^2 = 150.4$, df = 4, p < 0.001). Pairwise comparisons were made using Wilcoxon rank sum tests with correction for multiple testing (see Table C.1). Statistically significant pairwise comparisons are denoted by boxplots not sharing a letter.

	0	0.1	0.5	1
0.1	1.3e-12	-	-	-
0.5	1.3e-12	<2e-16	-	-
1	1.3e-12	<2e-16	1.0e-10	-
2	2.5e-13	<2e-16	<2e-16	3.9e=13

Table C.1 Pairwise comparisons of actual dopamine intake. P-values from *post-hoc* pairwise comparisons of mean dopamine doses received in each treatment group (dopamine dose mg/g) using Wilcoxon rank sum tests with correction for multiple testing.

Appendix D



Assessment of L-Dopa dosages received between treatments

Figure D.1 Actual L-Dopa doses received per bee per day. The actual dopamine doses received in each treatment group were calculated by multiplying the mean water intake (μ L/bee/day, where 1 mg water was assumed equivalent to 1 μ L) for each treatment by the concentration of L-Dopa supplied in the water. The actual quantity of L-Dopa received differed significantly between treatments (Kruskall-Wallis; χ 2 = 109.53, df = 3, p < 2.2e-16). Pairwise comparisons were made using Wilcoxon rank sum tests with correction for multiple testing (see Table D.1). Statistically significant pairwise comparisons (p < 0.001) are denoted by boxplots not sharing a letter.

	0	0.01	0.1
0.01	5.8e-12	-	-
1	5.8e-12	6.5e-11	-
0.5	5.8e-12	6.5e-11	6.5e-11

Table D.1 Pairwise comparisons of actual L-Dopa intake. P-values from *post-hoc* pairwise comparisons of mean L-Dopa doses received in each treatment group (L-Dopa dose mg/mL) using Wilcoxon rank sum tests with correction for multiple testing.

Appendix E



Assessment of iodotyrosine doses received between treatments

Figure E.1 Actual lodotyrosine doses received per bee per day. The actual iodotyrosine (IT) doses received differed significantly between each IT treatment group, but not between QMP+ and QMP- treatments. Actual IT doses received in each treatment group were calculated by multiplying the mean water intake (μ L/bee/day, where 1 mg water was assumed equivalent to 1 μ L) for each treatment by the concentration of IT supplied in the water. Differences in the quantity of IT received between treatment groups were assessed using negative binomial generalised linear mixed effects models (GLMMs). The negative binomial distribution was selected due to zero-inflation of the raw data. IT dose, QMP presence and their interaction were included as fixed effects. To account for repeated measures, day was included as a random effect nested within replicate for all models. The statistical significance of QMP and IT dose supplied on IT dose received was determined by comparing the likelihood ratio of the maximal model to that of the model without the fixed effect of interest. *Post-hoc* pairwise comparisons were made using estimated marginal means on the full GLMM (see Table E.1). Statistically significant pairwise comparisons (p < 0.001) are denoted by boxplots not sharing a letter.

Pairwise post-hoc comparisons of actual IT intake between QMP/IT treatments							
Contrast	Estimate	SE	df	Z-ratio	P-value		
00-10	0.040	0.3976	Inf	0.101	1.0000		
00-00.01	-20.437	0.4630	Inf	-44.145	<.0001		
00-10.01	-20.328	0.5491	Inf	-37.021	<.0001		
00-00.1	-22.713	0.4219	Inf	-53.839	<.0001		
00-10.1	-22.503	0.4309	Inf	-52.223	<.0001		
00-00.5	-24.474	0.4130	Inf	-59.260	<.0001		
00-10.5	-24.292	0.4145	Inf	-58.608	<.0001		
10-00.01	-20.477	0.6006	Inf	-34.093	<.0001		
10-10.01	-20.368	0.5821	Inf	-34.992	<.0001		
10-00.1	-22.753	0.5280	Inf	-43.090	<.0001		
10-10.1	-22.543	0.5213	Inf	-43.243	<.0001		
10-00.5	-24.514	0.5144	Inf	-47.653	<.0001		
10-10.5	-24.332	0.5141	Inf	-47.330	<.0001		
0 0.01 - 1 0.01	0.109	0.4308	Inf	0.252	1.0000		
0 0.01 - 0 0.1	-2.276	0.3298	Inf	-6.901	<.0001		
0 0.01 - 1 0.1	-2.066	0.3376	Inf	-6.120	<.0001		
0 0.01 - 0 0.5	-4.037	0.3152	Inf	-12.807	<.0001		
0 0.01 - 1 0.5	-3.855	0.3171	Inf	-12.157	<.0001		
10.01-00.1	-2.385	0.3652	Inf	-6.530	<.0001		
10.01 - 10.1	-2.175	0.3674	Inf	-5.919	<.0001		
1 0.01 - 0 0.5	-4.145	0.3504	Inf	-11.830	<.0001		
10.01 - 10.5	-3.964	0.3502	Inf	-11.321	<.0001		
00.1-10.1	0.210	0.1671	Inf	1.256	0.9148		
00.1-00.5	-1.761	0.1218	Inf	-14.450	<.0001		
00.1-10.5	-1.579	0.1237	Inf	-12.766	<.0001		
10.1-00.5	-1.970	0.1337	Inf	-14.735	<.0001		
10.1 - 10.5	-1.789	0.1353	Inf	-13.226	<.0001		
0 0.5 - 1 0.5	0.181	0.0696	Inf	2.605	0.1539		

Table E.1 Pairwise comparisons of actual iodotyrosine intake. *Post-hoc* pairwise comparisons of actual IT intake between all QMP/IT dose treatment combinations computed using estimated marginal means. "0" encodes QMP- treatments and "1" encodes QMP+ treatments. P-values are Tukey adjusted to control for multiple testing. Note that results are given on the log scale and not the response scale.

Appendix F

Unpublished In Situ Hybridisation data showing AmDop1 expression in the honey bee ovary



Figure F.1 Expression of *AmDop1* **in the honey bee ovary.** *In Situ Hybridisation* shows *AmDop1* is expressed in both the germarium and vittelarium. Top: unpublished data, provided by Dr Duncan. Bottom: Figure S11 from Duncan *et al.*, 2020, used with permission.

Appendix G

Pairwise Post-hoc comparison	s of ovary sco	ores hetw	een O	MP/iodotvi	rosine treatments
Contrast	Estimate	SE	df	Z-ratio	P-value
absent 0 - absent 0.1	-1.31433	0.184	Inf	-7.151	<.0001***
absent 0 - present 0.1	-1.45868	0.182	Inf	-8.026	<.0001***
absent 0 - absent 0.5	-2.03199	0.203	Inf	-10.015	<.0001***
absent 0 - present 0.5	-2.14067	0.220	Inf	-9.738	<.0001***
present 0 - absent 0.01	0.83548	0.181	Inf	4.621	0.0001***
present 0 - present 0.01	-0.00969	0.176	Inf	-0.055	1.0000
present 0 - absent 0.1	0.00704	0.180	Inf	0.039	1.0000
present 0 - present 0.1	-0.13731	0.177	Inf	-0.775	0.9944
present 0 - absent 0.5	-0.71061	0.197	Inf	-3.614	0.0073**
present 0 - present 0.5	-0.81929	0.213	Inf	-3.840	0.0031**
absent 0.01 - present 0.01	-0.8456	0.177	Inf	-4.777	<.0001***
absent 0.01 - absent 0.1	-0.82843	0.181	Inf	-4.573	0.0001***
absent 0.01 - present 0.1	-0.97278	0.179	Inf	-5.442	<.0001***
absent 0.01 - absent 0.5	-1.54608	0.199	Inf	-7.755	<.0001***
absent 0.01 - present 0.5	-1.65477	0.216	Inf	-7.657	<.0001***
present 0.01 - absent 0.1	0.01673	0.176	Inf	0.095	1.0000
present 0.01 - present 0.1	-0.12762	0.173	Inf	-0.738	0.9959
present 0.01 - absent 0.5	-0.70092	0.193	Inf	-3.628	0.0069**
present 0.01 - present 0.5	-0.80960	0.210	Inf	-3.847	0.0030**
absent 0.1 - present 0.1	-0.14435	0.177	Inf	-0.813	0.9924
absent 0.1 - absent 0.5	-0.71765	0.197	Inf	-3.638	0.0067**
absent 0.1 - present 0.5	-0.82633	0.214	Inf	-3.858	0.0029**
present 0.1 - absent 0.5	-0.57330	0.194	Inf	-2.952	0.0628
present 0.1 - present 0.5	-0.68198	0.211	Inf	-3.227	0.0274*
absent 0.5 - present 0.5	-0.10868	0.226	Inf	-0.480	0.9997

Tables of *post-hoc* pairwise comparisons from Chapter 4

Table G.1 Ovary scores by QMP/iodotyrosine treatment *Post-hoc* pairwise comparisons of ovary scores between all pairwise comparisons of QMP/ iodotyrosine dose treatment combinations. 'absent' refers to QL microcolonies (QMP-) and 'present' refers to QR microcolonies (QMP+). Iodotyrosine doses are in mg/mL. *Post-hoc* tests were computed using estimated marginal means on the maximal CLMM. P-values are Tukey adjusted.

Pairwise post-hoc comparisons of water intake with QMP/iodotyrosine treatments						
Contrast	Estimate	SE	df	T-ratio	P-value	
absent 0 - present 0	7.406	1.65	217	4.487	0.0003***	
absent 0 - absent 0.01	1.433	1.65	217	0.868	0.9885	
absent 0 - present 0.01	4.162	1.65	217	2.522	0.1920	
absent 0 - absent 0.1	2.125	1.65	217	1.287	0.9028	
absent 0 - present 0.1	7.013	1.65	217	4.249	0.0008***	
absent 0 - absent 0.5	-2.087	1.65	217	-1.265	0.9108	
absent 0 - present 0.5	2.895	1.65	217	1.754	0.6515	
present 0 - absent 0.01	-5.973	1.65	217	-3.619	0.0087**	
present 0 - present 0.01	-3.244	1.65	217	-1.966	0.5075	
present 0 – absent 0.1	- 5.281	1.65	217	-3.200	0.0333*	
present 0 - present 0.1	-0.393	1.65	217	-0.238	1.0000	
present 0 - absent 0.5	-9.493	1.65	217	-5.752	<.0001***	
present 0 - present 0.5	4.511	1.65	217	-2.733	0.1184	
absent 0.01 - present 0.01	2.729	1.65	217	1.653	0.7172	
absent 0.01 - absent 0.1	0.692	1.65	217	0.419	0.9999	
absent 0.01 - present 0.1	5.580	1.65	217	3.381	0.0191*	
absent 0.01 - absent 0.5	-3.520	1.65	217	-2.133	0.3972	
absent 0.01 - present 0.5	1.462	1.65	217	0.886	0.9871	
present0.01 - absent 0.1	-2.037	1.65	217	-1.234	0.9208	
present 0.01 - present 0.1	2.851	1.65	217	1.728	0.6692	
present 0.01 - absent 0.5	-6.249	1.65	217	-3.786	0.0048**	
present 0.01 - present 0.5	-1.266	1.65	217	-0.767	0.9946	
absent 0.1 - present 0.1	4.888	1.65	217	2.962	0.0658	
absent 0.1 - absent 0.5	-4.212	1.65	217	-2.552	0.1797	
absent 0.1 - present 0.5	0.771	1.65	217	0.467	0.9998	
present 0.1 – absent 0.5	-9.100	1.65	217	-5.514	<.0001***	
present 0.1 - present 0.5	-4.118	1.65	217	-2.495	0.2032	
absent 0.5 - present 0.5	4.983	1.65	217	3.019	0.0562	

Table G.2 Water intake by QMP/iodotyrosine treatment *Post-hoc* pairwise comparisons of ovary scores between all pairwise comparisons of QMP/ iodotyrosine dose treatment combinations. 'absent' refers to QL microcolonies (QMP-) and 'present' refers to QR microcolonies (QMP+). Iodotyrosine doses are in mg/mL. *Post-hoc* tests were computed using least squares means on the maximal LMM. P-values are Tukey adjusted.

Pairwise post-hoc comparisons of survival between QMP/iodotyrosine treatments						
Contrast	Estimate	SE	df	Z-ratio	P-value	
absent 0 - present 0	-1.8264	0.764	Inf	-2.391	0.2454	
absent 0 - absent 0.01	-1.2803	0.802	Inf	-1.597	0.7525	
absent 0 - present 0.01	-1.7290	0.769	Inf	-2.249	0.3222	
absent 0 - absent 0.1	-2.0529	0.753	Inf	-2.727	0.1144	
absent 0 - present 0.1	-2.0642	0.753	Inf	-2.742	0.1102	
absent 0 - absent 0.5	-4.0030	0.716	Inf	-5.593	<.0001***	
absent 0 - present 0.5	-4.2804	0.714	Inf	-5.996	<.0001***	
present 0 - absent 0.01	0.5461	0.476	Inf	1.148	0.9460	
present 0 - present 0.01	0.0974	0.417	Inf	0.233	1.0000	
present 0 - absent 0.1	-0.2266	0.387	Inf	-0.585	0.9991	
present 0 - present 0.1	-0.2378	0.387	Inf	-0.614	0.9987	
present 0 - absent 0.5	-2.1766	0.309	Inf	-7.043	<.0001***	
present 0 - present 0.5	-2.4540	0.305	Inf	-8.048	<.0001***	
absent 0.01 – present 0.01	-0.4487	0.483	Inf	-0.928	0.9834	
absent 0.01 - absent 0.1	-0.7726	0.458	Inf	-1.688	0.6952	
absent 0.01 - present 0.1	-0.7839	0.458	Inf	-1.712	0.6792	
absent 0.01 - absent 0.5	-2.7227	0.394	Inf	-6.914	<.0001***	
absent 0.01 - present 0.5	-3.0001	0.391	Inf	-7.681	<.0001***	
present 0.01 - absent 0.1	-0.3240	0.397	Inf	-0.816	0.9923	
present 0.01 - present 0.1	-0.3352	0.397	Inf	-0.844	0.9905	
present 0.01 - absent 0.5	-2.2740	0.321	Inf	-7.081	<.0001***	
present 0.01 - present 0.5	-2.5514	0.317	Inf	-8.044	<.0001***	
absent 0.1 - present 0.1	-0.0112	0.365	Inf	-0.031	1.0000	
absent 0.1 - absent 0.5	-1.9501	0.281	Inf	-6.945	<.0001***	
absent 0.1 – present 0.5	-2.2274	0.276	Inf	-8.063	<.0001***	
present 0.1 - absent 0.5	-1.9388	0.281	Inf	-6.904	<.0001***	
present 0.1 - present 0.5	-2.2162	0.276	Inf	-8.022	<.0001***	
absent 0.5 - present 0.5	-0.2774	0.147	Inf	-1.891	0.5574	

Table G.3 Survival by QMP/iodotyrosine treatment *Post-hoc* pairwise comparisons of ovary scores between all pairwise comparisons of QMP/ iodotyrosine dose treatment combinations. 'absent' refers to QL microcolonies (QMP-) and 'present' refers to QR microcolonies (QMP+). Iodotyrosine doses are in mg/mL. *Post-hoc* tests were computed using least squares means on the maximal CPH. P-values are Tukey adjusted.

Appendix H

RT-qPCR

 Table H.1 Oligonucleotide sequences used for RT-qPCR

Target gene	HoneybeeID*	5' primer	3' primer					
Oligonucleotide sequences used to detect expression of reference genes								
Rnp2 265	GB52526	CGC CTG TAA TGG AAA	ACA CGT TCT TGT TGC					
Proteasome non-ATPase regulatory subunit 1		CTG AAA	TCA CG					
Oligonucleotide see	quences used to	detect expression of target	genes					
bHLH2	GB43790	GGG AAG CGG GAT CAA	AGT CTG GGC GAG GAG					
		GAT A	ATG TA					
Her	GB43788	ACC ACC ACC GTA GCA	ACT TTG GGG AGG CGT					
		TCA TC	GTA A					

* gene identifiers are from BeeBase genome assembly v4.5 official gene set version 3.2

QMP Dose (QE/µL)	Ovary score	N
0	0	3
	1	3
	2	3
	3	3
0.01	0	0
	1	3
	2	3
	3	0
0.1	0	3
	1	3
	2	3
	3	1
1	0	0
	1	3
	2	3
	3	3

Table H.2 n values for each QMP dose/ ovary score combination



Figure H.1 Validation of single reference gene usage Past gene expression data measured using RT-qPCR of the genes A. bHLH2 and B. Her in worker ovaries in different worker groups. Raw data from (Duncan et al., 2016) was provided by E.J. Duncan and used with permission. Relative expression was normalized to expression levels of just one housekeeping gene as reference (Rpn2; "1HKG", red boxplots) or the geometric mean of expression levels of Rpn2 and a second housekeeper, mRPL44 ("2HKG", blue boxplots). Differences in expression of each gene between groups were assessed using ANOVA. To determine whether the number of housekeeping genes used would have altered experimental outcomes, each analyses was carried out on expression data normalized to one vs two housekeeping genes (1HKG vs 2HKG). The number of housekeepers used for normalization had a subtle effect on effect sizes, but was ultimately inconsequential to the results of statistical analyses. bHLH2 expression differed significantly between treatment groups whether normalized to 1HKG (F(5) = 8.20, p<0.001) or 2HKG (F(5) = 7.96, p<0.001). Her expression also was significantly different between treatment groups whether normalized to 1HKG (F(5) = 8.00, p<0.001) or 2HKG (F(5) = 8.62, p<0.001). The use of *Rpn2* only as a reference gene for RT-qPCR is therefore deemed to be appropriate.

Appendix I

Validation of QMP efficacy experiment two



Figure I.1 Validation of QMP's repression of ovary activity in experiment two. A subset of bees from +/- QMP control (0ppb IMI) cages were dissected to verify that QMP was causing the expected ovary repression phenotype. 100 ovaries were scored from each treatment across 3 reps (40 from rep 1, 30 from rep 2, 30 from rep 3). Ovaries were scored using blinder software. A Chi-squared test was performed in R to test whether the proportions of each ovary score differed with QMP treatment. QMP treatment effectively reduced ovary development (Pearson's Chi Squared: $\chi 2 = 35.92$, df = 3, p < 0.0001), validating the efficacy of QMP treatments in experiment 2.

Appendix J







Appendix K

Experiment One – Solution Intake							
Contrast	Estimate	SE	Df	T-ratio	P-value		
QMP- 0 ppb – QMP+ 0 ppb	7.93	2.16	138	3.67	0.0046**		
QMP- 0 ppb – QMP- 1 ppb	2.79	2.18	138	1.28	0.7966		
QMP- 0 ppb – QMP+ 1 ppb	7.11	2.16	137	3.29	0.0157*		
QMP- 0 ppb – QMP- 5 ppb	1.48	2.23	139	0.66	0.9855		
QMP- 0 ppb – QMP+ 5 ppb	7.81	2.23	138	3.50	0.0082**		
QMP+ 0 ppb – QMP- 1 ppb	-5.13	2.19	138	-2.35	0.1819		
QMP+ 0 ppb – QMP+ 1 ppb	-0.82	2.17	138	-0.38	0.9990		
QMP+ 0 ppb – QMP- 5 ppb	-6.45	2.22	138	-2.90	0.0485*		
QMP+ 0 ppb – QMP+ 5 ppb	-0.12	2.25	139	-0.05	1.0000		
QMP- 1 ppb – QMP+ 1 ppb	4.32	2.18	137	1.99	0.3560		
QMP- 1 ppb – QMP- 5 ppb	-1.31	2.23	138	-0.59	0.9918		
QMP- 1 ppb – QMP+ 5 ppb	5.02	2.26	138	2.22	0.2354		
QMP+ 1 ppb – QMP- 5 ppb	-5.63	2.21	138	-2.55	0.1181		
QMP+ 1 ppb – QMP+ 5 ppb	0.70	2.24	138	0.31	0.9996		
QMP- 5 ppb – QMP+ 5 ppb	6.33	2.31	140	2.75	0.0730		
Experiment Two – Solution Ir	ntake						
Contrast	Estimate	SE	Df	T-ratio	P-value		
QMP- 0 ppb – QMP+ 0 ppb	7.69	1.67	151	4.60	0.0001***		
QMP- 0 ppb – QMP- 1 ppb	-1.06	1.67	151	-0.64	0.9882		
QMP- 0 ppb – QMP+ 1 ppb	6.46	1.65	151	3.91	0.0019**		
QMP- 0 ppb – QMP- 5 ppb	-0.97	1.65	151	-0.59	0.9918		
QMP- 0 ppb – QMP+ 5 ppb	5.67	1.67	151	3.40	0.0111*		
QMP+ 0 ppb – QMP- 1 ppb	-8.75	1.67	151	-5.24	<.0001***		
QMP+ 0 ppb – QMP+ 1 ppb	-1.22	1.65	151	-0.74	0.98		
QMP+ 0 ppb – QMP- 5 ppb	-8.65	1.65	151	-5.24	<.0001***		

Tables of *post-hoc* pairwise comparisons from Chapter six

QMP+ 0 ppb – QMP+ 5 ppb	-2.02	1.67	151	-1.21	0.8332
QMP- 1 ppb – QMP+ 1 ppb	7.52	1.65	151	4.55	0.0002***
QMP- 1 ppb – QMP- 5 ppb	0.09	1.65	151	0.05	1.0000
QMP- 1 ppb – QMP+ 5 ppb	6.73	1.67	151	4.03	0.0012**
QMP+ 1 ppb – QMP- 5 ppb	-7.44	1.64	151	-4.54	0.0002***
QMP+ 1 ppb – QMP+ 5 ppb	-0.79	1.65	151	-0.48	0.9968
QMP- 5 ppb – QMP+ 5 ppb	6.64	1.65	151	4.02	0.0013**
Experiment Two – Food Intal	ke				
Contrast	Estimate	SE	Df	T-ratio	P-value
QMP- 0 ppb – QMP+ 0 ppb	-1.80	0.35	154	-5.12	<0.0001***
QMP- 0 ppb – QMP- 1 ppb	-0.04	0.35	154	-0.11	1.0000
QMP- 0 ppb – QMP+ 1 ppb	-1.13	0.35	154	-3.24	0.0181*
QMP- 0 ppb – QMP- 5 ppb	0.47	0.35	154	1.36	0.7531
QMP- 0 ppb – QMP+ 5 ppb	-0.86	0.35	154	-2.46	0.1443
QMP+ 0 ppb – QMP- 1 ppb	1.76	0.35	154	5.01	<0.0001***
QMP+ 0 ppb – QMP+ 1 ppb	0.67	0.35	154	1.91	0.4002
QMP+ 0 ppb – QMP- 5 ppb	2.27	0.35	154	6.46	<0.001***
QMP+ 0 ppb – QMP+ 5 ppb	0.944	0.35	154	2.69	0.0841
QMP- 1 ppb – QMP+ 1 ppb	-1.09	0.35	154	-3.13	0.0251*
QMP- 1 ppb – QMP- 5 ppb	0.51	0.35	154	1.47	0.6870
QMP- 1 ppb – QMP+ 5 ppb	-0.82	0.35	154	-2.35	0.1824
QMP+ 1 ppb – QMP- 5 ppb	1.60	0.35	154	4.59	0.0001***
QMP+ 1 ppb – QMP+ 5 ppb	0.27	0.35	154	0.78	0.9700
QMP- 5 ppb – QMP+ 5 ppb	-1.33	0.35	154	-3.811	0.0027**

Table K.1 Pairwise *post-hoc* comparisons for food and solution intake in experiments one and two between all QMP and imidacloprid treatment combinations computed using estimated marginal means on the maximal LMM. P-values are Tukey adjusted.

Brian Dopamine							
Contrast	Estimate	SE	Df	Z-ratio	P-value		
QMP- 0 ppb – QMP+ 0 ppb	-1.05e-04	0.000210	Inf	-0.49	0.99		
QMP- 0 ppb – QMP- 1 ppb	2.44e-04	0.000359	Inf	0.68	0.98		
QMP- 0 ppb – QMP+ 1 ppb	1.68e-04	0.000359	Inf	0.47	0.99		
QMP- 0 ppb – QMP- 5 ppb	2.18e-04	0.000185	Inf	1.17	0.85		
QMP- 0 ppb – QMP+ 5 ppb	-6.44e-04	0.000238	Inf	-2.70	0.07		
QMP+ 0 ppb – QMP- 1 ppb	3.49e-04	0.000350	Inf	1.00	0.91		
QMP+ 0 ppb – QMP+ 1 ppb	2.72e-04	0.000349	Inf	0.79	0.97		
QMP+ 0 ppb – QMP- 5 ppb	3.23e-04	0.000187	Inf	1.73	0.51		
QMP+ 0 ppb – QMP+ 5 ppb	-5.39e-04	0.000246	Inf	-2.19	0.24		
QMP- 1 ppb – QMP+ 1 ppb	-7.63e-05	0.000251	Inf	-0.30	0.99		
QMP- 1 ppb – QMP- 5 ppb	-2.59e-05	0.000346	Inf	-0.08	1.00		
QMP- 1 ppb – QMP+ 5 ppb	-8.88e-04	0.000383	Inf	-2.34	0.18		
QMP+ 1 ppb – QMP- 5 ppb	5.04e-05	0.000346	Inf	0.15	1.00		
QMP+ 1 ppb – QMP+ 5 ppb	-8.11e-04	0.000382	Inf	-2.14	0.27		
QMP- 5 ppb – QMP+ 5 ppb	-8.62e-04	0.000225	Inf	-3.84	0.0017**		

Table K.2 Pairwise *post-hoc* comparisons of brain dopamine levels between all QMP and imidacloprid treatment combinations computed using estimated marginal means on the maximal GLMM. GLMM was fitted with an inverse link function, hence contrast estimates are still on the inverse scale. P-values are Tukey adjusted.

Experiment One Survival					
Contrast	Estimate	SE	Df	Z-ratio	P-value
QMP- 0 ppb – QMP+ 0 ppb	-0.22	0.27	Inf	-0.82	0.96
QMP- 0 ppb – QMP- 1 ppb	0.76	0.35	Inf	2.17	0.25
QMP- 0 ppb – QMP+ 1 ppb	0.70	0.25	Inf	-2.85	<0.05*
QMP- 0 ppb – QMP- 5 ppb	0.36	0.31	Inf	1.18	0.85
QMP- 0 ppb – QMP+ 5 ppb	0.71	0.27	Inf	-2.62	0.09
QMP+ 0 ppb – QMP- 1 ppb	0.98	0.34	Inf	2.89	<0.05*
QMP+ 0 ppb – QMP+ 1 ppb	-0.48	0.23	Inf	-2.10	0.29
QMP+ 0 ppb – QMP- 5 ppb	0.58	0.30	Inf	1.97	0.36
QMP+ 0 ppb – QMP+ 5 ppb	0.49	0.26	Inf	-1.91	0.39
QMP- 1 ppb – QMP+ 1 ppb	-1.47	0.32	Inf	-4.54	<0.001***
QMP- 1 ppb – QMP- 5 ppb	-0.40	0.37	Inf	-1.07	0.89
QMP- 1 ppb – QMP+ 5 ppb	-1.47	0.34	Inf	-4.31	<0.001***
QMP+ 1 ppb – QMP- 5 ppb	1.07	0.28	Inf	3.86	<0.01**
QMP+ 1 ppb – QMP+ 5 ppb	-0.01	0.23	Inf	-0.021	1.00
QMP- 5 ppb – QMP+ 5 ppb	-1.07	0.30	Inf	-3.60	<0.01**
Experiment Two Survival					
Experiment Two Surviva	1				
Experiment Two Survival Contrast	Estimate	SE	Df	Z-ratio	P-value
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb	Estimate -1.21	SE 0.21	Df Inf	Z-ratio -5.87	P-value <0.001***
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb	Estimate -1.21 -0.24	SE 0.21 0.24	Df Inf Inf	Z-ratio -5.87 -1.01	P-value <0.001*** 0.92
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 1 ppb	Estimate -1.21 -0.24 -0.23	SE 0.21 0.24 0.24	Df Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95	P-value <0.001***
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 1 ppb QMP- 0 ppb – QMP- 5 ppb	Estimate -1.21 -0.24 -0.23 -0.13	SE 0.21 0.24 0.24 0.34	Df Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56	P-value <0.001***
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 1 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP+ 5 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95	SE 0.21 0.24 0.24 0.34 0.21	Df Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53	P-value <0.001*** 0.92 0.93 0.99 <0.001***
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 1 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP+ 5 ppb QMP+ 0 ppb – QMP- 1 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97	SE 0.21 0.24 0.24 0.34 0.21 0.19	Df Inf Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16	P-value <0.001*** 0.92 0.93 0.99 <0.001*** <0.001***
Experiment Two Survival Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 1 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP+ 1 ppb QMP- 0 ppb – QMP+ 5 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP+ 1 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19	Df Inf Inf Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17	P-value <0.001*** 0.92 0.93 0.99 <0.001*** <0.001*** <0.001***
Experiment Two Survival Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 1 ppb QMP- 0 ppb – QMP+ 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP- 1 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98 1.07	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19 0.20	Df Inf Inf Inf Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17 5.50	P-value <0.001*** 0.92 0.93 0.99 <0.001*** <0.001*** <0.001***
Experiment Two Survival Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 1 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP+ 5 ppb QMP+ 0 ppb – QMP+ 5 ppb QMP+ 0 ppb – QMP+ 5 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98 1.07 0.25	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19 0.20 0.16	Df Inf Inf Inf Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17 5.50 1.63	P-value <0.001*** 0.92 0.93 0.99 <0.001*** <0.001*** <0.001*** <0.001***
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP+ 5 ppb QMP- 1 ppb – QMP+ 1 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98 1.07 0.25 0.01	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19 0.20 0.16 0.22	Df Inf Inf Inf Inf Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17 5.50 1.63 0.05	P-value <0.001***
Experiment Two Surviva Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP- 5 ppb QMP+ 0 ppb – QMP- 5 ppb QMP- 1 ppb – QMP- 5 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98 1.07 0.25 0.01 0.10	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19 0.20 0.16 0.22 0.23	Df Inf Inf Inf Inf Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17 5.50 1.63 0.05 0.45	P-value <0.001***
Experiment Two Survival Contrast QMP- 0 ppb – QMP+ 0 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 5 ppb QMP- 0 ppb – QMP- 5 ppb QMP- 0 ppb – QMP- 1 ppb QMP- 0 ppb – QMP+ 5 ppb QMP+ 0 ppb – QMP- 1 ppb QMP+ 0 ppb – QMP+ 5 ppb QMP+ 0 ppb – QMP+ 5 ppb QMP+ 1 ppb – QMP+ 5 ppb QMP- 1 ppb – QMP+ 5 ppb QMP- 1 ppb – QMP+ 5 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98 1.07 0.25 0.01 0.10 -0.72	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19 0.20 0.16 0.22 0.23 0.19	Df Inf Inf Inf Inf Inf Inf Inf Inf Inf	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17 5.50 1.63 0.05 0.45 -3.70	P-value <0.001*** 0.92 0.93 0.99 <0.001*** <0.001*** <0.001*** <0.001*** 0.58 1.00 1.00 <0.01**
Experiment Two SurvivalContrast $QMP-0 ppb - QMP+0 ppb$ $QMP-0 ppb - QMP-1 ppb$ $QMP-0 ppb - QMP+1 ppb$ $QMP-0 ppb - QMP-5 ppb$ $QMP-0 ppb - QMP+5 ppb$ $QMP+0 ppb - QMP-1 ppb$ $QMP+0 ppb - QMP-1 ppb$ $QMP+0 ppb - QMP-5 ppb$ $QMP+0 ppb - QMP-5 ppb$ $QMP+1 ppb - QMP+5 ppb$ $QMP-1 ppb - QMP-5 ppb$ $QMP-1 ppb - QMP-5 ppb$ $QMP+1 ppb - QMP-5 ppb$	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98 1.07 0.25 0.01 0.10 -0.72 0.09	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19 0.20 0.16 0.22 0.23 0.19 0.23	Df Inf Inf Inf Inf Inf Inf Inf Inf Inf In	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17 5.50 1.63 0.05 0.45 -3.70 0.40	P-value <0.001***
Experiment Two SurvivalContrastQMP- 0 ppb - QMP+ 0 ppbQMP- 0 ppb - QMP- 1 ppbQMP- 0 ppb - QMP- 1 ppbQMP- 0 ppb - QMP- 5 ppbQMP- 0 ppb - QMP- 5 ppbQMP+ 0 ppb - QMP- 1 ppbQMP+ 0 ppb - QMP- 1 ppbQMP+ 0 ppb - QMP- 5 ppbQMP+ 0 ppb - QMP- 5 ppbQMP- 1 ppb - QMP+ 5 ppbQMP- 1 ppb - QMP- 5 ppbQMP+ 1 ppb - QMP+ 5 ppbQMP+ 1 ppb - QMP+ 5 ppb	Estimate -1.21 -0.24 -0.23 -0.13 -0.95 0.97 0.98 1.07 0.25 0.01 0.10 -0.72 0.09 -0.73	SE 0.21 0.24 0.24 0.34 0.21 0.19 0.19 0.20 0.20 0.23 0.23 0.23 0.20	Df Inf Inf Inf Inf Inf Inf Inf Inf Inf In	Z-ratio -5.87 -1.01 -0.95 -0.56 -4.53 5.16 5.17 5.50 1.63 0.05 0.45 -3.70 0.40 -3.73	P-value <0.001*** 0.92 0.93 0.99 <0.001*** <0.001*** <0.001*** <0.001*** 0.58 1.00 1.00 <0.01** 1.00 <0.01**

Table K.3 – Pairwise *post-hoc* comparisons of survival distributions between all QMP and imidacloprid treatment combinations computed using estimated marginal means on the maximal CPH model with mixed effects.

Appendix L

Vitellogenin expression unpublished data



Figure L.1 Ovary expression of vitellogenin measured using RT-qPCR. Vitellogenin expression was measured in QR workers, QL workers with differing degrees of ovary activity, and Queens. Expression of vitellogenin is reduced in QL relative to QR workers, even where there are no changes in ovary morphology (i.e. scores of 0 and 1). Unpublished data provided by Dr Duncan.

References

- Adamo, S.A., Linn, C.E. & Hoy, R.R. (1995) The role of neurohormonal octopamine during "fight or flight" behavior in the field cricket *Gryllus bimaculatus*. *Journal of Experimental Biology*, 198, 1691–700
- Aizen, M. A., & Harder, L. D. (2009). The Global Stock of Domesticated Honey Bees Is Growing Slower Than Agricultural Demand for Pollination. *Current Biology*, 19(11), 915–918. https://doi.org/10.1016/j.cub.2009.03.071
- Akasaka, S., Sasaki, K., Harano, K. ichi, & Nagao, T. (2010). Dopamine enhances locomotor activity for mating in male honeybees (Apis mellifera L.). *Journal of Insect Physiology*, 56(9), 1160–1166. https://doi.org/10.1016/j.jinsphys.2010.03.013
- Allen, M. D. (1955). Observations on honeybees attending their queen. *The British Journal of Animal Behaviour*, *3*(2), 66–69. https://doi.org/10.1016/S0950-5601(55)80015-9
- Amdam, G. V., & Omholt, S. W. (2002). The regulatory anatomy of honeybee lifespan. Journal of Theoretical Biology, 216(2), 209–228. https://doi.org/10.1006/jtbi.2002.2545
- Backx, A. G., Guzmán-Novoa, E., & Thompson, G. J. (2012). Factors affecting ovary activation in honey bee workers: A meta-analysis. *Insectes Sociaux*, 59(3), 381– 388. https://doi.org/10.1007/s00040-012-0230-1
- Barkham, P. (2022) UK charities condemn 'betrayal' of allowing bee-killing pesticide in sugar beet crops. The Guardian, 14 Jan 2022. Available at: https://www.theguardian.com/environment/2022/jan/14/uk-charities-condemnbetrayal-of-allowing-bee-killing-pesticide-in-sugar-beet-crops (Accessed: 3rd May 2022)
- Bates, D., Maechler, M., Bolker, B., Walker, S. (2015). Fitting Linear Mixed-Effects Models Using Ime4. *Journal of Statistical Software*, 67(1), 1-48. doi:10.18637/jss.v067.i01.
- Beggs, K. T., Glendining, K. A., Marechal, N. M., Vergoz, V., Nakamura, I., Slessor, K. N., & Mercer, A. R. (2007). Queen pheromone modulates brain dopamine function in worker honey bees. *Proceedings of the National Academy of Sciences of the United States of America*, 104(7), 2460–2464. https://doi.org/10.1073/pnas.0608224104

- Beggs, K. T., & Mercer, A. R. (2009). Dopamine receptor activation by honey bee queen pheromone. *Current Biology : CB, 19*(14), 1206–1209. https://doi.org/10.1016/j.cub.2009.05.051
- Blenau, W., & Baumann, A. (2001) Molecular and Pharmacological Properties of Insect
 Biogenic Amine Receptors: Lessons From Drosophila melanogaster and Apis
 mellifera. Archives of Insect Biochemistry and Physiology, 48, 13–38
- Bloch, G., & Meshi, A. (2007). Influences of octopamine and juvenile hormone on locomotor behavior and period gene expression in the honeybee, Apis mellifera. *Journal of Comparative Physiology A: Neuroethology, Sensory, Neural, and Behavioral Physiology, 193*(2), 181–199. https://doi.org/10.1007/s00359-006-0179-5
- Bloch, G., Simon, T., Robinson, G. E., & Hefetz, A. (2000). Brain biogenic amines and reproductive dominance in bumble bees (*Bombus terrestris*). Journal of Comparative Physiology A 186, 261–268
- Blomquist, G.J., Vogt, R.G., (2003) Insect Pheromone Biochemistry and Molecular Biology: The Biosynthesis and Detection of Pheromones and Plant Volatiles (Elsevier Academic Press, Amsterdam, 2003)
- Bonfini, A., Wilkin, M. B., & Baron, M. (2015). Reversible regulation of stem cell niche size associated with dietary control of Notch signalling. *BMC Developmental Biology*, 15(1), 1–14. https://doi.org/10.1186/s12861-015-0059-8
- Boulay, R., Hooper-Bui, L., & Woodring, J. (2001b). Oviposition and oogenesis in virgin fire ant females Solenopsis invicta are associated with a high level of dopamine in the brain. *Physiological Entomology*, *26*, 294–299.
- Bourke, A. F. G. (1988). Worker Reproduction in the Higher Eusocial Hymenoptera. In *The Quarterly Review of Biology* (Vol. 63, pp. 291–311). The University of Chicago Press. https://doi.org/10.2307/2830426
- Bourke, A. F. G., & Ratnieks, F. L. W. (1999). Kin conflict over caste determination in social Hymenoptera. *Behavioral Ecology and Sociobiology*, 46(5), 287–297. https://doi.org/10.1007/s002650050622
- Bray, S. J. (2006). Notch signalling: A simple pathway becomes complex. *Nature Reviews Molecular Cell Biology*, 7(9), 678–689. https://doi.org/10.1038/nrm2009
- Burland, T. M., Bennett, N. C., Jarvis, J. U. M., & Faulkes, C. G. (2002). Eusociality in African mole-rats: New insights from patterns of genetic relatedness in the

Damaraland mole-rat (Cryptomys damarensis). *Proceedings of the Royal Society B: Biological Sciences, 269*(1495), 1025–1030. https://doi.org/10.1098/rspb.2002.1978

- Busch, S., Selcho, M., Ito, K., & Tanimoto, H. (2009). A map of octopaminergic neurons in the Drosophila brain. *Journal of Comparative Neurology*, *513*(6), 643–667. https://doi.org/10.1002/cne.21966
- Butler, C. G., & Fairey, E. M. (1963) The role of the queen in preventing oogenesis in worker honeybees. *J. Apic. Res.* 2, 14–18. doi:10.1080/00218839.1963.11100051
- Calabrese, E. J., & Baldwin, L. A. (2002). Defining hormesis. *Human & Experimental Toxicology*, *21*, 91–97.
- Cameron, R. C., Duncan, E. J., & Dearden, P. K. (2013). Biased gene expression in early honeybee larval development. *BMC Genomics*, 14(1), 1–12. https://doi.org/10.1186/1471-2164-14-903
- Camiletti, A. L., Percival-Smith, A., & Thompson, G. J. (2013). Honey bee queen mandibular pheromone inhibits ovary development and fecundity in a fruit fly. *Entomologia Experimentalis et Applicata*, 147(3), 262–268. https://doi.org/10.1111/eea.12071
- Capella, I. C. S., & Hartfelder, K. (1998). Juvenile hormone effect on DNA synthesis and apoptosis in caste-specific differentiation of the larval honey bee (Apis mellifera L.) ovary. *Journal of Insect Physiology*, 44(5–6), 385–391. https://doi.org/10.1016/S0022-1910(98)00027-4
- Carcaud, J., Giurfa, M., & Sandoz, J. C. (2015). Differential combinatorial coding of pheromones in two olfactory subsystems of the honey bee brain. *Journal of Neuroscience*, 35(10), 4157–4167. https://doi.org/10.1523/JNEUROSCI.0734-14.2015
- Carlisle, D. B., & Butler, C. G. (1956). The "Queen-Substance" of Honeybees and the Ovary-inhibiting Hormone of crustaceans. *Nature*, *177*(4502), 276–277. https://doi.org/10.1038/177276a0
- Casida, J. E., & Durkin, K. A. (2013). Neuroactive insecticides: Targets, selectivity, resistance, and secondary effects. *Annual Review of Entomology*, *58*, 99–117. https://doi.org/10.1146/annurev-ento-120811-153645

- Casida, J. E., & Quistad, G. B. (2004). Why Insecticides Are More Toxic To Insects Than People: The Unique Toxicology of Insects. In *Journal of Pesticide Science* (Vol. 29, pp. 81–86).
- Cassier, P., André, M., Pastor, D., Piulachs, M. D., & Bellés, X. (1993). AGE-DEPENDENT NEUROSECRETION RELEASE INDUCED BY DOPAMINE IN THE CORPORA CARDIACA OF BLATTELLA GERMANICA (L.) (DICTYOPTERA: BLATTELLIDAE). *Int. J. Insect Morphol & Embryol*, *22*(1), 1–11.
- Chapuisat, M. (2014). Smells Like Queen Since the Cretaceous. *Science*, *343*, 254–255. http://www.sciencedirect.com/science/article/pii/B9780444594259000238
- Chauzat, M.-P., Laurent, M., Ribiere-Chabert, M., & Hendrikx, P. (2016). A pan-European epidemiological study on honeybee colony losses 2012-2014 This report has been prepared by Cécile European Union Reference Laboratory for honeybee health (EURL) Anses Honeybee pathology Unit. *EPILOBEE*, 2013(August), 44.
- Christen, V., Kunz, P. Y., & Fent, K. (2018). Endocrine disruption and chronic effects of plant protection products in bees : Can we better protect our pollinators ? *Environmental Pollution*, 243, 1588–1601. https://doi.org/10.1016/j.envpol.2018.09.117
- Christen, V., Mittner, F., & Fent, K. (2016). Molecular Effects of Neonicotinoids in Honey Bees (*Apis mellifera*). *Environmental Science and Technology, 50*, 4071–4081 https://doi.org/10.1021/acs.est.6b00678
- Christensen, R. H. B. (2019). ordinal Regression Models for Ordinal Data. R package version 2019.12-10. https://CRAN.R-project.org/package=ordinal.
- Clark, J., & Lange, A. B. (2003). Octopamine modulates spermathecal muscle contractions in Locusta migratoria. *Journal of Comparative Physiology A: Neuroethology, Sensory, Neural, and Behavioral Physiology, 189*(2), 105–114. https://doi.org/10.1007/s00359-002-0375-x
- Cook, C. N., Lawson, S. P., Brent, C. S., & Rehan, S. M. (2019). Biogenic amines shift during the pre-reproductive to reproductive transition in the small carpenter bee, *Ceratina calcarata*. *Apidologie*, *50*(1), 1–10. https://doi.org/10.1007/s13592-018-0624-9
- Cothren, S., Meyer, J., & Hartman, J. (2018). Blinded Visual Scoring of Images Using the Freely-available Software Blinder. *Bio-Protocol*, 8(23), 1–10. https://doi.org/10.21769/bioprotoc.3103

- Cresswell, J. E. (2011). A meta-analysis of experiments testing the effects of a neonicotinoid insecticide (imidacloprid) on honey bees. *Ecotoxicology*, 20(1), 149– 157. https://doi.org/10.1007/s10646-010-0566-0
- Darwin, C. (1859). On The Origin of Species by Means of Natural Selection, or the Preservation of Favoured Races in the Struggle for Life, London, John Murray.
- Deady, L. D., & Sun, J. (2015). A Follicle Rupture Assay Reveals an Essential Role for Follicular Adrenergic Signaling in Drosophila Ovulation. *PLoS Genetics*, 11(10), 1– 21. https://doi.org/10.1371/journal.pgen.1005604
- Decourtye, A., Armengaud, C., Renou, M., Devillers, J., Cluzeau, S., Gauthier, M., & Pham-Delègue, M. H. (2004). Imidacloprid impairs memory and brain metabolism in the honeybee (Apis mellifera L.). *Pesticide Biochemistry and Physiology*, *78*(2), 83–92. https://doi.org/10.1016/j.pestbp.2003.10.001
- Decourtye, A., Devillers, J., Cluzeau, S., Charreton, M., & Pham-Delègue, M. H. (2004). Effects of imidacloprid and deltamethrin on associative learning in honeybees under semi-field and laboratory conditions. *Ecotoxicology and Environmental Safety*, *57*(3), 410–419. https://doi.org/10.1016/j.ecoenv.2003.08.001
- Dively, G. P., & Kamel, A. (2012). Insecticide residues in pollen and nectar of a cucurbit crop and their potential exposure to pollinators. *Journal of Agricultural and Food Chemistry*, *60*(18), 4449–4456. https://doi.org/10.1021/jf205393x
- Dixon, S. E., & Shuel, R. W. (1963). Studies in the Mode of Action of Royal Jelly in Honey bee Development III. The effect of experimental variation in diet on growth and metabolism of honeybee larvae. *Canadian Journal of Zoology*, *41*, 733–739.
- Dombroski, T., Simoes, Z. L. P., & Bitondi, M. M. G. (2003). Dietary dopamine causes ovary activation in queenless Apis mellifera workers. *Apidologie*, *34*, 281–289. https://doi.org/10.1051/apido
- Duffy, J. E. (1973). Eusociality in a coral-reef shrimp. *Nature*, *246*(5429), 170. https://doi.org/10.1038/246170a0
- Duncan, E. J., & Dearden, P. K. (2010). Evolution of a genomic regulatory domain: The role of gene co-option and gene duplication in the enhancer of split complex. *Genome Research*, 20(7), 917–928. https://doi.org/10.1101/gr.104794.109
- Duncan, E. J., Hyink, O., & Dearden, P. K. (2016). Notch signalling mediates reproductive constraint in the adult worker honeybee. *Nature Communications*, 7(1), 12427. https://doi.org/10.1038/ncomms12427

- Duncan, E. J., Leask, M. P., Dearden, P. K., & Wittkopp, P. (2020). Genome Architecture Facilitates Phenotypic Plasticity in the Honeybee (Apis mellifera). *Molecular Biology and Evolution*, *37*(7), 1964–1978. https://doi.org/10.1093/molbev/msaa057
- Evans, P. D. (1980). Biogenic amines in the insect nervous system. *Adv. Insect Physiol. 15,* 317–473. doi: 10.1016/S0065-2806(08)60143-5
- Farooqui, T. (2007) Octopamine-Mediated Neuromodulation of Insect Senses. Neurochemical Research 32, 1511–1529, doi:10.1007/s11064-007-9344-7
- Farooqui, T. (2013). A potential link among biogenic amines-based pesticides, learning and memory, and colony collapse disorder: A unique hypothesis. *Neurochemistry International*, 62(1), 122–136. https://doi.org/10.1016/j.neuint.2012.09.020
- Fischer, J., Müller, T., Spatz, A. K., Greggers, U., Grünewald, B., & Menzel, R. (2014). Neonicotinoids interfere with specific components of navigation in honeybees. *PLoS ONE*, 9(3), 1–10. https://doi.org/10.1371/journal.pone.0091364
- Flatt, T., Amdam, G. v, Kirkwood, T. B., & Omholt, S. W. (2013). Life-History Evolution and the Polyphenic Regulation of Somatic Maintenance and Survival. *The Quarterly Review of Biology*, 88(3), 185–218.
- Flatt, T., Tu, M. P., & Tatar, M. (2005). Hormonal pleiotropy and the juvenile hormone regulation of Drosophila development and life history. *BioEssays*, 27(10), 999– 1010. https://doi.org/10.1002/bies.20290
- Fletcher, D. J. C., & Ross, K. G. (1985). Regulation of reproduction in eusocial hymenoptera. *Annual Review of Entomology*, *30*, 319–343. http://dspace.library.uu.nl/handle/1874/254%0Ahttp://localhost/handle/1874/25 4
- Fuenzalida-Uribe, N., Meza, R. C., Hoffmann, H. A., Varas, R., & Campusano, J. M. (2013). NAChR-induced octopamine release mediates the effect of nicotine on a startle response in Drosophila melanogaster. *Journal of Neurochemistry*, *125*(2), 281–290. https://doi.org/10.1111/jnc.12161
- Gadagkar, R. (1993). And now... Eusocial thrips! Current Science, 64(4), 215–216.
- Gallai, N., Salles, J. M., Settele, J., & Vaissière, B. E. (2009). Economic valuation of the vulnerability of world agriculture confronted with pollinator decline. *Ecological Economics*, *68*(3), 810–821. https://doi.org/10.1016/j.ecolecon.2008.06.014

- Genersch, E. (2010). Honey bee pathology: Current threats to honey bees and beekeeping. Applied Microbiology and Biotechnology, 87(1), 87–97. https://doi.org/10.1007/s00253-010-2573-8
- Gill, R. J., Ramos-Rodriguez, O., & Raine, N. E. (2012). Combined pesticide exposure severely affects individual-and colony-level traits in bees. *Nature*, 491(7422), 105– 108. https://doi.org/10.1038/nature11585
- Goosey, M.W. & Candy, D.J. (1982) The release and removal of octopamine by tissues of the locust *Schistocerca gregaria*. *Insect Biochemisty*, *12*, 681–85
- Granger, N. A., Sturgis, S. L., Ebersohl, R., Geng, C., & Sparks, T. C. (1996). Dopaminergic Control of Corpora Allata Activity in the Larval Tobacco Hornworm, Manduca sexta. Archives of Insect Biochemistry and Physiology, 32(3–4), 449–466.
- Gruntenko, N. E., Karpova, E. K., Alekseev, A. A., Chentsova, N. A., Bogomolova, E. v., Bownes, M., & Rauschenbach, I. Y. (2007). Effects of octopamine on reproduction, juvenile hormone metabolism, dopamine, and 20-hydroxyecdysone contents in Drosophila. *Archives of Insect Biochemistry and Physiology*, 65(2), 85–94. https://doi.org/10.1002/arch.20187
- Gruntenko, N. E., Karpova, E. K., Alekseev, A. A., Chentsova, N. A., & Saprykina, Z. V. (2005). Effects of dopamine on juvenile hormone metabolism and fitness in *Drosophila virilis. Journal of Insect Physiology 51*, 959–968. https://doi.org/10.1016/j.jinsphys.2005.04.010
- Gruntenko, N. E., Laukhina, O. v, Bogomolova, E. v, Karpova, E. K., Menshanov, P. N., Romanova, I. v, & Rauschenbach, I. Y. (2012). Downregulation of the dopamine D2-like receptor in corpus allatum affects juvenile hormone synthesis in Drosophila melanogaster females. *Journal of Insect Physiology*, *58*(3), 348–355. https://doi.org/10.1016/j.jinsphys.2011.12.006
- Guruharsha, K. G., Kankel, M. W., & Artavanis-Tsakonas, S. (2012). The Notch signalling system: Recent insights into the complexity of a conserved pathway. *Nature Reviews Genetics*, 13(9), 654–666. https://doi.org/10.1038/nrg3272
- Haddad, F., Sawalha, M., Khawaja, Y., Najjar, A., & Karaman, R. (2018). Dopamine and levodopa prodrugs for the treatment of Parkinson's disease. *Molecules*, 23(1). https://doi.org/10.3390/molecules23010040
- Hamilton, W. D. (1964a). The genetical evolution of social behavior. I. *Group Selection*, 7, 1–16. https://doi.org/10.4324/9780203790427-4

- Hamilton, W. D. (1964b). The genetical evolution of social behavior. II. *Group Selection*, 7, 17–52. https://doi.org/10.4324/9780203790427-5
- Hammond, R. L., & Keller, L. (2004). Conflict over male parentage in social insects. *PLoS Biology*, 2(9). https://doi.org/10.1371/journal.pbio.0020248
- Harris, J. W., & Woodring, J. (1992). Effects of stress, age, season, and source colony on levels of octapamine, dopamine and serotonin in the honeybee (Apis mellifera L.) brain. J. Insect Physiol, 38(1), 29–35.
- Harris, J. W., & Woodring, J. (1995). Elevated brain dopamine levels associated with ovary development in queenless worker honey bees (Apis mellifera L.).
 Comparative Biochemistry and Physiology. Part C: Comparative, 111(2), 271–279. https://doi.org/10.1016/0742-8413(95)00048-S
- Hartfelder, K., Bitondi, M. M. G., Brent, C. S., Guidugli-Lazzarini, K. R., Simões, Z. L. P., Stabentheiner, A., Tanaka, É. D., & Wang, Y. (2013). Standard methods for physiology and biochemistry research in *Apis mellifera*. *Journal of Apicultural Research*, *52*(1), 1–48. https://doi.org/10.3896/IBRA.1.52.1.06
- Hartfelder, K., & SteinbrÜck, G. (1997). Germ cell cluster formation and cell death are alternatives in caste-specific differentiation of the larval honey bee ovary.
 Invertebrate Reproduction and Development, *31*(1–3), 237–250.
 https://doi.org/10.1080/07924259.1997.9672582
- Heinze, J., & D'Ettorre, P. (2009). Honest and dishonest communication in social Hymenoptera. *Journal of Experimental Biology*, *212*(12), 1775–1779. https://doi.org/10.1242/jeb.015008
- Hefetz A, Katzav-Gozansky T. (2004). Are multiple honeybee queen pheromones indicators for a queen-workers arms race? *Apiacta 39,* 44–52
- Henry, M., Béguin, M., Requier, F., Rollin, O., Odoux, J., Aupinel, P., Aptel, J.,
 Tchamitchian, S., & Decourtye, A. (2012). A Common Pesticide Decreases Foraging
 Success and Survival in Honey Bees. *Science*, *336*(April), 348–350.
 https://doi.org/10.1126/science.1215039
- Henry, M., Cerrutti, N., Aupinel, P., Decourtye, A., Gayrard, M., Odoux, J. F., Pissard, A., Rüger, C., & Bretagnolle, V. (2015). Reconciling laboratory and field assessments of neonicotinoid toxicity to honeybees. *Proceedings of the Royal Society B: Biological Sciences, 282*(1819). https://doi.org/10.1098/rspb.2015.2110

- Hess, G. (1942) in Ein Beitrag zur Frage der Regulationen im Bienenstaat. *Beih. Schweiz. Bienen Ztg 2,* 33–111
- Hirashima, A., Sukhanova, M., Rauschenbach, I. (2000) Biogenic amines in *Drosophila virilis* under stress conditions. *Biosci Biotechnol Biochem*, *64*, 2625-2630
- Hölldobler B, Wilson EO. (1990). The Ants. Berlin, Heidelberg: Springer- Verlag.
- Holman, L. (2012). Costs and constraints conspire to produce honest signalling: Insights from an ant queen pheromone. *Evolution*, *66*(7), 2094–2105. https://doi.org/10.1111/j.1558-5646.2012.01603.x
- Holman, L. (2014). Bumblebee size polymorphism and worker response to queen pheromone. *PeerJ*, *2*, e604. https://doi.org/10.7717/peerj.604
- Holman, L., Jørgensen, C. G., Nielsen, J., & D'Ettorre, P. (2010). Identification of an ant queen pheromone regulating worker sterility. *Proceedings of the Royal Society B: Biological Sciences*, *277*(1701), 3793–3800. https://doi.org/10.1098/rspb.2010.0984
- Holman, L., Lanfear, R., & D'Ettorre, P. (2013). The evolution of queen pheromones in the ant genus Lasius. *Journal of Evolutionary Biology*, 26(7), 1549–1558. https://doi.org/10.1111/jeb.12162
- Hoover, S. E. R., Keeling, C. I., Winston, M. L., & Slessor, K. N. (2003). The effect of queen pheromones on worker honey bee ovary development. *Naturwissenschaften*, *90*(10), 477–480. https://doi.org/10.1007/s00114-003-0462-z
- Hrdý, I., Novák, V.J.A., Škrobal, D., (1960). Influence of the queen inhibitory substance of honeybee on the development of supplementary reproductives in the termite. *Insectes Sociaux, 24*, 61–70
- Hsu, H. J., & Drummond-Barbosa, D. (2011). Insulin signals control the competence of the Drosophila female germline stem cell niche to respond to Notch ligands. *Developmental Biology*, 350(2), 290–300. https://doi.org/10.1016/j.ydbio.2010.11.032
- Hughes, W. O. H., Oldroyd, B. P., Beekman, M., & Ratnieks, F. L. W. (2008). Ancestral monogamy shows kin selection is key to the evolution of eusociality. *Science (New York, N.Y.)*, 320(5880), 1213–1216.

- Ichinose, T., Tanimoto, H. & Yamagata, N. (2017) Behavioral Modulation by Spontaneous Activity of Dopamine Neurons. *Frontiers in Systems of Neuroscience*, 11(88), doi: 10.3389/fnsys.2017.00088
- Janner, D. E., Gomes, N. S., Poetini, M. R., Poleto, K. H., Musachio, E. A. S., de Almeida, F. P., de Matos Amador, E. C., Reginaldo, J. C., Ramborger, B. P., Roehrs, R., Prigol, M., & Guerra, G. P. (2021). Oxidative stress and decreased dopamine levels induced by imidacloprid exposure cause behavioral changes in a neurodevelopmental disorder model in Drosophila melanogaster. *NeuroToxicology*, *85*(May), 79–89. https://doi.org/10.1016/j.neuro.2021.05.006
- Jarosch, A., & Moritz, R. F. A. (2011). Systemic RNA-interference in the honeybee Apis mellifera: Tissue dependent uptake of fluorescent siRNA after intra-abdominal application observed by laser-scanning microscopy. *Journal of Insect Physiology*, 57(7), 851–857. https://doi.org/10.1016/j.jinsphys.2011.03.013
- Jarvis, J. U. M. (1981). Eusociality in a mammal: Cooperative breeding in naked molerat colonies. *Science*, *212*(4494), 571–573. https://doi.org/10.1126/science.7209555
- Jay, S. C. (1968). Factors influencing ovary development of worker honeybees under natural conditions. *Canadian Journal of Zoology*, 46(3), 345–347. https://doi.org/10.1139/z68-052
- Jennings, B., Preiss, A., Delidakis, C., & Bray, S. (1994). The Notch signalling pathway is required for Enhancer of split bHLH protein expression during neurogenesis in the Drosophila embryo. *Development*, *120*(12), 3537–3548. https://doi.org/10.1242/dev.120.12.3537
- Jindra, M., Bellés, X., & Shinoda, T. (2015). Molecular basis of juvenile hormone signaling. *Current Opinion in Insect Science*, 11, 39–46. https://doi.org/10.1016/j.cois.2015.08.004
- Jowett, T. & Postlethwait, J. H. (1980) The regulation of yolk polypeptide synthesis in Drosophila ovaries and fat body by 20-hydroxyecdysone and a juvenile hormone analog. *Developmental Biology, 80,* 225–234
- Kassambara, A. (2020). ggpubr: 'ggplot2' Based Publication Ready Plots. R package version 0.4.0. https://CRAN.R-project.org/package=ggpubr
- Kassambara, A. (2021). rstatix: Pipe-Friendly Framework for Basic Statistical Tests. R package version 0.7.0. https://CRAN.R-project.org/package=rstatix
- Katzav-Gozansky, T., Boulay, R., Soroker, V., & Hefetz, A. (2004). Queen-signal modulation of worker pheromonal composition in honeybees. *Proceedings of the Royal Society B: Biological Sciences*, 271(1552), 2065–2069. https://doi.org/10.1098/rspb.2004.2839
- Keeling, C. I., Slessor, K. N., Higo, H. A., & Winston, M. L. (2003). New components of the honey bee (Apis mellifera L.) queen retinue pheromone. *Proceedings of the National Academy of Sciences of the United States of America*, 100(8), 4486–4491. https://doi.org/10.1073/pnas.0836984100
- Keller, L., & Nonacs, P. (1993). The Role of Queen Pheromones in Social Insects: Queen Control or Queen Signal? *Animal Behavior*, 45, 787–794.
- Ken Sasaki and Ken-ichi Harano. (2010). Multiple regulatory roles of dopamine in behavior and reproduction of social insects. *Trends in Entomology*, 6
- Khila, A., & Abouheif, E. (2008). Reproductive constraint is a developmental mechanism that maintains social harmony in advanced ant societies. *Proceedings of the National Academy of Sciences of the United States of America*, 105(46), 17884–17889. https://doi.org/10.1073/pnas.0807351105
- Khila, A., & Abouheif, E. (2010). Evaluating the role of reproductive constraints in ant social evolution. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 365(1540), 617–630. https://doi.org/10.1098/rstb.2009.0257
- Kimura, T., Ohashi, M., Crailsheim, K., Schmickl, T., Okada, R., Radspieler, G., & Ikeno, H. (2014). Development of a new method to track multiple honey bees with complex behaviors on a flat laboratory arena. *PLoS ONE*, *9*(1). https://doi.org/10.1371/journal.pone.0084656
- Klein, A. M., Vaissière, B. E., Cane, J. H., Steffan-Dewenter, I., Cunningham, S. A., Kremen, C., & Tscharntke, T. (2007). Importance of pollinators in changing landscapes for world crops. *Proceedings of the Royal Society B: Biological Sciences*, 274(1608), 303–313. https://doi.org/10.1098/rspb.2006.3721
- Klein, S., Pasquaretta, C., He, X. J., Perry, C., Søvik, E., Devaud, J. M., Barron, A. B., & Lihoreau, M. (2019). Honey bees increase their foraging performance and frequency of pollen trips through experience. *Scientific Reports*, 9(1), 1–10. https://doi.org/10.1038/s41598-019-42677-x
- Knapp, R. A., Norman, V. N., Rouse, J. L., Duncan, E. J. (2022). Environmentally Responsive Reproduction: Neuroendocrine Signalling and the Evolution of Eusociality. *Current Opinion In Insect Science*, In Press

- Kocher, S. D., Ayroles, J. F., Stone, E. A., & Grozinger, C. M. (2010). Individual Variation in Pheromone Response Correlates with Reproductive Traits and Brain Gene Expression in Worker Honey Bees. *PLoS ONE*, 5(2). https://doi.org/10.1371/journal.pone.0009116
- Kocher, S. D., & Grozinger, C. M. (2011). Cooperation , Conflict , and the Evolution of Queen Pheromones. J Chem Ecol, 37, 1263–1275. https://doi.org/10.1007/s10886-011-0036-z
- Kocher, S. D., Richard, F. J., Tarpy, D. R., & Grozinger, C. M. (2008). Genomic analysis of post-mating changes in the honey bee queen (Apis mellifera). *BMC Genomics*, 9, 1–15. https://doi.org/10.1186/1471-2164-9-232
- Kocher, S. D., Richard, F., Tarpy, D. R., Grozinger, C. M., & State, N. C. (2009). Queen reproductive state modulates pheromone production and queen-worker interactions in honeybees. *Behavioural Ecology*, 20, 1007–1014. https://doi.org/10.1093/beheco/arp090
- Koudjil, M., & Doumandji, S. E. (2008). Ultrastructural characterization of the ovarian stages of the worker bee. *Comptes Rendus - Biologies*, 331(3), 185–197. https://doi.org/10.1016/j.crvi.2008.01.001
- Kucharski, R., Maleszka, J., Foret, S., & Maleszka, R. (2008). Nutritional control of reproductive status in honeybees via DNA methylation. *Science*, *319*(5871), 1827– 1830. https://doi.org/10.1126/science.1153069
- le Conte, Y., & Hefetz, A. (2008). Primer pheromones in social hymenoptera. Annual Review of Entomology, 53, 523–542. https://doi.org/10.1146/annurev.ento.52.110405.091434
- Lee, H. G., Rohila, S., & Han, K. A. (2009). The octopamine receptor OAMB mediates ovulation via Ca2+/calmodulin-dependent protein kinase II in the Drosophila oviduct epithelium. *PLoS ONE*, 4(3), 1–9. https://doi.org/10.1371/journal.pone.0004716
- Lee, K. v., Steinhauer, N., Rennich, K., Wilson, M. E., Tarpy, D. R., Caron, D. M., Rose, R., Delaplane, K. S., Baylis, K., Lengerich, E. J., Pettis, J., Skinner, J. A., Wilkes, J. T., Sagili, R., & VanEngelsdorp, D. (2015). A national survey of managed honey bee 2013–2014 annual colony losses in the USA. *Apidologie*, *46*(3), 292–305. https://doi.org/10.1007/s13592-015-0356-z

- Li, F., Li, K., Wu, L. J., Fan, Y. L., & Liu, T. X. (2020). Role of Biogenic Amines in Oviposition by the Diamondback Moth, Plutella xylostella L. *Frontiers in Physiology*, 11, 1–11. https://doi.org/10.3389/fphys.2020.00475
- Li, H., Liu, S., Chen, L., Luo, J., Zeng, D., & Li, X. (2021). Ecotoxicology and Environmental Safety Juvenile hormone and transcriptional changes in honey bee worker larvae when exposed to sublethal concentrations of thiamethoxam. *Ecotoxicology and Environmental Safety, 225,* 112744. https://doi.org/10.1016/j.ecoenv.2021.112744
- Li, Y., Fink, C., El-Kholy, S., & Roeder, T. (2015). The octopamine receptor octß2r is essential for ovulation and fertilization in the fruit fly drosophila melanogaster. *Archives of Insect Biochemistry and Physiology*, 88(3), 168–178. https://doi.org/10.1002/arch.21211
- Lim, J., Sabandal, P. R., Fernandez, A., Sabandal, J. M., Lee, H. G., Evans, P., & Han, K. A. (2014). The octopamine receptor Octβ2R regulates ovulation in Drosophila melanogaster. *PLoS ONE*, *9*(8). https://doi.org/10.1371/journal.pone.0104441
- Linksvayer, T. A., Kaftanoglu, O., Akyol, E., Blatch, S., Amdam, G. v., & Page, R. E. (2011). Larval and nurse worker control of developmental plasticity and the evolution of honey bee queen-worker dimorphism. *Journal of Evolutionary Biology*, 24(9), 1939–1948. https://doi.org/10.1111/j.1420-9101.2011.02331.x.Larval
- Linn, C. E., Poole, K. R., & Roelofs, W. L. (1994). Studies on biogenic amines and their metabolites in nervous tissue and hemolymph of adult male cabbage looper moths-I. Quantitation of photoperiod changes. *Comparative Biochemistry and Physiology. Part C: Pharmacology*, 108(1), 73–85. https://doi.org/10.1016/1367-8280(94)90092-2
- Litsey, E. M., Chung, S., & Fine, J. D. (2021). The Behavioral Toxicity of Insect Growth Disruptors on Apis mellifera Queen Care. *Frontiers in Ecology and Evolution*, 9, 729208. https://doi.org/10.3389/fevo.2021.729208
- Lovegrove, M. R., Dearden, P. K., & Duncan, E. J. (2019). Ancestral hymenopteran queen pheromones do not share the broad phylogenetic repressive effects of honeybee queen mandibular pheromone. *Journal of Insect Physiology*, *119*, 103968. https://doi.org/10.1016/j.jinsphys.2019.103968
- Lovegrove, M. R., Duncan, E. J., & Dearden, P. K. (2021). Honeybee Queen mandibular pheromone induces starvation in Drosophila melanogaster, implying a role for

nutrition signalling in the evolution of eusociality. *BioRxiv*. https://doi.org/https://doi.org/10.1101/2021.04.08.439099

- Lovegrove, M. R., Knapp, R. A., Duncan, E. J., & Dearden, P. K. (2020). Drosophila melanogaster and worker honeybees (Apis mellifera) do not require olfaction to be susceptible to honeybee queen mandibular pheromone. *Journal of Insect Physiology*, *127*, 104154. https://doi.org/10.1016/j.jinsphys.2020.104154
- Maisonnasse, A., Alaux, C., Beslay, D., Crauser, D., Gines, C., Plettner, E., & le Conte, Y. (2010). New insights into honey bee (Apis mellifera) pheromone communication. Is the queen mandibular pheromone alone in colony regulation? *Frontiers in Zoology*, *7*, 1–8. https://doi.org/10.1186/1742-9994-7-18
- Maisonnasse, A., Lenoir, J. C., Beslay, D., Crauser, D., & le Conte, Y. (2010). E-β-ocimene, a volatile brood pheromone involved in social regulation in the honey bee colony (apis mellifera). *PLoS ONE*, *5*(10). https://doi.org/10.1371/journal.pone.0013531
- Makert, G. R., Paxton, R. J., & Hartfelder, K. (2006). Ovariole number A predictor of differential reproductive success among worker subfamilies in queenless honeybee (Apis mellifera L.) colonies. *Behavioral Ecology and Sociobiology*, 60(6), 815–825. https://doi.org/10.1007/s00265-006-0225-x
- Maori, E., Garbian, Y., Kunik, V., Mozes-Koch, R., Malka, O., Kalev, H., Sabath, N., Sela,
 I., & Shafir, S. (2019). A Transmissible RNA Pathway in Honey Bees. *Cell Reports*, 27(7), 1949-1959.e6. https://doi.org/10.1016/j.celrep.2019.04.073
- Martin, S., Hogarth, A., van Breda, J., & Perrett, J. (1998). A scientific note on Varroa jacobsoni oudemans and the collapse of Apis mellifera L. colonies in the United Kingdom. *Apidologie*, *29*(4), 369–370. https://doi.org/10.1051/apido:19980407
- Matsushima, K., Watanabe, T., & Sasaki, K. (2018). Functional gene expression of dopamine receptors in the male reproductive organ during sexual maturation in the honey bee (Apis mellifera L.). *Journal of Insect Physiology*, *112*(August 2018), 9–14. https://doi.org/10.1016/j.jinsphys.2018.11.005
- Maynard Smith, J. (1964). Group selection and kin selection. *Nature, 201,* 1145–1147. https://doi.org/10.4324/9780203790427-8
- Mcquillan, H. J., Nakagawa, S., & Mercer, A. R. (2014). Juvenile Hormone Enhances Aversive Learning Performance in 2-Day Old Worker Honey Bees while Reducing Their Attraction to Queen Mandibular Pheromone. *Plos One*, *9*(11), 1–10. https://doi.org/10.1371/journal.pone.0112740

- Meiselman, M., Lee, S. S., Tran, R. T., Dai, H., Ding, Y., Rivera-Perez, C., Wijesekera, T.
 P., Dauwalder, B., Noriega, F. G., & Adams, M. E. (2017). Endocrine network essential for reproductive success in Drosophila melanogaster. *Proceedings of the National Academy of Sciences of the United States of America*, 114(19), E3849– E3858. https://doi.org/10.1073/pnas.1620760114
- Meiselman, M. R., Kingan, T. G., & Adams, M. E. (2018). Stress-induced reproductive arrest in Drosophila occurs through ETH deficiency-mediated suppression of oogenesis and ovulation. *BMC Biology*, 16(1), 1–15. https://doi.org/10.1186/s12915-018-0484-9
- Miller, D. G., & Ratnieks, F. L. W. (2001). The timing of worker reproduction and breakdown of policing behaviour in queenless honey bee (Apis mellifera L.) societies. *Insectes Sociaux*, 48(2), 178–184. https://doi.org/10.1007/pl00001762
- Misof, B., Liu, S.L., Meusemann, K., Peters, R.S., Donath, A., Mayer, C., Frandsen, P.B., Ware, J., Flouri, T., Beutel, R.G., Niehuis, O., Petersen, M., Izquierdo-Carrasco, F., Wappler, T., Rust, J., Aberer, A.J., Aspock, U., Aspock, H., Bartel, D., Blanke, A., Berger, S., Bohm, A., Buckley, T.R., Calcott, B., Chen, J.Q., Friedrich, F., Fukui, M., Fujita, M., Greve, C., Grobe, P., Gu, S.C., Huang, Y., Jermiin, L.S., Kawahara, A.Y., Krogmann, L., Kubiak, M., Lanfear, R., Letsch, H., Li, Y.Y., Li, Z.Y., Li, J.G., Lu, H.R., Machida, R., Mashimo, Y., Kapli, P., McKenna, D.D., Meng, G.L., Nakagaki, Y., Navarrete-Heredia, J.L., Ott, M., Ou, Y.X., Pass, G., Podsiadlowski, L., Pohl, H., von Reumont, B.M., Schutte, K., Sekiya, K., Shimizu, S., Slipinski, A., Stamatakis, A., Song, W.H., Su, X., Szucsich, N.U., Tan, M.H., Tan, X.M., Tang, M., Tang, J.B., Timelthaler, G., Tomizuka, S., Trautwein, M., Tong, X.L., Uchifune, T., Walzl, M.G., Wiegmann, B.M., Wilbrandt, J., Wipfler, B., Wong, T.K.F., Wu, Q., Wu, G.X., Xie, Y.L., Yang, S.Z., Yang, Q., Yeates, D.K., Yoshizawa, K., Zhang, Q., Zhang, R., Zhang, W.W., Zhang, Y.H., Zhao, J., Zhou, C.R., Zhou, L.L., Ziesmann, T., Zou, S.J., Li, Y.R., Xu, X., Zhang, Y., Yang, H.M., Wang, J., Wang, J., Kjer, K.M., Zhou, X. (2014) Phylogenomics resolves the timing and pattern of insect evolution. Science, 346, 763-767.
- Mohammedi, A., Paris, A., Crauser, D., & le Conte, Y. (1998). Effect of aliphatic esters on ovary development of queenless bees (Apis melliera L.). *Naturwissenschaften*, *85*(9), 455–458. https://doi.org/10.1007/s001140050531
- Monastirioti, M. (2003). Distinct octopamine cell population residing in the CNS abdominal ganglion controls ovulation in Drosophila melanogaster. *Developmental Biology*, *264*(1), 38–49. https://doi.org/10.1016/j.ydbio.2003.07.019

- Monnin, T. (2006). Chemical recognition of reproductive status in social insects. *Annuales Zoologici Fennici*, *43*(5), 515–530.
- Monnin, T., & Ratnieks, F. L. (2001). Policing in queenless ponerine ants. *Behavioral Ecology and Sociobiology*, *50*(2), 97–108. https://doi.org/10.1007/s002650100351
- Monnin, T., Ratnieks, F. L. W., Jones, G. R., & Beard, R. (2002). Pretender punishment induced by chemical signalling in a queenless ant. *Nature*, *419*(6902), 61–65. https://doi.org/10.1038/nature00932
- Mullin, C. A., Frazier, M., Frazier, J. L., Ashcraft, S., Simonds, R., vanEngelsdorp, D., & Pettis, J. S. (2010). High Levels of Miticides and Agrochemicals in North American Apiaries: Implications for Honey Bee Health. *PLoS ONE*, 5(3). https://doi.org/10.1371/journal.pone.0009754
- Nagatsua, T., & Sawadab, M. (2009). I-dopa therapy for Parkinson's disease: Past, present, and future. *Parkinsonism and Related Disorders*, *15*(1), 3–8. https://doi.org/10.1016/S1353-8020(09)70004-5
- Nakaoka, T., Takeuchi, H., & Kubo, T. (2008). Laying workers in queenless honeybee (Apis mellifera L.) colonies have physiological states similar to that of nurse bees but opposite that of foragers. *Journal of Insect Physiology*, 54(5), 806–812. https://doi.org/10.1016/j.jinsphys.2008.02.007
- National Center for Biotechnology Information (2022). PubChem Annotation Record for DOPAMINE, Source: Hazardous Substances Data Bank (HSDB). Retrieved April 28, 2022 from https://pubchem.ncbi.nlm.nih.gov/source/hsdb/3068.
- Naumann, K., Winston, M. L., Slessor, K. N., Prestwich, G. D., & Webster, F. X. (1991). Production and transmission of honey bee queen (Apis mellifera L.) mandibular gland pheromone. *Behavioral Ecology and Sociobiology*, 29(5), 321–332. https://doi.org/10.1007/BF00165956
- Nayar, J. K. (1963). Effect of Synthetic "Queen Substance" (9-oxodec-trans-2-enoic acid) on Ovary Development of the House-fly, Musca domestica L. *Nature*, 197(4870), 923–924. https://doi.org/10.1038/197923b0
- Neckameyer, W. S. (1996). Multiple roles for dopamine in Drosophila development. Developmental Biology, 176(2), 209–219. https://doi.org/10.1006/dbio.1996.0128
- Niño, E. L., Malka, O., Hefetz, A., Teal, P., Hayes, J., & Grozinger, C. M. (2012). Effects of honey bee (Apis mellifera L.) queen insemination volume on worker behavior and

physiology. *Journal of Insect Physiology*, *58*(8), 1082–1089. https://doi.org/10.1016/j.jinsphys.2012.04.015

- Noguchi, H., & Hayakawa, Y. (2001). Dopamine is a key factor for the induction of egg diapause of the silkworm, Bombyx mori. *European Journal of Biochemistry*, *268*(3), 774–780. https://doi.org/10.1046/j.1432-1327.2001.01933.x
- Nowak, M. A., Tarnita, C. E., & Wilson, E. O. (2010). The evolution of eusociality. *Nature*, *466*(7310), 1057–1062. https://doi.org/10.1038/nature09205
- Oi, C. A., van Zweden, J. S., Oliveira, R. C., van Oystaeyen, A., Nascimento, F. S., & Wenseleers, T. (2015). The origin and evolution of social insect queen pheromones: Novel hypotheses and outstanding problems. *BioEssays*, *37*(7), 808– 821. https://doi.org/10.1002/bies.201400180
- Okada, Y., Sasaki, K., Miyazaki, S., Shimoji, H., Tsuji, K., & Miura, T. (2015). Social dominance and reproductive differentiation mediated by dopaminergic signaling in a queenless ant. *Journal of Experimental Biology*, *218*(7), 1091–1098. https://doi.org/10.1242/jeb.118414
- Orchard, I., Carlisle, J.A., Loughton, B.G., Gole, J.W.D. & Downer, R.G.H. (1982) In vitro studies on the effects of octopamine on locust fat body. *General and Comparative Endocrinology*, 48, 7–13
- Pankiw, T., & Garza, C. (2007). Africanized and European honey bee worker ovarian follicle development response to racial brood pheromone extracts. *Apidologie*, *38*(2), 156–163. https://doi.org/10.1051/apido:2006066
- Pankiw, T., Huang, Z.-Y., Winston, M.-L., & Robinson, G.-E. (1998). Queen mandibular gland pheromone influences worker honey bee (Apis mellifera L.) foraging ontogeny and juvenile hormone titers. *Journal of Insect Physiology*, 44, 685–692.
- Pauls, D., Blechschmidt, C., Frantzmann, F., el Jundi, B., & Selcho, M. (2018). A comprehensive anatomical map of the peripheral octopaminergic/tyraminergic system of Drosophila melanogaster. *Scientific Reports*, 8(1), 11–14. https://doi.org/10.1038/s41598-018-33686-3
- Peeters, C., & Hölldobler, B. (1995). Reproductive cooperation between queens and their mated workers: The complex life history of an ant with a valuable nest.
 Proceedings of the National Academy of Sciences of the United States of America, 92(24), 10977–10979. https://doi.org/10.1073/pnas.92.24.10977

- Peeters, C., Liebig, J., & Hölldobler, B. (2000). Sexual reproduction by both queens and workers in the ponerine ant Harpegnathos saltator. *Insectes Sociaux*, 47(4), 325– 332. https://doi.org/10.1007/PL00001724
- Pendleton, R. G., Robinson, N., Roychowdhury, R., Rasheed, A., & Hillman, R. (1996).
 Reproduction and development in Drosophila are dependent upon catecholamines. *Life Sciences*, *59*(24), 2083–2091. https://doi.org/10.1016/S0024-3205(96)00562-0
- Pener, M. P., and Dhadialla, T. S. (2012). "Chapter one an overview of insect growth disruptors; applied aspects," in Advances in Insect Physiology Insect Growth Disruptors ed T. S. Dhadialla (Cambridge, MA: Academic Press), 1–162.
- Penick, C. A., Brent, C. S., Dolezal, K., & Liebig, J. (2014). Neurohormonal changes associated with ritualized combat and the formation of a reproductive hierarchy in the ant *Harpegnathos saltator*. *The Journal of Experimental Biology*, 217(Pt 9), 1496–1503. https://doi.org/10.1242/jeb.098301
- Peso, M., Elgar, M. A., & Barron, A. B. (2015). Pheromonal control: Reconciling physiological mechanism with signalling theory. *Biological Reviews*, 90(2), 542– 559. https://doi.org/10.1111/brv.12123
- Pfaffl, M. W. (2001). A new mathematial model for relative quantification in real-time RT-PCT. *Nucleic Acids Research*, *29*(9). https://doi.org/10.1093/nar/29.9.e45
- Pietrzak, D., Kania, J., Kmiecik, E., Malina, G., & Wątor, K. (2020). Fate of selected neonicotinoid insecticides in soil–water systems: Current state of the art and knowledge gaps. *Chemosphere*, 255. https://doi.org/10.1016/j.chemosphere.2020.126981
- Plettner, E., Otis, G. W., Wimalaratne, P. D. C., Winston, M. L., Slessor, K. N., Pankiw, T., & Punchihewa, P. W. K. (1997). Species- and caste-determined mandibular gland signals in honeybees (Apis). *Journal of Chemical Ecology*, *23*(2), 363–377. https://doi.org/10.1023/B:JOEC.0000006365.20996.a2
- Postlethwait, J. H. & Handler, A. M. (1979) The roles of juvenile hormone and 20hydroxy-ecdysone during vitellogenesis in isolated abdomens of Drosophila melanogaster. *Journal of Insect Physiology*, *25*, 455–460
- Potts, S. G., Biesmeijer, J. C., Kremen, C., Neumann, P., Schweiger, O., & Kunin, W. E. (2010). Global pollinator declines: Trends, impacts and drivers. *Trends in Ecology* and Evolution, 25(6), 345–353. https://doi.org/10.1016/j.tree.2010.01.007

- Princen, S. A., Oliveira, R. C., Ernst, U. R., Millar, J. G., van Zweden, J. S., & Wenseleers, T. (2019). Honeybees possess a structurally diverse and functionally redundant set of queen pheromones. *Proceedings of the Royal Society B: Biological Sciences*, 286(1905), 1–9. https://doi.org/10.1098/rspb.2019.0517
- Pyakurel, P., Shin, M., & Venton, B. J. (2018). Neurochemistry International Nicotinic acetylcholine receptor (nAChR) mediated dopamine release in larval Drosophila melanogaster. *Neurochemistry International*, *114*, 33–41. https://doi.org/10.1016/j.neuint.2017.12.012
- R Core Team (2021). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.Rproject.org/
- Rangel, J., Böröczky, K., Schal, C., Tarpy, D.R. (2016) Honey Bee (*Apis mellifera*) Queen Reproductive Potential Affects Queen Mandibular Gland Pheromone Composition and Worker Retinue Response. *PLoS ONE* 11(6) e0156027. doi:10.1371/ journal.pone.0156027
- Ratnieks, F. L. W., Foster, K. R., & Wenseleers, T. (2006b). Conflict resolution in insect societies. *Annual Review of Entomology*, *51*, 581–608. https://doi.org/10.1146/annurev.ento.51.110104.151003
- Ratnieks, F. L. W., & Reeve, H. K. (1992). Conflict in single-queen hymenopteran societies: the structure of conflict and processes that reduce conflict in advanced eusocial species. *Journal of Theoretical Biology*, *158*(1), 33–65. https://doi.org/10.1016/S0022-5193(05)80647-2
- Rauschenbach, I. Y., Bogomolova, E. v., Karpova, E. K., Adonyeva, N. v., Faddeeva, N. v., Menshanov, P. N., & Gruntenko, N. E. (2011). Mechanisms of age-specific regulation of dopamine metabolism by juvenile hormone and 20hydroxyecdysone in Drosophila females. *Journal of Comparative Physiology B: Biochemical, Systemic, and Environmental Physiology, 181*(1), 19–26. https://doi.org/10.1007/s00360-010-0512-8
- Rauschenbach, I. Y., Chentsova, N. A., Alekseev, A. A., Gruntenko, N. E., Adonyeva, N. v., Karpova, E. K., Komarova, T. N., Vasiliev, V. G., & Bownes, M. (2007). Dopamine and octopamine regulate 20-hydroxyecdysone level in vivo in Drosophila. *Archives of Insect Biochemistry and Physiology*, *65*(2), 95–102. https://doi.org/10.1002/arch.20183

- Reginato, R. D., & Cruz-Landim, C. (2003). Ovarian growth during larval development of queen and worker of Apis mellifera (Hymenoptera: Apidae): a morphometric and histological study. *Brazilian Journal of Biology*, 63(1), 121–127. https://doi.org/10.1590/S1519-69842003000100016
- Rennich, K., Kunkel, G., Abban, S., Fahey, R., Eversole, H., Evans, J., et al. (2014). 2013-2014 National Honey Bee Pests and Diseases Survey Report. Animal and Plant Health Inspection Service. Available online at: https://ushoneybeehealthsurvey.info/wp-content/uploads/sites/3/2020/04/2013-2014-NHBS-Report.pdf (accessed February 2, 2022).
- Richard, F. J., Tarpy, D. R., & Grozinger, C. M. (2007). Effects of insemination quantity on honey bee queen physiology. *PLoS ONE*, 2(10). https://doi.org/10.1371/journal.pone.0000980
- Riddiford, L. M. (2012). How does juvenile hormone control insect metamorphosis and reproduction? *General and comparative endocrinology, 179,* 477-484
- Rivera-Perez, C., Nouzova, M., & Noriega, F. G. (2012). A quantitative assay for the juvenile hormones and their precursors using fluorescent tags. *PLoS ONE*, 7(8), 1–8. https://doi.org/10.1371/journal.pone.0043784
- Robinson, G. E., Page, R. E., & Fondrk, M. K. (1990). Intracolonial behavioral variation in worker oviposition, oophagy, and larval care in queenless honey bee colonies. *Behavioral Ecology and Sociobiology*, *26*(5), 315–323. https://doi.org/10.1007/BF00171096
- Robinson, G. E., & Vargo, E. L. (1997). Juvenile hormone in adult eusocial hymenoptera: gonadotropin and behavioral pacemaker. *Archives of Insect Biochemistry and Physiology*, 35, 559–583. https://doi.org/10.1002/(SICI)1520-6327(1997)35:4<559::AID-ARCH13>3.0.CO;2-9
- Rodrigues, M. A., & Flatt, T. (2016). Endocrine uncoupling of the trade-off between reproduction and somatic maintenance in eusocial insects. *Current Opinion in Insect Science*, *16*(1995), 1–8. https://doi.org/10.1016/j.cois.2016.04.013
- Rodríguez-Valentín, R., López-González, I., Jorquera, R., Labarca, P., Zurita, M., & Reynaud, E. (2006). Oviduct contraction in Drosophila is modulated by a neural network that is both, octopaminergic and glutamatergic. *Journal of Cellular Physiology*, 209(1), 183–198. https://doi.org/10.1002/jcp.20722

- Roeder, T. R. (2005) TYRAMINE AND OCTOPAMINE: Ruling Behavior and Metabolism. *Annual Review of Entomology, 50,* 447–77. https://doi.org/10.1146/annurev.ento.50.071803.130404
- Rojek, W., & Kuszewska, K. (2022). Ovary activation correlates with the reproductive potential of honeybee (Apis mellifera) workers if they are in a foreign colony. *Insectes Sociaux*, 2018. https://doi.org/10.1007/s00040-022-00854-8
- Ronai, I., Allsopp, M. H., Tan, K., Dong, S., Liu, X., Vergoz, V., & Oldroyd, B. P. (2017). The dynamic association between ovariole loss and sterility in adult honeybee workers. *Proceedings of the Royal Society B*, 284, 20162693. https://doi.org/https://doi.org/10.1098/rspb.2016.2693
- Ronai, I., Barton, D. A., Oldroyd, B. P., & Vergoz, V. (2015). Regulation of oogenesis in honey bee workers via programed cell death. *Journal of Insect Physiology*, *81*, 36– 41. https://doi.org/10.1016/j.jinsphys.2015.06.014
- Rortais, A., Arnold, G., Halm, M. P., & Touffet-Briens, F. (2005). Modes of honeybees exposure to systemic insecticides: estimated amounts of contaminated pollen and nectar consumed by different categories of bees. *Apidologie*, *36*, 71–83. https://doi.org/10.1051/apido:2004071
- Roussel, E., Carcaud, J., Combe, M., Giurfa, M., & Sandoz, J. C. (2014). Olfactory coding in the honeybee lateral horn. *Current Biology*, 24(5), 561–567. https://doi.org/10.1016/j.cub.2014.01.063
- Ruepell, O., Bachelier, C., Fondrk, M. K., & Page, R. E. (2007). Regulation of life history determines lifespan of worker honey bees (Apis mellifera L.). *Exp Gerontol*, 42(10), 1020–1032.
- Russell, V. L. (2022). emmeans: Estimated Marginal Means, aka Least-Squares Means. R package version 1.7.2. https://CRAN.R-project.org/package=emmeans
- Salomon, M., Malka, O., vander Meer, R. K., & Hefetz, A. (2012). The role of tyramine and octopamine in the regulation of reproduction in queenless worker honeybees. *Naturwissenschaften*, 99(2), 123–131. https://doi.org/10.1007/s00114-011-0877-x
- Sandrock, C., Tanadini, M., Tanadini, L. G., Fauser-Misslin, A., Potts, S. G., & Neumann,
 P. (2014). Impact of chronic neonicotinoid exposure on honeybee colony
 performance and queen supersedure. *PLoS ONE*, *9*(8), 1–13.
 https://doi.org/10.1371/journal.pone.0103592

- Sasaki, K., & Harano, K. I. (2007). Potential effects of tyramine on the transition to reproductive workers in honeybees (Apis mellifera L.): Short communication. *Physiological Entomology*, 32(2), 194–198. https://doi.org/10.1111/j.1365-3032.2007.00566.x
- Sasaki, K., Matsuyama, H., Morita, N., & Ono, M. (2017). Caste differences in the association between dopamine and reproduction in the bumble bee Bombus ignitus. *Journal of Insect Physiology*, 103, 107–116. https://doi.org/10.1016/j.jinsphys.2017.10.013
- Sasaki, K., Matsuyama, S., Harano, K. ichi, & Nagao, T. (2012). Caste differences in dopamine-related substances and dopamine supply in the brains of honeybees (Apis mellifera L.). *General and Comparative Endocrinology*, *178*(1), 46–53. https://doi.org/10.1016/j.ygcen.2012.04.006
- Sasaki, K., & Nagao, T. (2001). Distribution and levels of dopamine and its metabolites in brains of reproductive workers in honeybees. *Journal of Insect Physiology*, 47(10), 1205–1216. https://doi.org/10.1016/S0022-1910(01)00105-6
- Sasaki, K., Okada, Y., Shimoji, H., Aonuma, H., Miura, T., & Tsuji, K. (2021). Social Evolution With Decoupling of Multiple Roles of Biogenic Amines Into Different Phenotypes in Hymenoptera. *Frontiers in Ecology and Evolution*, 9(May), 1–12. https://doi.org/10.3389/fevo.2021.659160
- Sasaki, K., & Watanabe, T. (2022). Sex-Specific Regulatory Systems for Dopamine Production in the Honey Bee. *Insects*, 13(2). https://doi.org/10.3390/insects13020128
- Sasaki, K., Yamasaki, K., & Nagao, T. (2007). Neuro-endocrine correlates of ovarian development and egg-laying behaviors in the primitively eusocial wasp (Polistes chinensis). *Journal of Insect Physiology*, 53(9), 940–949. https://doi.org/10.1016/j.jinsphys.2007.03.006
- Sasaki, K., Yamasaki, K., Tsuchida, K., & Nagao, T. (2009). Gonadotropic effects of dopamine in isolated workers of the primitively eusocial wasp, Polistes chinensis. *Naturwissenschaften*, *96*(5), 625–629. https://doi.org/10.1007/s00114-009-0510-4
- Schmidt Capella, I. C., & Hartfelder, K. (2002). Juvenile-hormone-dependent interaction of actin and spectrin is crucial for polymorphic differentiation of the larval honey bee ovary. *Cell and Tissue Research*, 307(2), 265–272. https://doi.org/10.1007/s00441-001-0490-y

- Schneider, C. W., Tautz, J., Grünewald, B., & Fuchs, S. (2012). RFID tracking of sublethal effects of two neonicotinoid insecticides on the foraging behavior of Apis mellifera. *PLoS ONE*, 7(1), 1–9. https://doi.org/10.1371/journal.pone.0030023
- Schulz, D. J., & Robinson, G. E. (2001). Octopamine influences division of labor in honey bee colonies. *Journal of Comparative Physiology - A Sensory, Neural, and Behavioral Physiology*, 187(1), 53–61. https://doi.org/10.1007/s003590000177
- Schulz, D. J., Sullivan, J. P., & Robinson, G. E. (2002). Juvenile hormone and octopamine in the regulation of division of labor in honey bee colonies. *Hormones and Behavior*, 42(2), 222–231. https://doi.org/10.1006/hbeh.2002.1806
- Seeley, T.D. (1985) Honeybee Ecology: A Study of Adaptation in Social Life. Princeton: Princeton University Press.
- Sharma, A., Kumar, V., Shahzad, B., Tanveer, M., Sidhu, G. P. S., Handa, N., Kohli, S. K., Yadav, P., Bali, A. S., Parihar, R. D., Dar, O. I., Singh, K., Jasrotia, S., Bakshi, P., Ramakrishnan, M., Kumar, S., Bhardwaj, R., & Thukral, A. K. (2019). Worldwide pesticide usage and its impacts on ecosystem. *SN Applied Sciences*, 1(11), 1–16. https://doi.org/10.1007/s42452-019-1485-1
- Sherer, L. M., Garrett, E. C., Morgan, H. R., Brewer, E. D., Sirrs, L. A., Shearin, H. K., Williams, J. L., McCabe, B. D., Stowers, R. S., & Certel, S. J. (2020). Octopamine neuron dependent aggression requires dVGLUT from dual-transmitting neurons. *PLoS Genetics*, 16(2), 1–26. https://doi.org/10.1371/journal.pgen.1008609
- Slessor, K., Kaminski, L.-A., King, G., Borden, J., & Wilston, M. (1988). Semiochemical basis of the retinue response to queen honey bees. *Letters to Nature*, 332, 354– 356.
- Slessor, K. N., Kaminski, L. A., King, G. G. S., & Winston, M. L. (1990). Semiochemicals of the honeybee queen mandibular glands. *Journal of Chemical Ecology*, 16(3), 851– 860. https://doi.org/10.1007/BF01016495
- Slessor, K. N., Winston, M. L., & Conte, Y. L. E. (2005). Pheromone Communication in the Honeybee (Apis mellifera L.). *Journal of Chemical Ecology*, 31(11), 2731–2745. https://doi.org/10.1007/s10886-005-7623-9
- Smith, A. A., Hölldober, B., & Liebig, J. (2009). Cuticular Hydrocarbons Reliably Identify Cheaters and Allow Enforcement of Altruism in a Social Insect. *Current Biology*, 19(1), 78–81. https://doi.org/10.1016/j.cub.2008.11.059

- Smith, S. M., Beattie, A. J., Kent, D. S., & Stow, A. J. (2009). Ploidy of the eusocial beetle austroplatypus incompertus (Schedl) (Coleoptera, Curculionidae) and implications for the evolution of eusociality. *Insectes Sociaux*, 56(3), 285–288. https://doi.org/10.1007/s00040-009-0022-4
- Snodgrass, R. (1956) Anatomy of the honey bee. Ithaca: Cornell Paperbacks
- Stern, D. L. (1994). A phylogenetic analysis of soldier evolution in the aphid family
 Hormaphididae. *Proceedings of the Royal Society B: Biological Sciences*, 256(1346),
 203–209. https://doi.org/10.1098/rspb.1994.0071
- Strauss, K., Scharpenberg, H., Crewe, R. M., Glahn, F., Foth, H., & Moritz, R. F. A. (2008). The role of the queen mandibular gland pheromone in honeybees (Apis mellifera): Honest signal or suppressive agent? *Behavioral Ecology and Sociobiology*, *62*(9), 1523–1531. https://doi.org/10.1007/s00265-008-0581-9
- Szathmáry, E., & Smith, J. M. (1995). The major evolutionary transitions. *Nature*, *374*(6519), 227–232. https://doi.org/10.1038/374227a0
- Tan ,K., Liu, X., Dong, S., Wang, C., Oldroyd, B.P. (2015) Pheromones affecting ovary activation and ovariole loss in the Asian honey bee *Apis cerana*. J. Insect. Physiol, 74, 25–29. doi:10.1016/j. jinsphys.2015.01.006
- Therneau, T. M. (2020). coxme: Mixed Effects Cox Models. R package version 2.2-16. https://CRAN.R-project.org/package=coxme
- Therneau T. M. (2021). _A Package for Survival Analysis in R_. R package version 3.2-13, <URL: https://CRAN.R-project.org/package=survival>
- Tosi, S., & Nieh, J. C. (2017). A common neonicotinoid pesticide, thiamethoxam, alters honey bee activity, motor functions, and movement to light. *Scientific Reports*, 7(1), 1–13. https://doi.org/10.1038/s41598-017-15308-6
- Traynor, K. S., le Conte, Y., & Page, R. E. (2014). Queen and young larval pheromones impact nursing and reproductive physiology of honey bee (Apis mellifera) workers. *Behavioral Ecology and Sociobiology*, 68(12), 2059–2073. https://doi.org/10.1007/s00265-014-1811-y
- Tsvetkov, N., Samson-Robert, O., Sood, K., Patel, H. S., Malena, D. A., Gajiwala, P. H., Maciukiewicz, P., Fournier, V., & Zayed, A. (2017). Chronic exposure to neonicotinoids reduces honey bee health near corn crops. *Science*, 356(6345), 1395–1397. https://doi.org/10.1126/science.aam7470

- van der Zee, R., Pisa, L., Andonov, S., Brodschneider, R., Charrière, J. D., Chlebo, R., Coffey, M. F., Crailsheim, K., Dahle, B., Gajda, A., Gray, A., Drazic, M. M., Higes, M., Kauko, L., Kence, A., Kence, M., Kezic, N., Kiprijanovska, H., Kralj, J., ... Wilkins, S. (2012). Managed honey bee colony losses in Canada, China, Europe, Israel and Turkey, for the winters of 2008-9 and 2009-10. *Journal of Apicultural Research*, *51*(1), 100–114. https://doi.org/10.3896/IBRA.1.51.1.12
- van Oystaeyen, A., Oliveira, R. C., Holman, L., van Zweden, J. S., Romero, C., Oi, C. A., d'Ettorre, P., Khalesi, M., Billen, J., Wäckers, F., Millar, J. G., & Wenseleers, T. (2014). Conserved class of queen pheromones stops social insect workers from reproducing. *Science*, 343(6168), 287–290. https://doi.org/10.1126/science.1244899
- van Zweden, J. S., Bonckaert, W., Wenseleers, T., & d'Ettorre, P. (2014). Queen signaling in social wasps. *Evolution*, 68(4), 976–986. https://doi.org/10.1111/evo.12314
- VanEngelsdorp, D., Hayes, J., Underwood, R. M., & Pettis, J. (2008). A survey of honey bee colony losses in the U.S., Fall 2007 to Spring 2008. *PLoS ONE*, *3*(12), 8–13. https://doi.org/10.1371/journal.pone.0004071
- vanEngelsdorp, D., & Meixner, M. D. (2010). A historical review of managed honey bee populations in Europe and the United States and the factors that may affect them. *Journal of Invertebrate Pathology*, *103*(SUPPL. 1), S80–S95. https://doi.org/10.1016/j.jip.2009.06.011
- Vergoz, V., Lim, J., & Oldroyd, B. P. (2012). Biogenic amine receptor gene expression in the ovarian tissue of the honey bee Apis mellifera. *Insect Molecular Biology*, 21, 21–29. https://doi.org/10.1111/j.1365-2583.2011.01106.x
- Vergoz, V., McQuillan, H. J., Geddes, L. H., Pullar, K., Nicholson, B. J., Paulin, M. G., & Mercer, A. R. (2009). Peripheral modulation of worker bee responses to queen mandibular pheromone. *Proceedings of the National Academy of Sciences of the United States of America*, 106(49), 20930–20935. https://doi.org/10.1073/pnas.0907563106
- Vergoz, V., Schreurs, H. A., & Mercer, A. R. (2007). Queen Pheromone Blocks Aversive Learning in Young Worker Bees. *Science*, *4297*, 6–8.
- Wagener-Hulme, C., Kuehn, J. C., Schulz, D. J., & Robinson, G. E. (1999). Biogenic amines and division of labor in honey bee colonies. *Journal of Comparative*

Physiology - A Sensory, Neural, and Behavioral Physiology, 184(5), 471–479. https://doi.org/10.1007/s003590050347

- Walton, A., Dolezal, A. G., Bakken, M. A., & Toth, A. L. (2018). Hungry for the queen : Honeybee nutritional environment affects worker pheromone response in a life stage - dependent manner. 32, 2699–2706. https://doi.org/10.1111/1365-2435.13222
- Walton, A., & Toth, A. L. (2016). Variation in individual worker honey bee behavior shows hallmarks of personality. *Behavioral Ecology and Sociobiology*, 70, 999– 1010. https://doi.org/10.1007/s00265-016-2084-4
- Watmough, J. (1997). A general model of pheromone transmission within honey bee hives. Journal of Theoretical Biology, 189(2), 159–170. https://doi.org/10.1006/jtbi.1997.0506
- Wenseleers, T., Helanterä, H., Hart, A., & Ratnieks, F. L. W. (2004). Worker reproduction and policing in insect societies: An ESS analysis. *Journal of Evolutionary Biology*, 17(5), 1035–1047. https://doi.org/10.1111/j.1420-9101.2004.00751.x
- Wenseleers, T., & Ratnieks, F. L. W. (2006). Comparative analysis of worker reproduction and policing in eusocial hymenoptera supports relatedness theory. *The American Naturalist*, 168(6). https://doi.org/10.1086/508619
- Wenseleers, T., Ratnieks, F. L. W., & Billen, J. (2003). Caste fate conflict in swarmfounding social Hymenoptera: An inclusive fitness analysis. *Journal of Evolutionary Biology*, 16(4), 647–658. https://doi.org/10.1046/j.1420-9101.2003.00574.x
- West, S. A., & Gardner, A. (2013). Adaptation and Inclusive Fitness. *Current Biology*, 23(13), R577–R584. https://doi.org/10.1016/j.cub.2013.05.031
- White, M. A., Chen, D. S., & Wolfner, M. F. (2021). She's got nerve: roles of octopamine in insect female reproduction. *Journal of Neurogenetics*, 35(3), 132–153. https://doi.org/10.1080/01677063.2020.1868457
- Whitehorn, P. R., O 'Connor, S., Wackers, F. L., & Goulson, D. (2012). Neonicotinoid Pesticide Reduces Bumble Bee Colony Growth and Neonicotinoid Pesticide Reduces Bumble Bee Colony Growth and Queen Production. *Science*, *336*(April), 351–352. https://doi.org/10.1126/science.1215025
- Wickham, H. (2016) ggplot2: Elegant Graphics for Data Analysis. Springer-Verlag New York

- Williams, J. R., Swale, D. R., & Anderson, T. D. (2020). Comparative effects of technicalgrade and formulated chlorantraniliprole to the survivorship and locomotor activity of the honey bee, Apis mellifera (L.). *Pest Management Science*, *76*(8), 2582–2588. https://doi.org/10.1002/ps.5832
- Williamson, S. M., Willis, S. J., & Wright, G. A. (2014). Exposure to neonicotinoids influences the motor function of adult worker honeybees. *Ecotoxicology*, 23(8), 1409–1418. https://doi.org/10.1007/s10646-014-1283-x
- Wilson, D. S., & Wilson, E. O. (2007). Rethinking the theoretical foundation of sociobiology. *Quarterly Review of Biology*, 82(4), 327–348. https://doi.org/10.1086/703580
- Wilson, E. O. (2008). One giant leap: How insects achieved altruism and colonial life. *BioScience*, 58(1), 17–25. https://doi.org/10.1641/B580106
- Wilson, E. O. (1975). The insect societies. Belknap-Harvard, Cambridge, Mass
- Wilson, E. O., & Hölldobler, B. (2005). Eusociality: Origin and consequences. Proceedings of the National Academy of Sciences of the United States of America, 102(38), 13367–13371. https://doi.org/10.1073/pnas.0505858102
- Winston, M. L., Higo, H. A., Slessor, K. N., Columbia, B., & Va, C. (1990). Effect of Various Dosages of Queen Mandibular Gland Pheromone on the Inhibition of Queen Rearing in the Honey Bee (Hymenoptera : Apidae). Annals of the Entomological Society of America, 83(2), 234–238.
- Winston, M. L., & Slessor, K. N. (1998). Honey bee primer pheromones and colony organization: Gaps in our knowledge. *Apidologie*, 29(1–2), 81–95. https://doi.org/10.1051/apido:19980105
- Wong, R., & Lange, A. B. (2014). Octopamine modulates a central pattern generator associated with egg-laying in the locust, Locusta migratoria. *Journal of Insect Physiology*, 63(1), 1–8. https://doi.org/10.1016/j.jinsphys.2014.02.002
- Woodcock, B. A., Bullock, J. M., Shore, R. F., Heard, M. S., Pereira, M. G., Redhead, J., Ridding, L., Dean, H., Sleep, D., Henrys, P., Peyton, J., Hulmes, S., Hulmes, L., Sárospataki, M., Saure, C., Edwards, M., Genersch, E., Knäbe, S., & Pywell, R. F. (2017). Country-specific effects of neonicotinoid pesticides on honey bees and wild bees. *Science*, *356*(6345), 1393–1395. https://doi.org/10.1126/science.aaa1190

- Wossler, T. C., & Crewe, R. M. (1999). Honeybee queen tergal gland secretion affects ovarian development in caged workers. *Apidologie*, *30*(4), 311–320. https://doi.org/10.1051/apido:19990407
- Wu, S. F., Xu, G., Stanley, D., Huang, J., & Ye, G. Y. (2015). Dopamine modulates hemocyte phagocytosis via a D1-like receptor in the rice stem borer, Chilo suppressalis. *Scientific Reports*, 5(June), 1–13. https://doi.org/10.1038/srep12247
- Wyatt, T. D. (2003). Pheromones and animal behavior: communication by smell and taste. Cambridge University Press, Cambridge, p 408
- Yamane, T. (2013). Effects of the biogenic amines on female ovipos- ition behavior in the rice leaf bug Trigonotylus caelestialium (Kirkaldy) (Heteroptera: Miridae). *Entomological News*, 123(2), 161–167. doi:10.3157/021.123.0208
- Yoshimura, H., Yamada, Y. Y., & Sasaki, K. (2021). Identification of biogenic amines involved in photoperiod-dependent caste-fate determination during the adult stage in a temperate paper wasp. *Journal of Insect Physiology*, 131(March), 104223. https://doi.org/10.1016/j.jinsphys.2021.104223
- Zhang, C., Kim, A.J., Rivera-Perez, C., Noriega, F.G. & Kim, Y.J. (2022) The insect somatostatin pathway gates vitellogenesis progression during reproductive maturation and the post-mating response. *Nature Communications*, 13, 969. https://doi.org/10.1038/s41467-022-28592-2
- Zillikens, A., Simões, Z. L. P., & Engels, W. (1998). Higher fertility of queenless workers in the Africanized honey bee. *Insectes Sociaux*, 45(4), 473–476. https://doi.org/10.1007/s000400050102